UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT
Pursuant To Section 13 or 15(d) of
the Securities Exchange Act of 1934

Date of Report: January 6, 2015 (Date of earliest event reported)

COHBAR, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation) 000-55334 (Commission File Number) 26-1299952 (I.R.S. Employer Identification No.)

2265 East Foothill Blvd.
Pasadena, CA 91107
(Address of principal executive offices and zip code)

(415) 388-2222 (Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):				
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)			
	Soliciting material pursuant to Rule 14a-12(b) under the Exchange Act (17 CFR 240.14a-12(b))			
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))			
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))			

COHBAR, INC. FORM 8-K

Item 3.02 Unregistered Sales of Equity Securities

As described below in Item 8.01, on January 6, 2015, CohBar, Inc. (the "Company") completed its initial public offering (the "IPO") of 11,250,000 units at a price of US\$1.00 per unit. Concurrently with the completion of the IPO, the Company issued and sold an additional 2,700,000 units to certain existing accredited investors pursuant to the terms of put agreements between the Company and such investors, at a price of US\$1.00 per unit. The concurrent placement was made pursuant to Section 4(a)(2) of the Securities Act of 1933, as amended, and Rule 506(b) promulgated thereunder. The terms of the concurrently placed units and the shares of common stock and warrants comprising such units are identical to the terms of the units and underlying securities issued in the IPO and described below.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers

In connection with the IPO, the Company amended and restated its 2011 Equity Incentive Plan (the "Plan"). The Plan authorizes the Company to grant up to 6,453,069 shares pursuant to stock awards and makes certain other changes to comply with the requirements of the TSX Venture Exchange.

A copy of the Amended and Restated 2011 Equity Incentive Plan is attached hereto as Exhibit 10.1.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year

Effective January 6, 2015, the Company amended and restated its certificate of incorporation and bylaws, as described in the Company's Registration Statement on Form S-1 filed in connection with the IPO, as amended. The Company's Third Amended and Restated Certificate of Incorporation, as filed with the Secretary of State of the State of Delaware (the "Restated Charter"), and its Amended and Restated Bylaws (the "Restated Bylaws"), are attached hereto as Exhibits 3.1 and 3.2, respectively.

The Restated Charter deletes the various rights, preferences and protective provisions specified for the Company's previously outstanding Series B Preferred Stock, which was automatically converted to common stock upon completion of the IPO. The Restated Charter authorizes a total of 80,000,000 shares of capital stock, consisting of 75,000,000 shares of common stock and 5,000,000 shares of preferred stock. The Restated Bylaws reflect modifications to stockholders' ability to call a special meeting of the stockholders, propose matters for consideration at stockholder meetings, and take action by written consent of the stockholders.

Item 8.01 Other Events

On January 6, 2015, the Company completed its initial public offering of 11,250,000 units at a price of US\$1.00 per unit, resulting in gross proceeds of US\$11,250,000. Haywood Securities Inc. acted as sole agent for the IPO. The Company issued a total of 786,696 compensation options to the agent and the selling group members in connection with the IPO. Each compensation option entitles the holder thereof to acquire a unit at a price of US\$1.00 at any time until July 6, 2016.

Each unit consists of one share of the Company's common stock and one-half of one common stock purchase warrant. Each whole warrant is exercisable to acquire one share of the Company's common stock at a price of US\$2.00 per share at any time up to January 6, 2017, subject to the Company's right to accelerate the expiration time of the warrants if at any time the volume-weighted average trading price of its common stock is equal to or exceeds US\$3.00 per share for twenty (20) consecutive trading days. The warrants are governed by a Warrant Indenture, dated as of January 6, 2015, between the Company and CST Trust Company, as Warrant Agent. A copy of the Warrant Indenture is attached as Exhibit 4.1 hereto.

The Company's common stock was accepted for listing and will be traded on the TSX Venture Exchange.

A copy of the press release issued by the Company in connection with the closing of the IPO is attached hereto as Exhibit 99.1.

Item 9.01 Financial Statements and Exhibits

- (d) Exhibits
- 3.1 Third Amended and Restated Certificate of Incorporation, as filed with the Secretary of State of the State of Delaware, effective January 6, 2015.
- 3.2 Amended and Restated Bylaws.
- 4.1 Warrant Indenture, dated January 6, 2015, between CST Trust Company and CohBar, Inc.
- 10.1 Amended and Restated 2011 Equity Incentive Plan.
- 99.1 CohBar, Inc., Press Release issued January 7, 2015.

Signature

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

COHBAR, INC. (Registrant)

January 7, 2015

(Date)

By: /s/ Jon Stern

Jon Stern

Chief Executive Officer

THIRD AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF COHBAR, INC. (Effective January 6, 2015)

CohBar, Inc., a corporation organized and existing under and by virtue of the General Corporation Law of the State of Delaware (the "Company"), does hereby certify that:

- 1. The Company's original Certificate of Incorporation was filed with the Secretary of State of the State of Delaware on September 16, 2009.
- 2. This Third Amended and Restated Certificate of Incorporation was duly adopted in accordance with Sections 242 and 245 of the General Corporation Law of the State of Delaware, and has been duly approved by the written consent of stockholders of the Company in accordance with Section 228 of the General Corporation Law of the State of Delaware.
 - 3. The text of the Certificate of Incorporation of CohBar, Inc., is hereby amended and restated in its entirety to read as follows:

ARTICLE I.

The name of this corporation is CohBar, Inc.

ARTICLE II.

The address of the Company's registered office in the State of Delaware is to be located at 160 Greentree Drive, Suite 101, in the City of Dover, County of Kent, 19904. The name of its registered agent at such address is National Registered Agents, Inc.

ARTICLE III.

The Company's purpose is to engage in any lawful act or activity for which corporations may be organized under the Delaware General Corporation Law, as the same exists or as may hereafter be amended from time to time (the "DGCL").

ARTICLE IV.

The Company is authorized to issue two (2) classes of stock, to be designated, respectively, "Common Stock" and "Preferred Stock." The total number of shares which the Company is authorized to issue is 80,000,000 shares, \$0.001 par value per share. 75,000,000 shares shall be Common Stock and 5,000,000 shares shall be Preferred Stock.

The following is a statement of the designations and the powers, privileges and rights, and the qualifications, limitations or restrictions thereof in respect of each class of capital stock of the Company.

1. Common Stock.

- (A) <u>General</u>. The voting, dividend, and liquidation rights of the holders of Common Stock are subject to and qualified by the rights of the holders of Preferred Stock of any series as may be designated by the Board of Directors upon issuance of Preferred Stock of any series.
- (B) <u>Voting</u>. Each holder of Common Stock, as such, shall be entitled to one vote for each share of Common Stock held of record by such holder on all matters on which stockholders generally are entitled to vote; provided, however, that, except as otherwise required by law, the holders of Common Stock, as such, shall not be entitled to vote on any amendment to this Certificate of Incorporation (which, as used herein, shall mean the certificate of incorporation of the Company, as amended from time to time, including the terms of any certificate of designation of any series of Preferred Stock) that relates solely to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together as a class with the holders of one or more other such series of Preferred Stock, to vote thereon pursuant to this Certificate of Incorporation or pursuant to the DGCL. There shall be no cumulative voting.
- (C) <u>Dividends</u>. Dividends may be declared and paid on Common Stock from funds lawfully available therefor as and when determined by the Board of Directors and subject to any preferential dividend or other rights of any then-outstanding Preferred Stock.
- (D) <u>Liquidation</u>. Upon the dissolution or liquidation of the Company, whether voluntary or involuntary, holders of Common Stock will be entitled to receive all assets of the Company available for distribution to its stockholders, subject to any preferential or other rights of any then-outstanding Preferred Stock.
- 2. <u>Preferred Stock</u>. The Board of Directors is hereby empowered, without any action or vote by the Company's stockholders, to authorize by resolution or resolutions from time to time the issuance of one or more classes or series of Preferred Stock and to fix the designations, powers, preferences and relative participating, optional or other rights, if any, and the qualifications, limitations or restrictions thereof, if any, with respect to each such class or series of Preferred Stock and the number of shares constituting each such class or series, and to increase or decrease the number of shares of any such class or series to the extent permitted by the DGCL. Any shares of Preferred Stock which may be redeemed, purchased or acquired by the Company may be reissued except as otherwise provided by the DGCL.

ARTICLE V.

1. <u>Limitation of Liability</u>. Except to the extent that the DGCL prohibits the elimination or limitation of liability of directors for breaches of fiduciary duty, no Director of the Company shall be personally liable to the Company or its stockholders for monetary damages for any breach of fiduciary duty as a director, notwithstanding any provision of law imposing such liability. No amendment to or repeal of this <u>Article V</u> shall apply to or have any effect on the liability or alleged liability of any Director of the Company for or with respect to any acts or omissions of such Director occurring prior to such amendment. If the DGCL is amended to permit further elimination or limitation of personal liability of directors, the liability of a director of the Company shall be eliminated or limited to the fullest extent permitted by the DGCL.

- 2. <u>Indemnification</u>. The Company shall, to the fullest extent permitted by the provisions of Section 145 of the DGCL, as the same may be amended or supplemented, indemnify past and present Directors, and the indemnification provided for herein shall not be deemed exclusive of any other rights to which those indemnified may be entitled under any by-law, agreement, vote of stockholders or disinterested Directors or otherwise, both as to action in his official capacity and as to action in another capacity while holding such office, and shall continue as to a person who has ceased to be a director, and shall inure to the benefit of the heirs, executors, and administrators of such person.
- 3. <u>Subsequent Amendment</u>. No amendment, termination or repeal of this <u>Article V</u> or of the relevant provisions of the DGCL or any other applicable laws shall affect or diminish in any way the rights of any Director to indemnification under the provisions hereof with respect to any action, suit, proceeding or investigation arising out of or relating to any actions, transactions or facts occurring prior to the final adoption of such amendment, termination or repeal.

ARTICLE VI.

The Court of Chancery of the State of Delaware shall be the sole and exclusive forum for (i) any derivative action or proceeding brought on behalf of the Company; (ii) any action asserting a claim of breach of fiduciary duty owed by a Director, officer or other employee of the Company to the Company or to the Company's stockholders; (iii) any action asserting a claim arising pursuant to any provision of the DGCL, this Certificate of Incorporation or the Company's By-Laws; or (iv) any action asserting a claim governed by the internal affairs doctrine. Any person or entity purchasing or otherwise acquiring any interest in shares of capital stock of the Company shall be deemed to have notice of and have consented to the provisions of this Article VI.

ARTICLE VII.

This Article VII is inserted for the management of the business and for the conduct of the affairs of the Company and for defining and regulating the powers of the Company and its directors and stockholders and is in furtherance and not in limitation of the powers conferred upon the Company by statute.

- 1. General Powers. The business and affairs of the Company shall be managed by or under the direction of a Board of Directors.
- 2. <u>Number: Election and Qualification</u>. Subject to the rights of holders of any class or series of Preferred Stock to elect persons to the Board of Directors, the number of Directors shall be established solely by the Board of Directors; provided, however that the Board of Directors shall have at least one (1) member. Election of persons to the Board of Directors need not be by written ballot.
- 3. <u>Tenure</u>. Subject to the rights of holders of any class or series of Preferred Stock to elect persons to the Board of Directors, Directors shall be elected at each annual meeting of

the stockholders; provided, that the term of each Director shall continue until the election and qualification of such Director's successor and be subject to such Director's earlier death, resignation or removal.

- 4. <u>Removal</u>. Subject to the rights of holders of any class or series of Preferred Stock to elect persons to the Board of Directors, members of the Board of Directors may be removed, with or without cause, by the affirmative vote of the holders of at least a majority of the total voting power of the issued and outstanding shares of the Company's capital stock, voting together as a single class.
- 5. <u>Vacancies</u>. Subject to the rights of holders of any class or series of Preferred Stock to elect persons to the Board of Directors, any vacancy on the Board of Directors, however occurring, including a vacancy resulting from an enlargement of the Board, may be filled by vote of a majority of the Directors then in office, although less than a quorum, or by a sole remaining Director. A person elected to fill a vacancy on the Board of Directors shall be elected for the unexpired term of such person's predecessor in office, and a person appointed to fill a position resulting from an increase in the size of the Board of Directors shall hold office until next annual meeting of stockholders and until the election and qualification of such person's successor and be subject to such person's earlier death, resignation or removal.
- 6. <u>Bylaws</u>. In furtherance and not in limitation of the powers conferred upon it by the DGCL, and subject to the rights of holders of any class or series of Preferred Stock, the Bylaws of the Company may be altered, amended or repealed or new Bylaws may be adopted only by (i) the affirmative vote of a majority of the Directors present at any regular or special meeting of the Board of Directors where a quorum is present or (ii) by the affirmative vote of the holders of at least a majority of the total voting power of the issued and outstanding shares of the Company's capital stock, voting together as a single class.
- 7. No Stockholder Action by Written Consent. Subject to the rights of holders of any class or series of Preferred Stock, any action required or permitted to be taken by the stockholders of the Company must be effected at a duly called annual or special meeting of stockholders of the Company and may not be effected by any consent in writing by such stockholders.
- 8. <u>Stockholder Nominations and Introduction of Business</u>. Advance notice of stockholder nominations for election of Directors and other business to be brought by stockholders before a meeting of stockholders shall be given in the manner provided by the Bylaws of the Company.
- 9. Special Meetings of Stockholders. Special meetings of stockholders may be called at any time only by the Board of Directors, Chairperson of the Board, or the Chief Executive Officer or, in the absence of the Chief Executive Officer, the President of the Company, and shall be called by the Company's Secretary upon the written request, validly given in the manner provided by the Bylaws of the Company, of one or more stockholders holding shares of record of the Company's capital stock representing in the aggregate at least twenty-five percent (25%) of the then outstanding shares of the Company's capital stock entitled to vote on the matter(s) proposed to be voted on at such meeting. The Board of Directors may

postpone or reschedule any previously scheduled special meeting of stockholders. Business transacted at any special meeting of stockholders shall be limited to matters relating to the purpose or purposes stated in the notice of meeting.

ARTICLE VIII.

The Company reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation in the manner now or hereafter permitted by the DGCL and all rights and powers conferred upon stockholders, Directors, and officers herein are granted subject to this reservation.

Notwithstanding anything contained in this Certificate of Incorporation or in the Company's Bylaws to the contrary, and notwithstanding the fact that a lesser percentage may be specified by the DGCL, the provisions set forth in Articles IV(2), V, VI, VII and this Article VIII may not be repealed or amended in any respect, and no other provisions may be adopted, amended or repealed that would have the effect of modifying or permitting the circumvention of the provisions set forth in Articles IV(2), V, VI, VII and this Article VIII, unless such action is approved by the affirmative vote of the holders of at least a majority of the total voting power of the issued and outstanding shares of the Company's capital stock, voting together as a single class.

IN WITNESS WHEREOF, the Company has caused this Third Amended and Restated Certificate of Incorporation to be signed by its duly authorized officer on this 5 day of January, 2015.

COHBAR, INC.

By: /s/ Jon Stern

Jon Stern

Its: Chief Executive Officer

AMENDED AND RESTATED

BYLAWS OF

COHBAR, INC.

Adopted January 6, 2015

AMENDED AND RESTATED BYLAWS OF COHBAR, INC.

ARTICLE I - STOCKHOLDERS

- 1.1 <u>Place of Meetings</u>. Meetings of stockholders of CohBar, Inc. (the "Corporation") shall be held at any place, within or outside the State of Delaware, determined by the Corporation's Board of Directors, the Chairperson of the Board of Directors, the Chief Executive Officer or the President (in the absence of a Chief Executive Officer). The Board of Directors may, in its sole discretion, determine that a meeting of stockholders shall not be held at any place, but may instead be held solely by means of remote communication as authorized by Section 211(a)(2) of the Delaware General Corporation Law (the "DGCL"). In the absence of any such designation or determination, stockholders' meetings shall be held at the Corporation's principal executive office.
- 1.2 <u>Annual Meeting</u>. The annual meeting of stockholders for the election of directors and for the transaction of such other business as may properly be brought before the meeting shall be held on a date and at a time designated by the Board of Directors, Chairperson of the Board of Directors, the Chief Executive Officer or the President (in the absence of a Chief Executive Officer), which date shall not be a legal holiday in the place where the meeting is to be held.
- 1.3 <u>Special Meetings</u>. Special meetings of the stockholders may be called at any time by the Board of Directors, the Chairperson of the Board of Directors, the Chief Executive Officer, or the President (in the absence of a Chief Executive Officer), and shall be called by the Secretary upon the valid written request of one or more stockholders holding shares of record of the Corporation's capital stock representing in the aggregate at least twenty-five percent (25%) (the "**Requisite Percentage**") of the then outstanding shares of the Corporation's capital stock entitled to vote on the matter(s) proposed to be voted on at such meeting (each, a "**Requesting Stockholder**").

Any meeting called at the valid request of the Requesting Stockholder(s) pursuant to this <u>Section 1.3</u> shall be held at such date, time and place as may be fixed by the Board of Directors, provided that the date of such special meeting shall not be more than ninety (90) days after the receipt by the Secretary of such request. To be valid, the request or requests must (i) be written, (ii) be delivered to the Secretary at the Corporation's principal executive office (the date on which the Secretary receives the request is the "**Delivery Date**"), (iii) include the specific purpose(s) of the special meeting of stockholders and the specific matter(s) proposed to be voted on at the meeting, (iv) comply with the requirements of <u>Section 1.11(a)(iii)</u> of these Bylaws, (v) include documentary evidence that the Requesting Stockholder(s) own the Requisite Percentage on the Delivery Date, (vi) include a certification that each such Requesting Stockholder will continue to hold at least the number of shares of common stock set forth in the request with respect to each such Requesting Stockholder through the date of the special meeting and (vii) be signed and dated by the Requesting Stockholder(s) or a duly authorized agent of such Requesting Stockholder(s) are not the beneficial owners of the shares

representing the Requisite Percentage, then the documentary evidence required by subsection (v) of this Section 1.3 must also include proof that the beneficial owners on whose behalf the request(s) are made beneficially own the Requisite Percentage on the Delivery Date in order for the request to be valid. For purposes of these Bylaws, the term "beneficial owner" shall have the meaning ascribed in Rule 13d-3 of the Securities Exchange Act of 1934, as amended.

Any Requesting Stockholder who submits a written request for a special meeting of stockholders may revoke that written request at any time by delivering a written revocation to the Secretary at the Corporation's principal executive office. The failure of any Requesting Stockholder to appear at the special meeting of stockholders or to send a qualified representative to the special meeting of stockholders to present such matter(s) to be voted on at the special meeting of stockholders shall also constitute a revocation of such request. If there is more than one Requesting Stockholder and the revocation or deemed revocation by one or more Requesting Stockholders causes the remaining Requesting Stockholders to hold in the aggregate less than the Requisite Percentage, the Board of Directors, in its discretion, may cancel the special meeting. If none of the Requesting Stockholder(s) appears or sends a qualified representative to present the nominations proposed to be presented or other business proposed to be conducted at the special meeting, the Corporation need not present such nominations or other business for a vote at the special meeting.

The Corporation is not required to call a special meeting of stockholders pursuant to this Section 1.3 with respect to any matter proposed to be presented by the Requesting Stockholder(s) if (w) the Delivery Date is during the period commencing 90 days prior to the first anniversary of the date of the immediately preceding annual meeting and ending on the date of the next annual meeting, (x) a substantially similar matter was included on the agenda of any annual or special meeting of stockholders held within 120 days prior to the Delivery Date, or will be included on the agenda at an annual or special meeting to be held within 90 days after the Delivery Date (and for purposes of this clause (x), a proposal involving an increase or decrease in the authorized number of directors, or the nomination, appointment, election or removal of directors, or the nomination, appointment, election or removal of directors, or the nomination, appointment applicable law, or (z) the written request for a special meeting of stockholders itself, including the matter proposed, fails to comply with applicable law(s) or this Section 1.3.

The business conducted at any special meeting of stockholders called in accordance with this <u>Section 1.3</u> shall be limited to the business set forth in the notice of the meeting; provided, however, that the Board of Directors may submit additional matters to the stockholders at the meeting by including those matters in the notice of the special meeting of stockholders. The Board of Directors may postpone or reschedule any previously scheduled special meeting of stockholders.

1.4 <u>Notice of Stockholders' Meetings</u>. Except as otherwise provided by law, written notice of each meeting of stockholders, whether annual or special, shall be given not less than ten (10) nor more than sixty (60) days before the date of the meeting to each stockholder entitled to vote at such meeting. The notices of all meetings shall state the place, date and time of the meeting. The notice of a special meeting shall state, in addition, the purpose or purposes for

which the meeting is called. Without limiting the manner in which notice may be given to stockholders, any notice shall be effective if given by a form of electronic transmission consented to (in a manner consistent with the DGCL) by the stockholder to whom the notice is given. If mailed, notice is given when deposited in the United States mail, postage prepaid, directed to the stockholder at such stockholder's address as it appears on the records of the Corporation. If notice is given by electronic transmission, such notice shall be deemed given at the time specified in Section 232 of the DGCL.

- 1.5 <u>Voting List</u>. The Secretary of the Corporation shall prepare, at least ten (10) days before each meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting, arranged in alphabetical order and showing the address of each stockholder and the number of shares registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours, for a period of at least ten (10) days prior to the meeting, at a place within the city where the meeting is to be held. The list shall also be produced and kept at the time and place of the meeting during the whole time of the meeting, and may be inspected by any stockholder who is present. The list shall presumptively determine the identity of the stockholders entitled to vote at the meeting and the number of shares held by each such stockholder.
- 1.6 Quorum. Except as otherwise provided by law, the Certificate of Incorporation or these Bylaws, the holders of not less than one-third (1/3) of the total voting power of shares of the capital stock of the Corporation issued and outstanding and entitled to vote at the meeting, present in person, present by means of remote communication in a manner, if any, authorized by the Board of Directors in its sole discretion or represented by proxy, shall constitute a quorum for the transaction of business; provided, however, that where a separate vote by a class or classes or series of capital stock is required by law or the Certificate of Incorporation, the holders of not less than one-third (1/3) of the total voting power of the shares of such class or classes or series of the capital stock of the Corporation issued and outstanding and entitled to vote on such matter, present in person, present by means of remote communication in a manner, if any, authorized by the Board of Directors in its sole discretion or represented by proxy, shall constitute a quorum entitled to take action with respect to the vote on such matter. A quorum, once established at a meeting, shall not be broken by the withdrawal of a sufficient number of votes to leave less than a quorum remaining at the meeting.
- 1.7 <u>Adjournments</u>. Any meeting of stockholders may be adjourned to any other time and to any other place at which a meeting of stockholders may be held under these Bylaws by the stockholders present or represented at the meeting and entitled to vote, although less than a quorum, or, if no stockholder is present, any officer entitled to preside at or to act as secretary of such meeting. It shall not be necessary to notify any stockholder of any adjournment of less than thirty (30) days if the time and place of the adjourned meeting, and the means of remote communication, if any, by which stockholders and proxyholders may be deemed to be present in person and vote at such adjourned meeting, are announced at the meeting at which adjournment is taken, unless after the adjournment a new record date is fixed for the adjourned meeting. At the adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting.

- 1.8 <u>Voting and Proxies</u>. Each stockholder shall have one (1) vote for each share of stock entitled to vote held of record by such stockholder and a proportionate vote for each fractional share so held, unless otherwise provided by law or the Certificate of Incorporation. Each stockholder of record entitled to vote at a meeting of stockholders may vote in person (including by means of remote communication, if any, by which stockholders may be deemed to be present in person and vote at such meeting) or may authorize another person or persons to vote for such stockholder by a proxy executed or transmitted in a manner permitted by the DGCL by the stockholder or such stockholder's authorized agent and delivered (including by electronic transmission) to the Secretary of the Corporation. No such proxy shall be voted upon after three years from the date of its execution, unless the proxy expressly provides for a longer period.
- 1.9 <u>Action at Meeting</u>. When a quorum is present at any meeting, any matter other than the election of directors to be voted upon by the stockholders at such meeting shall be decided by the vote of the holders of shares of stock having a majority in voting power of the votes cast by the holders of all shares of stock present or represented at the meeting and voting on such matter (or if there are two or more classes or series of stock entitled to vote as separate classes, then in the case of each such class or series, the holders of a majority in voting power of the shares of stock of that class or series present or represented and voting on such matter), except where a different vote is required by law, the Certificate of Incorporation or these Bylaws. When a quorum is present at any meeting, any election by stockholders of persons to the Board of Directors shall be determined by a plurality of the votes cast by the stockholders entitled to vote on the election.
- 1.10 No Action by Written Consent. Subject to the rights of holders of any class or series of preferred stock then outstanding, as may be set forth in the certificate of designation relating to such preferred stock, any action required or permitted to be taken at any annual or special meeting of stockholders may be taken only upon the vote of stockholders at an annual or special meeting duly noticed and called in accordance with these Bylaws and the DGCL and may not be taken by written consent of stockholders without a meeting.
- 1.11 Notice of Nominations and Stockholder Business. Except for (1) any persons entitled to be elected to the Board of Directors by the holders of preferred stock, (2) any directors elected in accordance with Section 2.5 hereof by the Board of Directors to fill any vacancy or newly-created directorship, or (3) as otherwise required by law or stock exchange regulation, only persons who are nominated in accordance with the procedures of this Section 1.11 shall be eligible for election to the Board of Directors.

(a) Annual Meetings of Stockholders.

(i) Nominations of persons for election to the Board of Directors or the proposal of other business to be transacted by stockholders at an annual meeting of stockholders may be made only (A) pursuant to the provisions of Section 1.4 hereof, (B) by or at the direction of the Board of Directors, or (C) by any stockholder of the Corporation who is a stockholder of record at the time of giving of notice provided for in this Section 1.11(a) and at the time of the annual meeting, who shall be entitled to vote at the meeting, and who complies with the procedures set forth in this Section 1.11(a).

(ii) For nominations or other business to be properly brought before an annual meeting of stockholders pursuant to Section 1.11(a)(i)(C) hereof, the stockholder must have given timely notice thereof in writing to the Secretary of the Corporation and any such proposed business (other than the nomination of persons for election the Board of Directors) must constitute a proper matter for stockholder action. To be timely, a stockholder's notice shall be delivered to, or mailed and received by, the Secretary of the Corporation at the principal executive offices of the Corporation not less than ninety (90) days nor more than one hundred twenty (120) days prior to the first anniversary of the preceding year's annual meeting of stockholders; provided, however, that in the event that the date of the annual meeting is advanced more than thirty (30) days prior to such anniversary date or delayed more than seventy (70) days following such anniversary date, such notice must be received by the Corporation no earlier than one hundred twenty (120) days prior to such annual meeting and no later than the later of seventy (70) days prior to the date of the meeting or the tenth (10th) day following the day on which public announcement of the date of the meeting was first made by the Corporation. In no event shall the public announcement of an adjournment or postponement of an annual meeting commence a new time period (or extend any time period) of the giving of a stockholder's notice as described in this Section 1.11(a)(ii).

(iii) A stockholder's notice to the Secretary of the Corporation shall set forth (A) as to each person whom the stockholder proposes to nominate for election or reelection to the Board of Directors, all information relating to such person that that is required to be disclosed in solicitations of proxies for the election of directors, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), including such person's written consent to be named in the proxy statement as a nominee and to serving on the Board of Directors if elected, (B) as to any other business that the stockholder proposes to bring before the meeting, a brief description of the business desired to be brought before the meeting, the text of the proposal or business (including the text of any resolution proposed for consideration and, in the event that such business includes a proposal to amend these Bylaws, the text of the proposed amendment), the reasons for conducting such business and any material interest in such business of such stockholder and the Stockholder Associated Person (as defined below), if any, on whose behalf the proposal is made, and (C) as to the stockholder giving the notice and the Stockholder Associated Person, if any, on whose behalf the proposal is made: (1) the name and address of such stockholder (as they appear on the Corporation's books) and any such Stockholder Associated Person, (2) the class or series and number of shares of capital stock of the Corporation which are held of record or are beneficially owned by such stockholder and by any such Stockholder Associated Person, (3) a description of any agreement, arrangement or understanding between or among such stockholder and any such Stockholder Associated Person, any of their respective affiliates or associates, and any other person or persons (including their names) in connection with the proposal of such nomination or other business, (4) a description of any agreement, arrangement, or understanding (including, regardless of the form of settlement, any derivatives, long or short positions, profit interests, forwards, futures, swaps, options, warrants, convertible securities, stock appreciation or similar rights, hedging transactions and borrowed or loaned shares) that has been entered into by or on behalf of, or any other agreement, arrangement or understanding that has been made, the effect or intent of which is to create or mitigate loss to, manage risk or benefit of share price changes for, or increase or decrease the voting power of, such stockholder or any Stockholder Associated Person or any such nominee with respect to the Corporation's securities, (5) a representation that

the stockholder is a holder of record of the stock of the Corporation entitled to vote at such meeting and intends to appear in person or by proxy at the meeting to bring such nomination or other business before the meeting, and (6) a representation as to whether such stockholder or any Stockholder Associated Person intends or is part of a group that intends to (i) deliver a proxy statement and/or form a proxy of holders of at least the percentage of the voting power of the Corporation's outstanding capital stock required to approve or adopt the proposal or elect each such nominee and/or (ii) otherwise to solicit proxies from stockholders in support of such proposal or nomination. If requested by the Corporation, the information required pursuant to Section 1.11(a)(iii)(C)(2), (3) and (4) hereof shall be supplemented by such stockholder and any such Stockholder Associated Person not less than ten (10) days after the record date for the meeting to disclose such information as of the record date.

- (iv) For the purposes of this <u>Section 1.11</u>, the term "**Stockholder Associated Person**" as used in relation to any stockholder shall mean (A) any person controlling, directly or indirectly, or acting in concert with, such stockholder, (B) any beneficial owner of shares of stock of the Corporation owned of record or beneficially by such stockholder, and (C) any person controlling, controlled by, or under common control with such Stockholder Associated Person.
- (b) Special Meetings of Stockholders. If the election of persons to the Board of Directors is included as business to be brought before a special meeting of stockholders in the Corporation's notice of meeting delivered pursuant to Section 1.4 hereof, nominations of persons for election to the Board of Directors at a special meeting of stockholders may be made only (A) by or at the direction of the Board of Directors or (B) by any stockholder of the Corporation who is a stockholder of record at the time of giving of notice provided for in this Section 1.11(b) and at the time of the special meeting, who shall be entitled to vote at the meeting, and who complies with the procedures set forth in this Section 1.11(b). For nominations to be properly brought before any special meetings pursuant to Section 1.11(b)(B) hereof, the stockholder must have given timely notice thereof in writing to the Secretary of the Corporation. To be timely, a stockholder's notice shall be delivered to, or mailed to and received by, the Secretary of the Corporation at the principal executive offices of the Corporation (1) no earlier than one hundred twenty (120) days prior to the date of the special meeting and (2) no later than the later of ninety (90) days prior to the date of the special meeting was first made by the Corporation. A stockholder's notice to the Secretary of the Corporation pursuant to this Section 1.11(b) shall comply with the requirements of Section 1.11(a)(iii).

(c) General.

(i) At the request of the Board of Directors, any person nominated by the Board of Directors for election to the Board of Directors shall furnish to the Secretary of the Corporation the information required to be set forth in stockholder's notice of nomination that pertains to the nominee. No person shall be eligible to be nominated for election to the Board of Directors by a stockholder unless nominated in accordance with the procedures of this Section 1.11. No business proposed by a stockholder shall be conducted at a meeting of stockholders except in accordance with Section 1.11. The chairperson of the meeting shall, if the facts warrant, determine and declare to the meeting that a nomination was not made in accordance with the procedures prescribed by these Bylaws or that business was not properly brought before

the meeting, and if he should so determine, he shall so declare to the meeting and the defective nomination shall be disregarded or such business shall not be transacted, as the case may be. Notwithstanding the foregoing provisions of this Section 1.11, unless otherwise required by law, if the stockholder (or a qualified representative of the stockholder) does not appear at the annual or special meeting of stockholders of the Corporation to present a nomination or other proposed business, such nomination shall be disregarded or such proposed business shall not be transacted, as the case may be, notwithstanding that proxies in respect of such vote may have been received by the Corporation. For the purposes of this Section 1.11, to be considered a qualified representative of the stockholder, a person must be a duly authorized officer, manager or partner of such stockholder or must be authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders.

- (ii) Without limiting the foregoing provisions of this <u>Section 1.11</u>, a stockholder shall also comply with all applicable requirements of the Exchange Act, and the rules and regulations promulgated thereunder, with respect to the matters set forth in this <u>Section 1.11</u>; provided, however, that any reference in these Bylaws to the Exchange Act or such rules and regulations promulgated thereunder are not intended to, and shall not, limit any requirement applicable to nominations or proposals as to any other business to be considered pursuant to this <u>Section 1.11</u>.
- (d) <u>Proxy Access Rules</u>. Notwithstanding anything contained herein to the contrary, the notice requirements set forth herein with respect to nomination of persons for election to the Board of Directors or other business pursuant to this <u>Section 1.11</u> shall be deemed satisfied by a stockholder if such stockholder has submitted a proposal to the Corporation, a nomination in compliance with Rule 14a-11 under the Exchange Act, or a proposal in compliance with Rule 14a-8 under the Exchange Act, and such stockholder's nomination or proposal, as the case may be, has been included in a proxy statement that has been prepared by the Corporation to solicit proxies for the meeting of stockholders.
- 1.12 <u>Conduct of Meetings</u>. At each meeting of stockholders, the Chairperson of the Board, if one shall have been elected, or in the Chairperson of the Board's absence or if one shall not have been elected, the Director designated by the vote of a majority of the Directors present at such meeting, shall act as chairperson of the meeting. The Secretary of the Corporation shall act as secretary of the meeting and keep the minutes thereof or, in the Secretary's absence or inability to act, the person whom the Chairperson of the Board shall appoint shall perform such tasks. The order of business at all meetings of stockholders shall be as determined by the chairperson of the meeting.

ARTICLE 2 – DIRECTORS

2.1 <u>General Powers</u>. The business and affairs of the Corporation shall be managed by or under the direction of a Board of Directors, which may exercise all of the powers of the Corporation except as otherwise provided by law, the Certificate of Incorporation, or these Bylaws. In the event of a vacancy on the Board of Directors, the remaining members of the Board of Directors, except as otherwise provided by law, may exercise the powers of the full Board of Directors until the vacancy is filled.

- 2.2 <u>Number; Election and Qualification</u>. Subject to the rights of holders of any class or series of preferred stock to elect persons to the Board of Directors and the rules of any stock exchange on which the any securities of the Corporation may be listed, the number of Directors shall be established by the Board of Directors; provided, however that the Board of Directors shall have at least one (1) member. Election of persons to the Board of Directors need not be by written ballot. Members of the Board of Directors need not be stockholders of the Corporation.
- 2.3 <u>Chairperson of the Board; Vice Chairperson of the Board.</u> The Board of Directors may appoint from its members a Chairperson of the Board and a Vice Chairperson of the Board. If the Board of Directors appoints a Chairperson of the Board, such Chairperson shall perform such duties and possess such powers as are assigned by the Board of Directors and, if the Chairperson of the Board is also designated as the Corporation's Chief Executive Officer, shall have the powers and duties of the Chief Executive Officer prescribed in <u>Section 3.7</u> hereof. If the Board of Directors appoints a Vice Chairperson of the Board, such Vice Chairperson shall perform such duties and possess such powers as are assigned by the Board of Directors. Unless otherwise provided by the Board of Directors, the Chairperson of the Board, or in the Chairperson's absence, the Vice Chairperson of the Board, if any, shall preside at all meetings of the Board of Directors.
- 2.4 <u>Tenure</u>. Subject to the rights of holders of any class or series of preferred stock to elect persons to the Board of Directors, directors shall be elected at each annual meeting of the stockholders; provided, that the term of each Director shall continue until the election and qualification of such Director's successor and be subject to such Director's earlier death, resignation or removal.
- 2.5 <u>Vacancies</u>. Subject to the rights of holders of any class or series of Preferred Stock to elect persons to the Board of Directors, any vacancy on the Board of Directors, however occurring, including a vacancy resulting from an enlargement of the Board of Directors, may be filled by vote of a majority of the Directors then in office, although less than a quorum, or by a sole remaining Director. A person elected to fill a vacancy on the Board of Directors shall be elected for the unexpired term of such person's predecessor in office, and a person appointed to fill a position resulting from an increase in the size of the Board of Directors shall hold office until next annual meeting of stockholders and until the election and qualification of such person's successor and be subject to such person's earlier death, resignation or removal.
- 2.6 <u>Resignation</u>. Any Director may resign by delivering a resignation in writing or by electronic transmission to the Corporation at its principal executive office or to the Chairperson of the Board, if any, the Chief Executive Officer, the President, or the Secretary of the Corporation. Such resignation shall be effective upon delivery unless it is specified to be effective at some time later or upon the occurrence of some later event.
- 2.7 <u>Regular Meetings</u>. Regular meetings of the Board of Directors may be held without notice at such time and place, either within or outside the State of Delaware, as shall be determined from time to time by the Board of Directors; provided that any Director who is

absent when such a determination is made shall be given notice of the determination. A regular meeting of the Board of Directors may be held without notice immediately after and at the same place as the annual meeting of stockholders.

- 2.8 <u>Special Meetings</u>. Special meetings of the Board of Directors may be held at any time and place, within or without the State of Delaware, designated in a call by the Chairperson of the Board, if any, the Chief Executive Officer, the President, three (3) or more members of the Board of Directors, or by one (1) member of the Board of Directors in the event that there is only a single Director in office.
- 2.9 Notice of Special Meetings. Notice of the date, place, and time of any special meeting of the Board of Directors shall be given to each Director by the Secretary of the Corporation or by the officer or one of the Directors calling the meeting. Notice shall be duly given to each member of the Board of Directors: (i) in person or by telephone at least twenty-four (24) hours in advance of the meeting, (ii) by sending written notice by reputable overnight courier, telecopy, facsimile, electronic transmission, or delivering written notice by hand to such Director's last known business, home, or electronic transmission address at least forty-eight (48) hours in advance of the meeting, or (iii) by sending written notice by first-class mail to such Director's last known business or home address at least ninety-six (96) hours in advance of the meeting. A notice or waiver of notice of a meeting of the Board of Directors need not specify the purposes of the meeting.
- 2.10 Meetings by Conference Communications Equipment. Members of the Board of Directors may participate in meetings of the Board of Directors or of any committee thereof by means of conference telephone or similar communications equipment by means of which all persons participating in the meeting can hear each other, and participation by such means shall constitute presence in person at such meeting.
- 2.11 Quorum. A majority of the total number of members of the whole Board of Directors shall constitute a quorum at all meetings of the Board of Directors. In the absence of a quorum at any such meeting, a majority of the Directors present may adjourn the meeting from time to time without further notice other than announcement at the meeting, until a quorum shall be present.
- 2.12 Action at Meeting. At any meeting of the Board of Directors at which a quorum is present, the vote of a majority of those present shall be sufficient to take any action, unless a different vote is specified by law, the Certificate of Incorporation or these Bylaws.
- 2.13 <u>Action by Consent</u>. Any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting, if all members of the Board of Directors or committee, as the case may be, consent to the action in writing or by electronic transmission, and the written consents or electronic transmissions are filed with the minutes of proceedings of the Board of Directors or committee.
- 2.14 <u>Removal</u>. Subject to the rights of holders of any class or series of preferred stock to elect persons to the Board of Directors and the requirements of any exchange on which the Corporation's securities may be listed, members of the Board of Directors may be removed, with

or without cause, by the affirmative vote of the holders of at least a majority of the total voting power of the issued and outstanding shares of the Corporation's capital stock, voting together as a single class.

- 2.15 Committees. The Board of Directors may, by resolution passed by a majority of the entire Board of Directors, designate one (1) or more committees, each committee to consist of one (1) or more of the members of the Board of Directors of the Corporation. The Board of Directors may designate one (1) or more of its members as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. In the absence or disqualification of a member of a committee, the member or members of the committee present at any meeting and not disqualified from voting, whether or not he, she, or they constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board of Directors and subject to the provisions of the DGCL, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation and may authorize the seal of the Corporation to be affixed to all papers which may require it. Each such committee shall keep minutes and make such reports as the Board of Directors may from time to time request. Except as the Board of Directors may otherwise determine, any committee may make rules for the conduct of its business, but unless otherwise provided by the members of such committee or in such rules, its business shall be conducted as nearly as possible in the same manner as is provided in these Bylaws for the Board of Directors.
- 2.16 <u>Compensation of Directors</u>. Members of the Board Directors may be paid such compensation for their services and such reimbursement for expenses of attendance at meetings as the Board of Directors may from time to time determine. No such payment shall preclude any Director from serving the Corporation or any of its parent or subsidiary corporations in any other capacity and receiving compensation for such service.

ARTICLE III - OFFICERS

- 3.1 <u>Enumeration</u>. The officers of the Corporation shall consist of a Chief Executive Officer, a Chief Financial Officer, a President, a Secretary, a Treasurer, one (1) or more Vice Presidents and such other officers with such other titles as the Board of Directors shall determine, including one or more Assistant Treasurers and Assistant Secretaries. The Board of Directors may appoint such other officers as it may deem appropriate.
- 3.2 <u>Election of Officers</u>. The Chief Executive Officer, Chief Financial Officer, President, Treasurer and Secretary shall be elected annually by the Board of Directors at its first meeting following the annual meeting of stockholders. Other officers may be appointed by the Board of Directors at such meeting or at any other meeting.
 - 3.3 Qualification. No officer need be a stockholder. Any two (2) or more offices may be held by the same person.
- 3.4 <u>Tenure</u>. Except as otherwise provided by law, by the Certificate of Incorporation or by these Bylaws, each officer shall hold office until such officer's successor is elected and qualified, unless a different term is specified in the resolution choosing or appointing such officer, or until such officer's earlier death, resignation or removal.

- 3.5 <u>Resignation and Removal</u>. Any officer may resign by delivering a resignation in writing or by electronic transmission to the Corporation at its principal executive office or to the Chairperson of the Board, if any, the Chief Executive Officer, the President or the Secretary of the Corporation. Such resignation shall be effective upon delivery unless it is specified to be effective at some time later or upon the occurrence of some later event. Any officer may be removed at any time, with or without cause, by vote of a majority of the members of the Board of Directors then in office. Except as the Board of Directors may otherwise determine, no officer who resigns or is removed shall have any right to any compensation as an officer for any period following such officer's resignation or removal, or any right to damages on account of such removal, whether such officer's compensation be by the month or by the year or otherwise, unless such compensation is expressly provided in a duly authorized written agreement with the Corporation.
- 3.6 <u>Vacancies</u>. The Board of Directors may fill any vacancy occurring in any office for any reason and may, in its discretion, leave unfilled for such period as it may determine any offices other than those of Chief Executive Officer, Chief Financial Officer, President, Treasurer and Secretary. Each such successor shall hold office for the unexpired term of such officer's predecessor and until such officer's successor is elected and qualified, or until such officer's earlier death, resignation or removal.
- 3.7 <u>Chief Executive Officer</u>; <u>President</u>. Unless the Board of Directors has designated another person as the Corporation's Chief Executive Officer, the President shall be the Chief Executive Officer of the Corporation. The Chief Executive Officer shall have general charge and supervision of the business of the Corporation subject to the direction of the Board of Directors, and shall perform all duties and have all powers that are commonly incident to the office of chief executive officer or that are delegated to such officer by the Board of Directors. The President shall perform such other duties and shall have such other powers as the Board of Directors or the Chief Executive Officer (if the President is not the Chief Executive Officer) may from time to time prescribe. In the event of the absence, inability or refusal to act of the Chief Executive Officer or the President (if the President is not the Chief Executive Officer), the Chief Financial Officer shall perform the duties of the Chief Executive Officer and when so performing such duties shall have all the powers of and be subject to all the restrictions upon the Chief Executive Officer.
- 3.8 <u>Chief Financial Officer</u>. The Chief Financial Officer shall have general charge and supervision of the financial operations of the Corporation subject to the direction of the Board of Directors and the Chief Executive Officer, and shall perform all duties and have all powers that are commonly incident to the office of chief financial officer or that are delegated to such officer by the Board of Directors or the Chief Executive Officer from time to time.
- 3.9 <u>Vice Presidents</u>. Each Vice President shall perform such duties and possess such powers as the Board of Directors or the Chief Executive Officer may from time to time prescribe. The Board of Directors may assign to any Vice President the title of Executive Vice President, Senior Vice President or any other title selected by the Board of Directors.

3.10 Secretary and Assistant Secretaries. The Secretary shall perform such duties and shall have such powers as the Board of Directors or the Chief Executive Officer may from time to time prescribe. In addition, the Secretary shall perform such duties and have such powers as are incident to the office of the secretary, including without limitation the duty and power to give notices of all meetings of stockholders and special meetings of the Board of Directors, to attend all meetings of stockholders and the Board of Directors and keep a record of the proceedings, to maintain a stock ledger and prepare lists of stockholders and their addresses as required, to be custodian of corporate records and the corporate seal and to affix and attest to the same on documents.

Any Assistant Secretary shall perform such duties and possess such powers as the Board of Directors, the Chief Executive Officer or the Secretary may from time to time prescribe. In the event of the absence, inability or refusal to act of the Secretary, the Assistant Secretary (or if there shall be more than one, the Assistant Secretaries in the order determined by the Board of Directors) shall perform the duties and exercise the powers of the Secretary.

In the absence of the Secretary or any Assistant Secretary at any meeting of stockholders or the Board of Directors, the person presiding at the meeting shall designate a temporary secretary to keep a record of the meeting.

3.11 <u>Treasurer and Assistant Treasurers</u>. The Treasurer shall perform such duties and shall have such powers as the Board of Directors or the Chief Executive Officer may from time to time prescribe. In addition, the Treasurer shall perform such duties and have such powers as are incident to the office of treasurer, including without limitation the duty and power to keep and be responsible for all funds and securities of the Corporation, to deposit funds of the Corporation in depositories selected in accordance with these Bylaws, to disburse such funds as ordered by the Board of Directors, to make proper accounts of such funds and to render as required by the Board of Directors statements of all such transactions and of the financial condition of the Corporation.

The Assistant Treasurer shall perform such duties and possess such powers as the Board of Directors, the Chief Executive Officer or the Treasurer may from time to time prescribe. In the event of the absence, inability, or refusal to act of the Treasurer, the Assistant Treasurer (or if there shall be more than one, the Assistant Treasurers in the order determined by the Board of Directors) shall perform the duties and exercise the powers of the Treasurer.

- 3.12 <u>Salaries</u>. Officers of the Corporation shall be entitled to such salaries, compensation or reimbursement as shall be fixed or allowed from time to time by the Board of Directors.
- 3.13 <u>Delegation</u>. The Board of Directors may from time to time delegate the powers or duties of any officer to any other officer or agent, notwithstanding any provision hereof.

ARTICLE IV - CAPITAL STOCK

4.1 <u>Issuance of Stock</u>. Subject to the provisions of the Certificate of Incorporation, the whole or any part of any unissued balance of the authorized capital stock of the Corporation or the whole or any part of any unissued balance of the authorized capital stock of the

Corporation held in its treasury may be issued, sold, transferred or otherwise disposed of by vote of the Board of Directors in such manner, for such consideration and on such terms as the Board of Directors may determine; provided, that, other than in connection with the issuance of capital stock or securities convertible into capital stock by way of stock dividend or other distribution made to substantially all holders of the Company's capital stock, the Corporation shall not issue shares of its capital stock while its securities are listed on the Toronto Venture Exchange unless (i) the consideration for such shares has been fully paid in cash, or if not for cash, in property or past services that have a fair value not less than the value of such shares, as determined by the Board of Directors, and (ii) such shares are fully paid and non-assessable.

4.2 <u>Certificates of Stock.</u> The shares of the Corporation shall be represented by certificates, provided that the Board of Directors may provide by resolution or resolutions that some or all of any class or series of the Corporation's stock shall be uncertificated shares. Every holder of stock of the Corporation represented by certificates shall be entitled to have a certificate, in such form as may be prescribed by law and by the Board of Directors, representing the number and class of shares held by such holder registered in certificate form. Each such certificate shall be signed in a manner that complies with Section 158 of the DGCL. Each certificate for shares of stock which are subject to any restriction on transfer pursuant to the Certificate of Incorporation, these Bylaws, applicable securities laws or any agreement among any number of stockholders or among such holders and the Corporation shall have conspicuously noted on the face or back of the certificate either the full text of the restriction or a statement of the existence of such restriction.

If the Corporation shall be authorized to issue more than one class of stock or more than one series of any class, the powers, designations, preferences and relative participating, optional, or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights shall be set forth in full or summarized on the face or back of each certificate representing shares of such class or series of stock, provided that in lieu of the foregoing requirements there may be set forth on the face or back of each certificate representing shares of such class or series of stock a statement that the Corporation will furnish without charge to each stockholder who so requests a copy of the full text of the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualification, limitations or restrictions of such preferences and/or rights.

Within a reasonable time after the issuance or transfer of uncertificated shares, the Corporation shall send to the registered holder thereof a written notice containing the information required to be set forth or stated on certificates pursuant to Sections 151, 202(a), or 218 of the DGCL, with respect to Section 151 of the DGCL, a statement that the corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

4.3 <u>Transfers</u>. Shares of stock of the Corporation shall be transferrable in the manner prescribed by law and in these Bylaws. Transfers of shares of stock of the Corporation shall be made only on the books of the Corporation or by transfer agents designated to transfer shares of stock of the Corporation. Subject to applicable law, shares of stock represented by certificates

shall be transferred only on the books of the corporation by surrender to the Corporation or its transfer agent of the certificate representing such shares properly endorsed or accompanied by a written assignment or power of attorney properly executed, and with such proof of authority or the authenticity of signatures as the Corporation or its transfer agent may reasonably require. Except as may be otherwise required by law, by the Certificate of Incorporation or by these Bylaws, the Corporation shall be entitled to treat the record holder of stock as shown on its books as the owner of such stock for all purposes, including the payment of dividends and the right to vote with respect to such stock, regardless of any transfer, pledge or other disposition of such stock until the shares have been transferred on the books of the Corporation in accordance with the requirements of these Bylaws.

- 4.4 <u>Lost, Stolen or Destroyed Certificates</u>. The Corporation may issue a new certificate of stock in place of any previously issued certificate alleged to have been lost, stolen or destroyed, upon such terms and conditions as the Board of Directors may prescribe, including the presentation of reasonable evidence of such loss, theft or destruction and the giving of such indemnity and posting of such bond as the Board of Directors may require for the protection of the Corporation, any transfer agent or registrar.
- 4.5 Record Date. The Board of Directors may fix in advance a date as a record date for the determination of the stockholders entitled to notice of or to vote at any meeting of stockholders, or entitled to receive payment of any dividend or other distribution or allotment of any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action. Such record date shall not precede the date on which the resolution fixing the record date is adopted, and such record date shall not be more than sixty (60) nor less than ten (10) days before the date of such meeting, nor more than sixty (60) days prior to any other action to which such record date relates. If no record date is fixed, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day before the day on which notice is given, or, if notice is waived, at the close of business on the day before the day on which the meeting is held. If no record date is fixed, the record date for determining stockholders for any other purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating to such purpose.

A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.

4.6 <u>Regulations</u>. The issue, transfer, conversion and registration of shares of stock of the Corporation shall be governed by such regulations as the Board of Directors may establish.

ARTICLE V - INDEMNIFICATION

5.1 Actions, Suits and Proceedings Other than by or in the Right of the Corporation. The Corporation shall indemnify each person who was or is a party or is threatened to be made or party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the Corporation), by reason of the fact that such person is or was, or has agreed to become, a

Director or officer of the Corporation, or is, or was serving, or has agreed to serve, at the request of the Corporation, as a director, officer, trustee, general partner, managing member, fiduciary, board of directors' committee member, employee or agent of, or in a similar capacity with, another corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other enterprise (all such persons being referred to hereafter as an "Indemnitee"), or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees), liabilities, losses, judgments, fines, excise taxes and penalties arising under the Employee Retirement Income Security Act of 1974, as amended, and amounts paid in settlement actually and reasonably incurred by or on behalf of Indemnitee in connection with such action, suit or proceeding and any appeal therefrom, if Indemnitee acted in good faith and in a manner that Indemnitee reasonably believed to be in, or not opposed to, the best interests of the Corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his or her conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that Indemnitee did not act in good faith and in a manner that Indemnitee reasonably believed to be in, or not opposed to, the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that his or her conduct was unlawful. Notwithstanding anything to the contrary in this Article V, and except as set forth in Section 5.7 below, the Corporation shall not indemnify an Indemnitee seeking indemnification in connection with a proceeding (or part thereof) initiated by the Indemnitee unless the initiation thereof was approved by the Board of Directors of the Corporation.

5.2 Actions or Suits by or in the Right of the Corporation. The Corporation shall indemnify any Indemnitee who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the Corporation to procure a judgment in its favor by reason of the fact that Indemnitee is or was, or has agreed to become, a Director or officer of the Corporation, or is or was serving, or has agreed to serve, at the request of the Corporation, as a director, officer, trustee, general partner, managing member, fiduciary, board of directors' committee member, employee, or agent of, or in a similar capacity with, another corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other enterprise, or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees) actually and reasonably incurred by or on behalf of Indemnitee in connection with such action, suit or proceeding and any appeal therefrom, if Indemnitee acted in good faith and in a manner that Indemnitee reasonably believed to be in, or not opposed to, the best interests of the Corporation, except that no indemnification shall be made in respect of any claim, issue or matter as to which Indemnitee shall have been adjudged to be liable to the Corporation, unless, and only to the extent, that the Court of Chancery of the State of Delaware or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of such liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnity for such expenses (including attorneys' fees) which the Court of Chancery of the State of Delaware or such other court shall deem proper.

5.3 <u>Indemnification for Expenses of Successful Party</u>. Notwithstanding the other provisions of this <u>Article V</u>, to the extent that an Indemnitee has been successful, on the merits or otherwise, in defense of any action, suit, investigation or proceeding referred to in <u>Section 5.1</u> or <u>Section 5.2</u> of this <u>Article V</u>, or in defense of any claim, issue, or matter therein, or on appeal

from any such action, suit, investigation or proceeding, Indemnitee shall be indemnified against all expenses (including attorneys' fees) actually and reasonably incurred by or on behalf of Indemnitee in connection therewith.

- 5.4 Notification and Defense of Claim. As a condition precedent to an Indemnitee's right to be indemnified, such Indemnitee must notify the Corporation in writing as soon as practicable of any action, suit, proceeding or investigation involving such Indemnitee for which indemnity will or could be sought. With respect to any action, suit, proceeding or investigation of which the Corporation is so notified, the Corporation will be entitled to participate therein at its own expense and/or to assume the defense thereof at its own expense, with legal counsel reasonably acceptable to Indemnitee. After notice from the Corporation to Indemnitee of its election so to assume such defense, the Corporation shall not be liable to Indemnitee for any legal or other expense subsequently incurred by Indemnitee in connection with such action, suit, proceeding or investigation, other than as provided below in this Section 5.4. Indemnitee shall have the right to employ his or her own counsel in connection with such action, suit, proceeding or investigation, but the fees and expenses of such counsel incurred after notice from the Corporation of its assumption of the defense thereof shall be at the expense of Indemnitee unless (i) the employment of counsel by Indemnitee has been authorized by the Corporation; (ii) counsel to Indemnitee shall have reasonably concluded that there may be a conflict of interest or position on any significant issue between the Corporation and Indemnitee in the conduct of the defense of such action, suit, proceeding or investigation; or (iii) the Corporation shall not in fact have employed counsel to assume the defense of such action, suit, proceeding or investigation, in each of which cases the fees and expenses of counsel for Indemnitee shall be at the expense of the Corporation, except as otherwise expressly provided by this Article V. The Corporation shall not be entitled, without the consent of Indemnitee, to assume the defense of any claim brought by or in the right of the Corporation or as to which counsel for Indemnitee shall have reasonably reached the conclusion provided for in clause (ii) above. The Corporation shall not be required to indemnify an Indemnitee under this Article V for any amounts paid in settlement of any action, suit, proceeding or investigation effected without its written consent. The Corporation shall not settle any action, suit, proceeding or investigation in any manner which would impose any penalty or limitation on an Indemnitee without the Indemnitee's written consent. Neither the Corporation nor an Indemnitee will unreasonably withhold or delay its consent to any proposed settlement.
- 5.5 <u>Advancement of Expenses</u>. Subject to the provisions of <u>Section 5.6</u> of this <u>Article V</u>, in the event of any threatened or pending action, suit, proceeding or investigation of which the Corporation received notice under this <u>Article V</u> and with regard to which the Corporation does not assume the defense pursuant to <u>Section 5.4</u> of this <u>Article V</u>, any expenses (including attorneys' fees) incurred by or on behalf of Indemnitee in defending a civil or criminal action, suit, proceeding or investigation or any appeal therefrom shall be paid by the Corporation in advance of the final disposition of such matter; provided, however, that the payment of such expenses incurred by or on behalf of Indemnitee in advance of the final disposition of such matter shall be made only upon receipt of an undertaking by or on behalf of Indemnitee to repay all amounts so advanced in the event that it shall ultimately be determined that Indemnitee is not entitled to be indemnified by the Corporation as authorized in this <u>Article V</u>. Such undertaking may be accepted without reference to the financial ability of such person to make such repayment.

5.6 Procedure for Indemnification. In order to obtain indemnification or advancement of expenses pursuant to Section 5.1, 5.2, 5.3 or 5.5 of this Article V, an Indemnitee shall submit to the Corporation a written request, including in such request such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification or advancement of expenses. Within thirty (30) days of receipt by the Corporation of such written request for indemnification, subject to the provision of this Article V, the Corporation shall advance to Indemnitee all expenses (including attorneys' fees) actually and reasonably incurred by or on behalf of Indemnitee in connection with the relevant action, suit, proceeding or investigation, as set forth in such written request. As soon as practicable (but in no event later than sixty (60) days) after final disposition of the relevant action, suit, proceeding or investigation, a determination with respect to Indemnitee's entitlement to indemnification shall be made in the specific case: (i) if a Change of Control shall not have occurred, (A) by a majority vote of the Disinterested Directors, even though less than a quorum of the Board of Directors, or (C) if there are no such Disinterested Directors or, if such Disinterested Directors so direct, by Independent Counsel in a written opinion to the Board of Directors, a copy of which shall be delivered to Indemnitee; or (ii) if a Change of Control shall have occurred, by Independent Counsel in a written opinion to the Board, a copy of which shall be delivered to Indemnitee.

For purposes of this <u>Section 5.6</u>, the following terms shall be defined as follows:

"Change of Control" means any one of the follow circumstances occurring after the date of filing of this Certificate of Incorporation with the Secretary of State of the State of Delaware: (i) there shall have occurred an event required to be reported with respect to the Corporation in response to Item 6(e) of Schedule 14A of Regulation 14A (or in response to any similar item or any similar schedule or form) under the Exchange Act, regardless of whether the Corporation is then subject to such reporting requirement; (ii) any "person" or "group" (as such terms are used in Section 13(d) and 14(d) of the Exchange Act) shall have become, without prior approval of the Corporation's Board of Directors by approval of at least two-thirds of the Continuing Directors, the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of outstanding capital stock of the Corporation representing fifteen percent (15%) or more of the combined voting power of the Corporation's then outstanding capital stock (provided that, for the purpose of this Section 13(a) (ii), the term "person" shall exclude (x) the Corporation, (y) any trustee or other fiduciary holding stock of the Corporation under an employee benefit plan of the Corporation, and (z) any corporation owned, directly or indirectly, by the stockholders of the Corporation in substantially the same proportions as their ownership of stock of the Corporation); (iii) there occurs a merger or consolidation of the Corporation with any other entity, other than a merger or consolidation which would result in the voting capital stock of the Corporation outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity) more than fifty-one percent (51%) of the combined voting power of the voting securities of the surviving entity outstanding immediately after such merger or consolidation and with the power to elect at least a majority of the board of directors or other governing body of such surviving entity; (iv) all or substantially all of the assets of the Corporation are sold or disposed of in a transaction or series of related transactions; (v) the

approval of the stockholders of the Corporation of a complete liquidation of the Corporation; or (vi) the Continuing Directors cease for any reason to constitute at least a majority of the members of the Board of Directors.

"Continuing Director" means (i) each member of the Board of Directors on the date of filing of this Certificate of Incorporation with the Secretary of State of the State of Delaware; or (ii) any new Director whose election or nomination for election by the Corporation's stockholders was approved by a vote of at least two-thirds of the Directors then still in office who were Directors on the date of filing of this Certificate of Incorporation with the Secretary of State of the State of Delaware.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Independent Counsel" means a law firm, or a member of a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five (5) years has been, retained to represent: (i) the Corporation or an Indemnitee in any matter material to either such party (other than with respect to matters concerning an Indemnitee or other indemnitees under any indemnification agreement with the Corporation); or (ii) any other party to the action, suit, proceeding or investigation giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Corporation or an Indemnitee in an action to determine an Indemnitee's rights to indemnification.

- 5.7 Remedies. The right to indemnification or advancement of expenses as granted by this Article V shall be enforceable by an Indemnitee in any court of competent jurisdiction. Unless otherwise required by law, the burden of proving that an Indemnitee is not entitled to indemnification or advancement of expenses under this Article V shall be on the Corporation. Neither the failure of the Corporation to have made a determination prior to the commencement of such action that indemnification is proper in the circumstances because an Indemnitee has met the applicable standard of conduct, nor an actual determination by the Corporation pursuant to Section 5.6 of this Article V that an Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that an Indemnitee has not met the applicable standard of conduct. An Indemnitee's expenses (including attorneys' fees) incurred in connection with successfully establishing his right to indemnification, in whole or any part, in any such proceeding shall also be indemnified by the Corporation.
- 5.8 <u>Subsequent Amendment</u>. No amendment, termination or repeal of this <u>Article V</u> or of the relevant provisions of the DGCL or any other applicable laws shall affect or diminish in any way the rights of any Indemnitee to indemnification under the provisions hereof with respect to any action, suit, proceeding or investigation arising out of or relating to any actions, transactions or facts occurring prior to the final adoption of such amendment, termination or repeal.
- 5.9 Other Rights. The indemnification and advancement of expenses provided by this Article V shall not be deemed exclusive of any other rights to which an Indemnitee seeking indemnification or advancement of expenses may be entitled under any law (common or

statutory), agreement or vote of stockholders or Disinterested Directors or otherwise, both as to action in an Indemnitee's official capacity and as to action in any other capacity while holding office for the Corporation, and shall continue as to an Indemnitee who has ceased to be a Director or officer, and shall inure to the benefit of the estate, heirs, executors and administrators of Indemnitee. Nothing contained in this Article V shall be deemed to prohibit, and the Corporation is specifically authorized to enter into, agreements with officers and Directors providing indemnification rights and procedures different from those set forth in this Article V. In addition, the Corporation may, to the extent authorized from time to time by its Board of Directors, grant indemnification rights to other employees or agents of the Corporation or other persons serving the Corporation and such rights may be equivalent to, or greater or less than, those set forth in this Article V.

- 5.10 <u>Partial Indemnification</u>. If an Indemnitee is entitled under any provision of this <u>Article V</u> to indemnification by the Corporation for some or a portion of the expenses (including attorneys' fees), judgments, fines or amounts paid in settlement actually and reasonably incurred by or on behalf of the Indemnitee in connection with any action, suit, proceeding or investigation and any appeal therefrom but not, however, for the total amount thereof, the Corporation shall nevertheless indemnify the Indemnitee for the portion of such expenses (including attorneys' fees), judgments, fines or amounts paid in settlement to which the Indemnitee is entitled.
- 5.11 <u>Insurance</u>. The Corporation may purchase and maintain insurance, at its expense, to protect itself and any Director, officer, trustee, general partner, managing member, fiduciary, Board of Directors' committee member, employee or agent of the Corporation or another corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other enterprise (including any employee benefit plan) against any expense, liability or loss incurred by him or her in any such capacity, or arising out of his or her status as such, whether or not the Corporation would have the power to indemnify such person against such expense, liability or loss under the DGCL.
- 5.12 Merger or Consolidation. If the Corporation is merged into or consolidated with another corporation and the Corporation is not the surviving corporation, the surviving corporation shall assume the obligations of the Corporation under this Article V with respect to any action, suit, proceeding or investigation arising out of or relating to any actions, transactions or facts occurring prior to the date of such merger or consolidation.
- 5.13 <u>Savings Clause</u>. If this <u>Article V</u> or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Corporation shall nevertheless indemnify each Indemnitee as to any expenses (including attorneys' fees), judgments, fines and amounts paid in settlement in connection with any action, suit, proceeding or investigation, whether civil, criminal or administrative, including an action by or in the right of the Corporation, to the fullest extent permitted by any applicable portion of this <u>Article V</u> that shall not have been invalidated and to the fullest extent permitted by applicable law.
- 5.14 <u>Definitions</u>. Terms used herein and defined in Section 145(h) and Section 145(i) of the DGCL shall have the respective meanings assigned to such terms in such Section 145(h) and Section 145(i).

5.15 <u>Subsequent Legislation</u>. If the DGCL is amended after adoption of this <u>Article V</u> to expand further the indemnification permitted to Indemnitees, then the Corporation shall indemnify such persons to the fullest extent permitted by the DGCL, as so amended.

ARTICLE VI - GENERAL PROVISIONS

- 6.1 <u>Fiscal Year</u>. Except as from time to time otherwise designated by the Board of Directors, the fiscal year of the Corporation shall be the calendar year.
 - 6.2 Corporate Seal. The corporate seal shall be in such form as shall be approved by the Board of Directors.
- 6.3 <u>Waiver of Notice</u>. Whenever any notice whatsoever is required to be given by law, by the Certificate of Incorporation, or by these Bylaws, a written waiver signed by the person entitled to notice, or a waiver by electronic transmission by the person entitled to notice, whether before, at, or after the time stated in such waiver, shall be deemed equivalent to notice required to be given to such person. Neither the business nor the purpose of any meeting need be specified in any such waiver. Attendance of persons at a meeting shall constitute waiver of notice of such meeting, except when the person attends a meeting for the express purpose of objecting at the beginning of the meeting to the transaction of any business because the meeting is not lawfully called or convened.
- 6.4 <u>Voting of Securities</u>. Except as the Board of Directors may otherwise designate, the Chief Executive Officer, the President or the Treasurer may waive notice of, and act as, or appoint any person or persons to act as, proxy or attorney-in-fact for the Corporation (with or without power of substitution) at, any meeting of stockholders or securityholders of any other entity, the securities of which may be held by the Corporation.
- 6.5 Evidence of Authority. A certificate by the Secretary, an Assistant Secretary or a temporary Secretary as to any action taken by the stockholders, the Board of Directors, a committee or any officer or representative of the Corporation shall as to all persons who rely on the certificate in good faith be conclusive evidence of such action.
- 6.6 <u>Certificate of Incorporation</u>. All references in these Bylaws to the Certificate of Incorporation shall be deemed to refer to the Certificate of Incorporation of the Corporation, as amended and in effect from time to time.
- 6.7 <u>Severability</u>. Any determination that any provision of these Bylaws is for any reason inapplicable, illegal or ineffective shall not affect or invalidate any other provision of these Bylaws.
- 6.8 <u>Pronouns</u>. All pronouns used in these Bylaws shall be deemed to refer to the masculine, feminine, or neuter, singular or plural, as the identity of the person or persons may require.

ARTICLE VII – AMENDMENTS

7.1 By the Board of Directors. These Bylaws may be altered, amended or repealed or new bylaws may be adopted by the affirmative vote of a majority of the Directors present at any regular or special meeting of the Board of Directors at which a quorum is present. By the Stockholders. Subject to the Certificate of Incorporation, these Bylaws may be altered, amended or repealed or new bylaws may be adopted by the holders of at least a majority of the total voting power of the issued and outstanding shares of the Corporation' capital stock, voting together as a single class.

COHBAR, INC.

as the Corporation

and

CST TRUST COMPANY

as the Warrant Agent

WARRANT INDENTURE Providing for the Issue of

5,625,000 IPO Unit Warrants 1,350,000 Put Unit Warrants and Up to 393,750 Compensation Unit Warrants

Dated as of January 6, 2015

TABLE OF CONTENTS

		Page No.
ARTICLE 1 I	NTERPRETATION	6
1.1	Definitions	6
1.2	Gender and Number	10
1.3	Headings, Etc.	10
1.4	Day not a Business Day	10
1.5	Time of the Essence	10
1.6	Monetary References	10
1.7	Applicable Law	11
ARTICLE 2 I	SSUE OF WARRANTS	11
2.1	Creation and Issue of Warrants	11
2.2	Terms of Warrants	11
2.3	Warrantholder not a Shareholder	12
2.4	Warrants to Rank Pari Passu	12
2.5	Form of Warrants, Certificated Warrants	12
2.6	Book Entry Only Warrants	12
2.7	Warrant Certificate	14
2.8	Legends	16
2.9	Register of Warrants	17
2.10	Issue in Substitution for Warrant Certificates Lost, etc.	18
2.11	Exchange of Warrant Certificates	18
2.12	Transfer and Ownership of Warrants	19
2.13	Cancellation of Surrendered Warrants	20
ARTICLE 3 EXERCISE OF WARRANTS		20
3.1	Right of Exercise	20
3.2	Warrant Exercise	20
3.3	Prohibition on Exercise by U.S. Persons; Legended Certificates; Special Conditions to Exercise of Put Unit	
	Warrants.	22
3.4	Transfer Fees and Taxes	24
3.5	Warrant Agency	24
3.6	Effect of Exercise of Warrant Certificates	25
3.7	Partial Exercise of Warrants; Fractions	25
3.8	Expiration of Warrants	25
3.9	Accounting and Recording	25
3.10	Securities Restrictions	26
ARTICLE 4	ADJUSTMENT OF NUMBER OF COMMON SHARES AND EXERCISE PRICE	26
4.1	Adjustment of Number of Common Shares and Exercise Price	26
4.2	Entitlement to Common Shares on Exercise of Warrant	30
4.3	No Adjustment for Certain Transactions	30
4.4	Determination by Independent Firm	30
4.5	Proceedings Prior to any Action Requiring Adjustment	30
4.6	Certificate of Adjustment	30
4.7	Notice of Special Matters	31
4.8	No Action after Notice	31
4.9	Other Action	31

4.10	Protection of Warrant Agent	31
4.11	Participation by Warrantholder	32
ARTICLE 5 B	RIGHTS OF THE CORPORATION AND COVENANTS	32
5.1	Optional Purchases by the Corporation	32
5.2	General Covenants	32
5.3	Warrant Agent's Remuneration and Expenses	33
5.4	Performance of Covenants by Warrant Agent	33
5.5	Enforceability of Warrants	34
ARTICI E 6 E	ENFORCEMENT	34
6.1	Suits by Registered Warrantholders	34
6.2	Suits by the Corporation	34
6.3	Immunity of Shareholders, etc.	34
6.4	Waiver of Default	34
ADTICLE 7 N	MEETINGS OF DECISTEDED WADDANTHOLDEDS	25
7.1	MEETINGS OF REGISTERED WARRANTHOLDERS Right to Convene Meetings	35 35
7.1	Notice	
7.2	Chairman	35
7.3 7.4		35 35
7.4	Quorum Power to Adjourn	33
7.5 7.6	Show of Hands	36
7.0 7.7	Poll and Voting	36
7.7		36
	Regulations Comparation and Warrant A cent May be Represented	37
7.9 7.10	Corporation and Warrant Agent May be Represented Powers Exercisable by Extraordinary Resolution	37
7.10	Meaning of Extraordinary Resolution	38
7.11	Powers Cumulative	38
7.12	Minutes	38
7.13		38
7.14	Instruments in Writing Distribution Effect of Resolutions	39
7.13	Binding Effect of Resolutions Holdings by Corporation Disregarded	39
ADTICLE 0.0	NUDDI EMENTAL INDENTUDEO	20
	SUPPLEMENTAL INDENTURES Provision for Symplemental Indentures for Cortain Dymasses	39 39
8.1	Provision for Supplemental Indentures for Certain Purposes	
8.2	Successor Entities	40
ARTICLE 9 C	CONCERNING THE WARRANT AGENT	40
9.1	Trust Indenture Legislation	40
9.2	Rights and Duties of Warrant Agent	40
9.3	Evidence, Experts and Advisers	41
9.4	Documents, Monies, etc. Held by Warrant Agent	42
9.5	Actions by Warrant Agent to Protect Interest	43
9.6	Warrant Agent Not Required to Give Security	43
9.7	Protection of Warrant Agent	43
9.8	Replacement of Warrant Agent; Successor by Merger	44
9.9	Conflict of Interest	44
9.10	Acceptance of Agency	45
9.11	Warrant Agent Not to be Appointed Receiver	45
9.12	Warrant Agent Not Required to Give Notice of Default	45

9.13	Anti-Money Laundering	45
9.14	Compliance with Privacy Code	46
9.15	Securities Exchange Commission Certification	46
ARTICLE 10 GENERAL		47
10.1	Notice to the Corporation and the Warrant Agent	47
10.2	Notice to Registered Warrantholders	47
10.3	Ownership of Warrants	48
10.4	Counterparts	48
10.5	Satisfaction and Discharge of Indenture	48
10.6	Provisions of Indenture and Warrants for the Sole Benefit of Parties and Registered Warrantholders	49
10.7	Common Shares or Warrants Owned by the Corporation or its Subsidiaries—Certificate to be Provided	49
10.8	Severability	49
10.9	Force Majeure	49
10.10	Assignment, Successors and Assigns	50
10.11	Rights of Rescission and Withdrawal for Holders	50

Schedule "A"—Form of Warrant Certificate

Schedule "B"—Form of Transfer

Schedule "C"—Exercise Form
Schedule "D"—Form of Warrant Exercise Certification

WARRANT INDENTURE

THIS WARRANT INDENTURE is dated as of January 6, 2015.

BETWEEN:

COHBAR, INC., a corporation incorporated under the laws of the State of Delaware

(the "Corporation"),

AND:

CST TRUST COMPANY, a trust company existing under the laws of Canada and authorized to carry on business in all provinces of Canada

(the "Warrant Agent")

WHEREAS Haywood Securities Inc. is acting as agent in connection with the Corporation's issuance and sale by means of a prospectus of 11,250,000 of the Corporation's units (individually a "**Unit**" and collectively, the "**Units**");

AND WHEREAS each Unit, issuable at a price of US \$1.00 per Unit, consists of one (1) Common Share (as hereinafter defined) and one half (1/2) of one Common Share purchase warrant (an "**IPO Unit Warrant**");

AND WHEREAS in connection with the Offering the Corporation shall issue up to 787,500 compensation options (the "Compensation Options"), each entitling the holder thereof to purchase, at any time prior to 5:00 pm (Toronto Time) on the date that is eighteen (18) months following the Issue Date (as hereafter defined), one (1) Unit (a "Compensation Unit"), each such Compensation Unit consisting of one Common Share and one half (1/2) of one Common Share purchase warrant (a "Compensation Unit Warrant");

AND WHEREAS concurrently with the completion of the Offering the Corporation will issue and sell 2,700,000 Units (the "**Put Units**") to certain existing investors pursuant to the exercise by the Corporation of Put Rights granted by such investors to the Corporation, each such Put Unit consisting of one Common Share and one half (1/2) of one Common Share Purchase Warrant (a "**Put Unit Warrant"**);

AND WHEREAS pursuant to this Indenture, each Warrant (as hereunder defined) shall, subject to adjustment, entitle the holder thereof to acquire one Common Share upon payment of the Exercise Price upon the terms and conditions herein set forth;

AND WHEREAS all acts and deeds necessary have been done and performed to make the Warrants, when created and issued as provided in this Indenture, legal, valid and binding upon the Corporation with the benefits and subject to the terms of this Indenture;

AND WHEREAS the foregoing recitals are made as representations and statements of fact by the Corporation and not by the Warrant Agent;

NOW THEREFORE, in consideration of the premises and mutual covenants hereinafter contained and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Corporation hereby appoints the Warrant Agent as warrant agent to hold the rights, interests and benefits contained herein for and on behalf of those persons who from time to time become the holders of Warrants issued pursuant to this Indenture and the parties hereto agree as follows:

ARTICLE 1 INTERPRETATION

1.1 Definitions

In this Indenture, including the recitals and schedules hereto, and in all indentures supplemental hereto:

- (a) "Acceleration Notice" means the written notice of acceleration deliverable to holders for Warrants upon the Corporation's exercise of the Acceleration Right;
- (b) "Acceleration Right" means the right of the Corporation to accelerate the Expiry Time to a date that is not the less than 30 days following delivery of the Acceleration Notice if, at any time after the Common Shares are first traded on the TSXV, the volume weighted average trading price of the Common Shares on the TSXV, or if the Common Shares are not then listed on the TSXV, on such other stock exchange on which the Common Shares are principally traded, equals or exceeds \$3.00 for 20 consecutive Trading Days;
- (c) "Adjustment Period" means the period from the Effective Date up to and including the Expiry Time;
- (d) "Applicable Legislation" means any statute of Canada or a province thereof, and the regulations under any such named or other statute, relating to warrant indentures or to the rights, duties and obligations of warrant agents under warrant indentures, to the extent that such provisions are at the time in force and applicable to this Indenture;
- (e) "Auditors" means Marcum LLP or such other firm of chartered accountants duly appointed as auditors of the Corporation, from time to time;
- (f) "Authenticated" means (a) with respect to the issuance of a Warrant Certificate, one which has been duly signed by the Corporation and authenticated by manual signature of an authorized officer of the Warrant Agent, (b) with respect to the issuance of an Uncertificated Warrant, one in respect of which the Warrant Agent has completed all Internal Procedures such that the particulars of such Uncertificated Warrant as required by Section 2.7 are entered in the register of holders of Warrants, "Authenticate", "Authenticating" and "Authentication" have the appropriate correlative meanings;
- (g) "Book Entry Only Participants" means institutions that participate directly or indirectly in the Depository's book entry registration system for the Warrants;
- (h) "Book Entry Only Warrants" means Warrants that are to be held only by or on behalf of the Depository;
- "Business Day" means any day other than Saturday, Sunday or a statutory or civic holiday, or any other day on which banks are not open for business in the City of Vancouver, British Columbia or Los Angeles, California and shall be a day on which the TSXV is open for trading;

- (j) "CDS Global Warrants" means Warrants representing all or a portion of the aggregate number of Warrants issued in the name of the Depository represented by an Uncertificated Warrant, or if requested by the Depository or the Corporation, by a Warrant Certificate:
- (k) "Certificated Warrant" means a Warrant evidenced by a writing or writings substantially in the form of Schedule "A", attached hereto:
- (1) "Common Shares" means, subject to Article 4, fully paid and non-assessable shares in the common stock of the Corporation as presently constituted;
- (m) "Counsel" means a barrister and/or solicitor or a firm of barristers and/or solicitors retained by the Warrant Agent or retained by the Corporation, which may or may not be counsel for the Corporation;
- (n) "Current Market Price" of the Common Shares at any date means the weighted average of the trading price per Common Share on the TSXV, or, if the Common Shares are not then listed on the TSXV, on such other stock exchange on which the Common Shares are principally traded (as determined by the Corporation's Board of Directors) or, if the Common Shares are not then listed on any stock exchange, in the over-the-counter market, during a period of 20 consecutive Trading Days ending not more than five (5) Business Days prior to such date, provided that the weighted average price shall be determined by dividing the aggregate sale price of all such Common Shares sold on such exchange or market, as the case may be, during such 20 consecutive Trading Days by the total number of such Common Shares so sold;
- (o) "Depository" means CDS Clearing and Depository Services Inc. or such other person as is designated in writing by the Corporation to act as depository in respect of the Warrants;
- (p) "Dividends" means any dividends paid by the Corporation;
- (q) "Effective Date" means the date of this Indenture;
- (r) "Exchange Rate" means the number of Common Shares subject to the right of purchase under each Warrant;
- (s) "Exercise Date" means, in relation to a Warrant, the Business Day on which such Warrant is validly exercised or deemed to be validly exercised in accordance with Article 3 hereof;
- (t) "Exercise Notice" has the meaning set forth in Section 3.2(1);
- (u) "Exercise Price" at any time means the price at which a whole Common Share may be purchased by the exercise of a whole Warrant, which is initially \$2.00 per Common Share, payable in immediately available U.S. funds, subject to adjustment in accordance with the provisions of Section 4.1;

- (v) "Expiry Time" means 5:00 p.m. (Toronto time) on January 6, 2017; provided that if at any time after the Common Shares are first traded on the TSXV, the volume weighted average trading price of the Common Shares on the TSXV, or if the Common Shares are not then listed on the TSXV, on such other stock exchange on which the Common Shares are principally traded, equals or exceeds \$3.00 for 20 consecutive Trading Days, the Corporation shall be entitled, at the option of the Corporation, to exercise the Acceleration Right by delivering an Acceleration Notice to the Warrantholders within five (5) Business Days to the effect that the Warrants will be expire at 5:00 p.m. (Toronto time) on the date specified in such notice, provided that such date shall not be less than 30 days following the date of such notice;
- (w) "Extraordinary Resolution" has the meaning set forth in 7.11(1);
- (x) "Internal Procedures" means in respect of the making of any one or more entries to, changes in or deletions of any one or more entries in the register at any time (including without limitation, original issuance or registration of transfer of ownership) the minimum number of the Warrant Agent's internal procedures customary at such time for the entry, change or deletion made to be complete under the operating procedures followed at the time by the Warrant Agent, it being understood that neither preparation and issuance shall constitute part of such procedures for any purpose of this definition;
- (y) "Issue Date" means January 6, 2015;
- (z) "person" means an individual, body corporate, partnership, trust, warrant agent, executor, administrator, legal representative or any unincorporated organization;
- (aa) "register" means the one set of records and accounts maintained by the Warrant Agent pursuant to Section 2.9:
- (bb) "Registered Warrantholders" means the persons who are registered owners of Warrants as such names appear on the register, and for greater certainty, shall include the Depository as well as the holders of Uncertificated Warrants appearing on the register of the Warrant Agent;
- (cc) "Registration Statement" means the Corporation's registration statement on Form S-1 (SEC File No.: 333-200033) filed with the SEC under the U.S. Securities Act, registering the Common Shares issuable upon exercise of the IPO Warrants and the Compensation Unit Warrants;
- (dd) "Regulation D" means Regulation D as promulgated by the United States Securities and Exchange Commission under the U.S. Securities Act;
- (ee) "Regulation S" means Regulation S as promulgated by the United States Securities and Exchange Commission under the U.S. Securities Act:
- (ff) "SEC" means the United States Securities and Exchange Commission;
- (gg) "Shareholders" means holders of Common Shares;
- (hh) "this Warrant Indenture", "this Indenture", "this Agreement", "hereto" "herein", "hereby", "hereof" and similar expressions mean and refer to this Indenture and any

- indenture, deed or instrument supplemental hereto; and the expressions "Article", "Section", "subsection" and "paragraph" followed by a number, letter or both mean and refer to the specified article, section, subsection or paragraph of this Indenture;
- (ii) "Trading Day" means, with respect to the TSXV, a day on which such exchange is open for the transaction of business and with respect to another exchange or an over-the-counter market means a day on which such exchange or market is open for the transaction of business;
- (jj) "TSXV" means the TSX Venture Exchange
- (kk) "Uncertificated Warrant" means any Warrant which is not a Certificated Warrant;
- (ll) "United States" means the United States of America, its territories and possessions, any state of the United States, and the District of Columbia;
- (mm)"U.S. Exchange Act" means the United States Securities Exchange Act of 1934, as amended;
- (nn) "U.S. Person" has the meaning set forth in Rule 902(k) of Regulation S;
- (00) "U.S. Prospectus" means a prospectus meeting the requirements of Section 10(a) of the U.S. Securities Act;
- (pp) "U.S. Securities Act" means the *United States Securities Act of 1933*, as amended;
- (qq) "U.S. Warrantholder" means any Warrantholder that is a U.S. Person, acquired Warrants in the United States or for the account or benefit of any U.S. Person or Person in the United States;
- (rr) "Warrant Exercise Certification" means the warrant exercise certification attached hereto as Schedule "D";
- (ss) "Warrants" means the IPO Unit Warrants, the Put Unit Warrants and the Compensation Unit Warrants created by and authorized by and issuable under this Indenture, to be issued and countersigned hereunder as a Certificated Warrant and /or Uncertificated Warrant held through the book entry registration system on a no certificate issued basis, entitling the holder or holders thereof to purchase Common Shares at the Exercise Price prior to the Expiry Time and, where the context so requires, also means the warrants issued and Authenticated hereunder, whether by way of Warrant Certificate or Uncertificated Warrant;
- (tt) "Warrant Agency" means the principal office of the Warrant Agent in the City of Vancouver or such other place as may be designated in accordance with Section 3.5;
- (uu) "Warrant Agent" means CST Trust Company, in its capacity as warrant agent of the Warrants, or its successors from time to time:
- (vv) "Warrant Certificate" means a certificate, substantially in the form set forth in Schedule "A" hereto, to evidence those Warrants that will be evidenced by a certificate;

- (ww) "Warrantholders", or "holders" without reference to Warrants, means the warrantholders as and in respect of Warrants registered in the name of the Depository and includes owners of Warrants who beneficially hold securities entitlements in respect of the Warrants through a Book Entry Only Participant or means, at a particular time, the persons entered in the register hereinafter mentioned as holders of Warrants outstanding at such time;
- (xx) "Warrantholders' Request" means an instrument signed in one or more counterparts by Registered Warrantholders entitled to acquire in the aggregate not less than 50% of the aggregate number of Common Shares which could be acquired pursuant to all Warrants then unexercised and outstanding, requesting the Warrant Agent to take some action or proceeding specified therein; and "written order of the Corporation", "written request of the Corporation", "written consent of the "Corporation" and "certificate of the Corporation" mean, respectively, a written order, request, consent and certificate signed in the name of the Corporation by any two duly authorized signatories of the Corporation and may consist of one or more instruments so executed.

1.2 Gender and Number

Words importing the singular number or masculine gender shall include the plural number or the feminine or neuter genders, and vice versa.

1.3 Headings, Etc.

The division of this Indenture into Articles and Sections, the provision of a Table of Contents and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Indenture or of the Warrants.

1.4 Day not a Business Day

If any day on or before which any action or notice is required to be taken or given hereunder is not a Business Day, then such action or notice shall be required to be taken or given on or before the requisite time on the next succeeding day that is a Business Day.

1.5 Time of the Essence

Time shall be of the essence of this Indenture.

1.6 Monetary References

Whenever any amounts of money are referred to herein, such amounts shall be deemed to be in lawful money of the United States unless otherwise expressed.

1.7 Applicable Law

This Indenture, the Warrants, the Warrant Certificates (including all documents relating thereto, which by common accord have been and will be drafted in English) shall be construed in accordance with the laws of the Province of British Columbia and the federal laws applicable therein and shall be treated in all respects as contracts. Each of the parties hereto, which shall include the Warrantholders, irrevocably attorns to the exclusive jurisdiction of the courts of the Province of British Columbia with respect to all matters arising out of this Indenture and the transactions contemplated herein.

ARTICLE 2 ISSUE OF WARRANTS

2.1 Creation and Issue of Warrants

A maximum of 5,625,000 IPO Unit Warrants, 1,350,000 Put Unit Warrants, and 393,750 Compensation Unit Warrants (subject to adjustment as herein provided) are hereby created and authorized to be issued in accordance with the terms and conditions hereof. By written order of the Corporation, the Warrant Agent shall deliver Warrant Certificates to Registered Warrantholders and record the name of the Registered Warrantholders on the Warrant register. Registration of interests in Warrants held by the Depository may be evidenced by a position appearing on the register for Warrants of the Warrant Agent for an amount representing the aggregate number of such Warrants outstanding from time to time.

2.2 Terms of Warrants

- (a) Subject to the applicable conditions for exercise set out in Article 3 having been satisfied and subject to adjustment in accordance with Section 4.1, each Warrant shall entitle the holder thereof, upon exercise at any time after the Issue Date and prior to the Expiry Time, to acquire one Common Share upon payment of the Exercise Price.
- (b) No fractional Warrants shall be issued or otherwise provided for hereunder and Warrants may only be exercised in a sufficient number to acquire whole numbers of Common Shares.
- (c) Each Warrant shall entitle the holder thereof to such other rights and privileges as are set forth in this Indenture.
- (d) The number of Common Shares which may be purchased pursuant to the Warrants and the Exercise Price therefor shall be adjusted upon the events and in the manner specified in Section 4.1.
- (e) If at any time after the Common Shares are first traded on the TSXV, the volume weighted average trading price of the Common Shares on the TSXV, or if the Common Shares are not then listed on the TSXV, on such other stock exchange on which the Common Shares are principally traded, equals or exceeds \$3.00 for 20 consecutive Trading Days, the Corporation shall be entitled, at the option of the Corporation, to exercise the Acceleration Right by delivering an Acceleration Notice to the Warrantholders within five (5) Business Days. An Acceleration Notice shall be delivered to each Warrantholder in the manner in Section 10.1. Upon the Corporation's exercise of the Acceleration Right the Corporation shall issue a press release to the effect that the Warrants will expire at 5:00 p.m. (Toronto time) on the date specified in the Acceleration Notice.

2.3 Warrantholder not a Shareholder

Except as may be specifically provided herein, nothing in this Indenture or in the holding of a Warrant Certificate, entitlement to a Warrant or otherwise, shall, in itself, confer or be construed as conferring upon a Warrantholder any right or interest whatsoever as a Shareholder, including, but not limited to, the right to vote at, to receive notice of, or to attend, meetings of Shareholders or any other proceedings of the Corporation, or the right to Dividends and other allocations.

2.4 Warrants to Rank Pari Passu

All Warrants shall rank equally and without preference over each other, whatever may be the actual date of issue thereof.

2.5 Form of Warrants, Certificated Warrants

The Warrants may be issued in both certificated and uncertificated form. Each Put Unit Warrant will be evidenced in certificated form only and bear the applicable legends as set forth in Schedule "A" hereto. All Warrants issued in certificated form shall be evidenced by a Warrant Certificate (including all replacements issued in accordance with this Indenture), substantially in the form set out in Schedule "A" hereto, which shall be dated as of the Issue Date, shall bear such distinguishing letters and numbers as the Corporation may, with the approval of the Warrant Agent, prescribe, and shall be issuable in any denomination excluding fractions. All Warrants issued to the Depository may be in either a certificated or uncertificated form, such uncertificated form being evidenced by a book position on the register of Warrantholders to be maintained by the Warrant Agent in accordance with Section 2.6.

2.6 Book Entry Only Warrants

(a) Registration of beneficial interests in and transfers of Warrants held by the Depository shall be made only through the book entry registration system and no Warrant Certificates shall be issued in respect of such Warrants except where physical certificates evidencing ownership in such securities are required or as set out herein or as may be requested by the Depository, as determined by the Corporation, from time to time. Except as provided in this Section 2.6, owners of beneficial interests in any Global Warrants shall not be entitled to have Warrants registered in their names and shall not receive or be entitled to receive Warrants in definitive form or to have their names appear in the register referred to in Section 2.9 herein. Notwithstanding any terms set out herein, Warrants having any legend set forth in Section 2.8 herein and held in the name of the Depository may only be held in the form of Uncertificated Warrants with the prior consent of the Warrant Agent and in accordance with the internal procedures of the Warrant Agent.

- (b) Notwithstanding any other provision in this Indenture, no Global Warrants may be exchanged in whole or in part for Warrants registered, and no transfer of any Global Warrants in whole or in part may be registered, in the name of any person other than the Depository for such Global Warrants or a nominee thereof unless:
 - the Depository notifies the Corporation that it is unwilling or unable to continue to act as depository in connection with the Book Entry Only Warrants and the Corporation is unable to locate a qualified successor;
 - (ii) the Corporation determines that the Depository is no longer willing, able or qualified to discharge properly its responsibilities as holder of the CDS Global Warrants and the Corporation is unable to locate a qualified successor;
 - (iii) the Depository ceases to be a clearing agency or otherwise ceases to be eligible to be a depository and the Corporation is unable to locate a qualified successor;
 - (iv) the Corporation determines that the Warrants shall no longer be held as Book Entry Only Warrants through the Depository;
 - (v) such right is required by Applicable Law, as determined by the Corporation and the Corporation's Counsel;
 - (vi) the Warrant is to be Authenticated to or for the account or benefit of a person in the United States or a U.S. Person; or
 - (vii) such registration is effected in accordance with the internal procedures of the Depository and the Warrant Agent,

following which, Warrants for those holders requesting the same shall be registered and issued to the beneficial owners of such Warrants or their nominees as directed by the holder. The Corporation shall provide an Officer's Certificate giving notice to the Warrant Agent of the occurrence of any event outlined in this Section 2.6 (2)(a)—(f).

- (c) Subject to the provisions of this Section 2.6, any exchange of CDS Global Warrants for Warrants which are not CDS Global Warrants may be made in whole or in part in accordance with the provisions of Section 2.11, *mutatis mutandis*. All such Warrants issued in exchange for a CDS Global Warrant or any portion thereof shall be registered in such names as the Depository for such CDS Global Warrants shall direct and shall be entitled to the same benefits and subject to the same terms and conditions (except insofar as they relate specifically to CDS Global Warrants) as the CDS Global Warrants or portion thereof surrendered upon such exchange.
- (d) Every Warrant that is Authenticated upon registration or transfer of a CDS Global Warrant, or in exchange for or in lieu of a CDS Global Warrant or any portion thereof, whether pursuant to this Section 2.6, or otherwise, shall be Authenticated in the form of, and shall be, a CDS Global Warrant, unless such Warrant is registered in the name of a person other than the Depository for such CDS Global Warrant or a nominee thereof.
- (e) Notwithstanding anything to the contrary in this Indenture, subject to Applicable Law, the CDS Global Warrant will be issued as an Uncertificated Warrant, unless otherwise requested in writing by the Depository or the Corporation.
- (f) The rights of beneficial owners of Warrants who hold securities entitlements in respect of the Warrants through the book entry registration system shall be limited to those established by applicable law and agreements between the Depository and the Book Entry Only Participants and between such Book Entry Only Participants and the

beneficial owners of Warrants who hold securities entitlements in respect of the Warrants through the book entry registration system, and such rights must be exercised through a Book Entry Only Participant in accordance with the rules and procedures of the Depository.

- (g) Notwithstanding anything herein to the contrary, neither the Corporation nor the Warrant Agent nor any agent thereof shall have any responsibility or liability for:
 - (i) the electronic records maintained by the Depository relating to any ownership interests or any other interests in the Warrants or the depository system maintained by the Depository, or payments made on account of any ownership interest or any other interest of any person in any Warrant represented by an electronic position in the book entry registration system (other than the Depository or its nominee);
 - (ii) maintaining, supervising or reviewing any records of the Depository or any Book Entry Only Participant relating to any such interest; or
 - (iii) any advice or representation made or given by the Depository or those contained herein that relate to the rules and regulations of the Depository or any action to be taken by the Depository on its own direction or at the direction of any Book Entry Only Participant.
- (h) The Corporation may terminate the application of this Section 2.6 in its sole discretion in which case all Warrants shall be evidenced by Warrant Certificates registered in the name of a Person other than the Depository.

2.7 Warrant Certificate

- (a) For Warrants issued in certificated form, the form of certificate representing Warrants shall be substantially as set out in Schedule "A" hereto or such other form as is authorized from time to time by the Warrant Agent, subject to the consent of the Corporation. Each Warrant Certificate shall be Authenticated manually on behalf of the Warrant Agent. Each Warrant Certificate shall be signed by any two duly authorized signatories of the Corporation; whose signature shall appear on the Warrant Certificate and may be printed, lithographed or otherwise mechanically reproduced thereon and, in such event, certificates so signed are as valid and binding upon the Corporation as if it had been signed manually. Any Warrant Certificate which has two signatures as hereinbefore provided shall be valid notwithstanding that one or more of the persons whose signature is printed, lithographed or mechanically reproduced no longer holds office at the date of issuance of such certificate. The Warrant Certificates may be engraved, printed or lithographed, or partly in one form and partly in another, as the Warrant Agent may determine.
- (b) The Warrant Agent shall Authenticate Uncertificated Warrants (whether upon original issuance, exchange, registration of transfer, partial payment, or otherwise) by completing its Internal Procedures and the Corporation shall, and hereby acknowledges that it shall, thereupon be deemed to have duly and validly issued such Uncertificated Warrants under this Indenture. Such Authentication shall be conclusive evidence that such Uncertificated Warrant has been duly issued hereunder and that the holder or holders are entitled to the benefits of this Indenture. The register shall be final and conclusive evidence as to all

matters relating to Uncertificated Warrants with respect to which this Indenture requires the Warrant Agent to maintain records or accounts. In case of differences between the register at any time and any other time the register at the later time shall be controlling, absent manifest error and such Uncertificated Warrants are binding on the Corporation.

- (c) Any Warrant Certificate validly issued in accordance with the terms of this Indenture in effect at the time of issue of such Warrant Certificate shall, subject to the terms of this Indenture and applicable law, validly entitle the holder to acquire the number of Common Shares set out in such Warrant Certificate in accordance with the terms hereof, notwithstanding that the form of such Warrant Certificate may not be in the form currently required by this Indenture.
- (d) No Warrant shall be considered issued and shall be valid or obligatory or shall entitle the holder thereof to the benefits of this Indenture, until it has been Authenticated by the Warrant Agent. Authentication by the Warrant Agent, including by way of entry on the register, shall not be construed as a representation or warranty by the Warrant Agent as to the validity of this Indenture or of such Warrant Certificates or Uncertificated Warrants (except the due Authentication thereof) or as to the performance by the Corporation of its obligations under this Indenture and the Warrant Agent shall in no respect be liable or answerable for the use made of the Warrants or any of them or of the consideration thereof. Authentication by the Warrant Agent shall be conclusive evidence as against the Corporation that the Warrants so Authenticated have been duly issued hereunder and that the holder thereof is entitled to the benefits of this Indenture.
- (e) No Certificated Warrant shall be considered issued and Authenticated or, if Authenticated, shall be obligatory or shall entitle the holder thereof to the benefits of this Indenture, until it has been Authenticated by manual signature by or on behalf of the Warrant Agent substantially in the form of the Warrant set out in Schedule "A" hereto. Such Authentication on any such Certificated Warrant shall be conclusive evidence that such Certificated Warrant is duly Authenticated and is valid and a binding obligation of the Corporation and that the holder is entitled to the benefits of this Indenture.
- (f) No Uncertificated Warrant shall be considered issued and shall be obligatory or shall entitle the holder thereof to the benefits of this Indenture, until it has been Authenticated by entry on the register of the particulars of the Uncertificated Warrant. Such entry on the register of the particulars of an Uncertificated Warrant shall be conclusive evidence that such Uncertificated Warrant is a valid and binding obligation of the Corporation and that the holder is entitled to the benefits of this Indenture.
- (g) The Authentication by the Warrant Agent of any Warrants whether by way of entry on the register or otherwise shall not be construed as a representation or warranty by the Warrant Agent as to the validity of the Indenture or such Warrants (except the due Authentication thereof) or as to the performance by the Corporation of its obligations under this Indenture and the Warrant Agent shall in no respect be liable or answerable for the use made of the Warrants or any of them or the proceeds thereof.

2.8 Legends

(a) Each Warrant Certificate evidencing a Put Option Warrant and each Warrant Certificate issued in exchange therefor or in substitution thereof shall bear or be deemed to bear the following legends or such variations thereof as the Corporation may prescribe from time to time:

"THESE WARRANTS AND THE SECURITIES DELIVERABLE UPON THE EXERCISE THEREOF HAVE NOT BEEN AND WILL NOT BE REGISTERED UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "U.S. SECURITIES ACT"). THESE SECURITIES MAY BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED ONLY (A) TO THE CORPORATION, (B) IF THE SECURITIES HAVE BEEN REGISTERED IN COMPLIANCE WITH THE REGISTRATION REQUIREMENTS UNDER THE U.S. SECURITIES ACT AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS (C) IN COMPLIANCE WITH THE EXEMPTION FROM THE REGISTRATION REQUIREMENTS UNDER THE U.S. SECURITIES ACT IN ACCORDANCE WITH RULE 144 THEREUNDER, IF APPLICABLE, AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS, OR (D) IN A TRANSACTION THAT DOES NOT REQUIRE REGISTRATION UNDER THE U.S. SECURITIES ACT OR ANY APPLICABLE STATE LAWS AND REGULATIONS GOVERNING THE OFFER AND SALE OF SECURITIES, AND THE HOLDER HAS, PRIOR TO SUCH SALE, FURNISHED TO THE CORPORATION AN OPINION OF COUNSEL OF RECOGNIZED STANDING, OR OTHER EVIDENCE OF EXEMPTION, REASONABLY SATISFACTORY TO THE CORPORATION. HEDGING TRANSACTIONS INVOLVING THE SECURITIES REPRESENTED HEREBY MAY NOT BE CONDUCTED UNLESS IN COMPLIANCE WITH U.S. SECURITIES LAWS.

THE SECURITIES EVIDENCED HEREBY AND THE SECURITIES ISSUABLE UPON EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE U.S. SECURITIES ACT OR U.S. STATE SECURITIES LAWS. THESE WARRANTS MAY NOT BE EXERCISED IN THE UNITED STATES OR BY OR ON BEHALF OF, OR FOR THE ACCOUNT OR BENEFIT OF, A U.S. PERSON UNLESS THIS SECURITY AND THE COMMON SHARES ISSUABLE UPON EXERCISE OF THIS SECURITY HAVE BEEN REGISTERED UNDER THE U.S. SECURITIES ACT AND THE APPLICABLE STATE SECURITIES LEGISLATION OR AN EXEMPTION FROM SUCH REGISTRATION REQUIREMENTS IS AVAILABLE. "UNITED STATES" AND "U.S. PERSON" ARE AS DEFINED BY REGULATION S UNDER THE U.S. SECURITIES ACT.";

The Warrant Agent shall be entitled to request any other documents that it may require in accordance with its internal policies for the removal of the legend set forth above.

(b) Each Warrant evidenced by the CDS Global Warrant originally issued in Canada and held by the Depository, and each CDS Global Warrant issued in exchange therefor or in substitution thereof shall bear or be deemed to bear the following legend or such variations thereof as the Corporation may prescribe from time to time:

"UNLESS THIS CERTIFICATE IS PRESENTED BY AN AUTHORIZED REPRESENTATIVE OF CDS CLEARING AND DEPOSITORY SERVICES INC. ("CDS") TO COHBAR, INC. (THE "ISSUER") OR ITS AGENT FOR REGISTRATION OF TRANSFER, EXCHANGE OR PAYMENT, AND ANY CERTIFICATE ISSUED IN RESPECT THEREOF IS REGISTERED IN THE NAME OF CDS & CO. OR IN SUCH OTHER NAME AS IS REQUESTED BY AN

AUTHORIZED REPRESENTATIVE OF CDS (AND ANY PAYMENT IS MADE TO CDS & CO. OR TO SUCH OTHER ENTITY AS IS REQUESTED BY AN AUTHORIZED REPRESENTATIVE OF CDS), ANY TRANSFER, PLEDGE OR OTHER USE HEREOF FOR VALUE OR OTHERWISE BY OR TO ANY PERSON IS WRONGFUL SINCE THE REGISTERED HOLDER HEREOF, CDS & CO., HAS A PROPERTY INTEREST IN THE SECURITIES REPRESENTED BY THIS CERTIFICATE HEREIN AND IT IS A VIOLATION OF ITS RIGHTS FOR ANOTHER PERSON TO HOLD, TRANSFER OR DEAL WITH THIS CERTIFICATE."

(c) Notwithstanding any other provisions of this Indenture, in processing and registering transfers of Warrants, no duty or responsibility whatsoever shall rest upon the Warrant Agent to determine the compliance by any transferor or transfere with the terms of the legend contained in subsections 2.8(1) or 2.8(2), or with the relevant securities laws or regulations, including, without limitation, Regulation S, and the Warrant Agent shall be entitled to assume that all transfers are legal and proper.

2.9 Register of Warrants

- (a) The Warrant Agent shall maintain records and accounts concerning the Warrants, whether certificated or uncertificated, which shall contain the information called for below with respect to each Warrant, together with such other information as may be required by law or as the Warrant Agent may elect to record. All such information shall be kept in one set of accounts and records which the Warrant Agent shall designate (in such manner as shall permit it to be so identified as such by an unaffiliated party) as the register of the holders of Warrants. The information to be entered for each account in the register of Warrants at any time shall include (without limitation):
 - (i) the name and address of the holder of the Warrants, the date of Authentication thereof and the number of Warrants;
 - (ii) whether such Warrant is a Certificated Warrant or an Uncertificated Warrant and, if a Warrant Certificate, the unique number or code assigned to and imprinted thereupon and, if an Uncertificated Warrant, the unique number or code assigned thereto if any;
 - (iii) a register of transfers in which all transfers of Warrants and the date and other particulars of each transfer shall be entered.

The register shall be available for inspection by the Corporation and or any Warrantholder during the Warrant Agent's regular business hours on a Business Day and upon payment to the Warrant Agent of its reasonable fees. Any Warrantholder exercising such right of inspection shall first provide an affidavit in form satisfactory to the Corporation and the Warrant Agent stating the name and address of the Warrantholder and agreeing not to use the information therein except in connection with an effort to call a meeting of Warrantholders or to influence the voting of Warrantholders at any meeting of Warrantholders.

(b) Once an Uncertificated Warrant has been Authenticated, the information set forth in the register with respect thereto at the time of Authentication may be altered, modified, amended, supplemented or otherwise changed only to reflect exercise or proper

instructions to the Warrant Agent from the holder as provided herein, except that the Warrant Agent may act unilaterally to make purely administrative changes internal to the Warrant Agent and changes to correct errors. Each person who becomes a holder of an Uncertificated Warrant, by his, her or its acquisition thereof shall be deemed to have irrevocably (i) consented to the foregoing authority of the Warrant Agent to make such minor error corrections and (ii) agreed to pay to the Warrant Agent, promptly upon written demand, the full amount of all loss and expense (including without limitation reasonable legal fees of the Corporation and the Warrant Agent plus interest, at an appropriate then prevailing rate of interest to the Warrant Agent), sustained by the Corporation or the Warrant Agent as a proximate result of such error if but only if and only to the extent that such present or former holder realized any benefit as a result of such error and could reasonably have prevented, forestalled or minimized such loss and expense by prompt reporting of the error or avoidance of accepting benefits thereof whether or not such error is or should have been timely detected and corrected by the Warrant Agent; provided, that no person who is a bona fide purchaser shall have any such obligation to the Corporation or to the Warrant Agent.

2.10 Issue in Substitution for Warrant Certificates Lost, etc.

- (a) If any Warrant Certificate becomes mutilated or is lost, destroyed or stolen, the Corporation, subject to applicable law, shall issue and thereupon the Warrant Agent shall certify and deliver, a new Warrant Certificate of like tenor, and bearing the same legend, if applicable, as the one mutilated, lost, destroyed or stolen in exchange for and in place of and upon cancellation of such mutilated Warrant Certificate, or in lieu of and in substitution for such lost, destroyed or stolen Warrant Certificate, and the substituted Warrant Certificate shall be in a form approved by the Warrant Agent and the Warrants evidenced thereby shall be entitled to the benefits hereof and shall rank equally in accordance with its terms with all other Warrants issued or to be issued hereunder.
- (b) The applicant for the issue of a new Warrant Certificate pursuant to this Section 2.10 shall bear the cost of the issue thereof and in case of loss, destruction or theft shall, as a condition precedent to the issuance thereof, furnish to the Corporation and to the Warrant Agent such evidence of ownership and of the loss, destruction or theft of the Warrant Certificate so lost, destroyed or stolen as shall be satisfactory to the Corporation and to the Warrant Agent, in their sole discretion, and such applicant shall also be required to furnish an indemnity and surety bond in amount and form satisfactory to the Corporation and the Warrant Agent, in their sole discretion, and shall pay the reasonable charges of the Corporation and the Warrant Agent in connection therewith.

2.11 Exchange of Warrant Certificates

- (a) Any one or more Warrant Certificates representing any number of Warrants may, upon compliance with the reasonable requirements of the Warrant Agent (including compliance with applicable securities legislation), be exchanged for one or more other Warrant Certificates representing the same aggregate number of Warrants, and bearing the same legend, if applicable, as represented by the Warrant Certificate or Warrant Certificates so exchanged.
- (b) Warrant Certificates may be exchanged only at the Warrant Agency or at any other place that is designated by the Corporation with the approval of the Warrant Agent. Any Warrant Certificate from the holder (or such other instructions, in form satisfactory to the Warrant Agent), tendered for exchange shall be surrendered to the Warrant Agency and cancelled by the Warrant Agent.

(c) Warrant Certificates exchanged for Warrant Certificates that bear the legend set forth in Section 2.8(a) shall bear the same legend.

2.12 Transfer and Ownership of Warrants

- (a) The Warrants may only be transferred on the register kept by the Warrant Agent at the Warrant Agency by the holder or its legal representatives or its attorney duly appointed by an instrument in writing in form and execution satisfactory to the Warrant Agent only upon (a) in the case of a Warrant Certificate, surrendering to the Warrant Agent at the Warrant Agency the Warrant Certificates representing the Warrants to be transferred together with a duly executed transfer form as set forth in Schedule:"B" and (b) in the case of Book Entry Only Warrants, in accordance with procedures prescribed by the Depository under the book entry registration system, and (c) upon compliance with:
 - (i) the conditions herein:
 - (ii) such reasonable requirements as the Warrant Agent may prescribe; and
 - (iii) all applicable securities legislation and requirements of regulatory authorities;

and such transfer shall be duly noted in such register by the Warrant Agent. Upon compliance with such requirements, the Warrant Agent shall issue to the transferee of a Warrant Certificate, or the Warrant Agent shall Authenticate and deliver a Warrant Certificate upon request that part of the CDS Global Warrant be certificated. Transfers within the systems of the Depository are not the responsibility of the Warrant Agent and will not be noted on the register maintained by the Warrant Agent.

- (b) If a Warrant Certificate tendered for transfer bears the legend set forth in Section 2.8(a), the Warrant Agent shall not register such transfer unless the transferor has provided the Warrant Agent with the Warrant Certificate and (A) the transfer is made to the Corporation or (B) a declaration to the effect set forth in Schedule "B" to this Warrant Indenture, or in such other form as the Corporation may from time to time prescribe, is delivered to the Warrant Agent, and if required by the Warrant Agent, the transferor provides an opinion of counsel of recognized standing, reasonably satisfactory to the Corporation and the Warrant Agent that the transfer is in compliance with applicable state laws and the U.S. Securities Act.
- (c) Subject to the provisions of this Indenture, Applicable Legislation and applicable law, the Warrantholder shall be entitled to the rights and privileges attaching to the Warrants, and the issue of Common Shares by the Corporation upon the exercise of Warrants in accordance with the terms and conditions herein contained shall discharge all responsibilities of the Corporation and the Warrant Agent with respect to such Warrants and neither the Corporation nor the Warrant Agent shall be bound to inquire into the title of any such holder.

2.13 Cancellation of Surrendered Warrants

All Warrant Certificates surrendered pursuant to Article 3 shall be cancelled by the Warrant Agent and upon such circumstances all such Uncertificated Warrants shall be deemed cancelled and so noted on the register by the Warrant Agent. Upon request by the Corporation, the Warrant Agent shall furnish to the Corporation a cancellation certificate identifying the Warrant Certificates so cancelled, the number of Warrants evidenced thereby, the number of Common Shares, if any, issued pursuant to such Warrants and the details of any Warrant Certificates issued in substitution or exchange for such Warrant Certificates cancelled.

ARTICLE 3 EXERCISE OF WARRANTS

3.1 Right of Exercise

Subject to the provisions hereof, each Registered Warrantholder may exercise the right conferred on such holder to subscribe for and purchase one Common Share for each Warrant after the Issue Date and prior to the Expiry Time and in accordance with the conditions herein

3.2 Warrant Exercise

- (a) Registered holders of Warrant Certificates who wish to exercise the Warrants held by them in order to acquire Common Shares must complete the exercise form (the "Exercise Notice") attached to the Warrant Certificate(s) which form is attached hereto as Schedule "C", which may be amended by the Corporation with the consent of the Warrant Agent, if such amendment does not, in the reasonable opinion of the Corporation and the Warrant Agent, which may be based on the advice of Counsel, materially and adversely affect the rights, entitlements and interests of the Warrantholders, and deliver such certificate(s), the executed Exercise Notice and a certified cheque, bank draft or money order payable to or to the order of the Corporation for the aggregate Exercise Price to the Warrant Agent at the Warrant Agency. The Warrants represented by a Warrant Certificate shall be deemed to be surrendered upon personal delivery of such certificate, Exercise Notice and aggregate Exercise Price or, if such documents are sent by mail or other means of transmission, upon actual receipt thereof by the Warrant Agent at the office referred to above.
- (b) A Registered holder of Uncertificated Warrants evidenced by a security entitlement in respect of Warrants must complete the Exercise Notice and deliver the executed Exercise Notice and a certified cheque, bank draft or money order payable to or to the order of the Corporation for the aggregate Exercise Price to the Warrant Agent at the Warrant Agency. The Uncertificated Warrants shall be deemed to be surrendered upon receipt of the Exercise Notice and aggregate Exercise Price or, if such documents are sent by mail or other means of transmission, upon actual receipt thereof by the Warrant Agent at the office referred to above.
- (c) A beneficial owner of Uncertificated Warrants who desires to exercise his, her or its Warrants must do so by causing a Book Entry Only Participant to deliver to the Depository on behalf of the entitlement holder, notice of the owner's intention to exercise Warrants in a manner acceptable to the Depository. Forthwith upon receipt by the Depository of such notice, as well as payment for the aggregate Exercise Price, the Depository shall deliver to the Warrant Agent confirmation of its intention to exercise

Warrants (a "Confirmation") in a manner acceptable to the Warrant Agent, including by electronic means through a book based registration system. If the Book Entry Only Participant is not able to make or deliver the foregoing representation by initiating the electronic exercise of the Warrants, then such Warrants shall be withdrawn from the book based registration system, by the Book Entry Only Participant and an individually registered Warrant Certificate shall be issued by the Warrant Agent to such Beneficial Owner or Book Entry Only Participant and the exercise procedures set forth in Section 3.2(a) shall be followed.

- (d) Payment representing the aggregate Exercise Price must be provided to the appropriate office of the Book Entry Only Participant in a manner acceptable to it. A notice in form acceptable to the Book Entry Only Participant and payment from such beneficial holder should be provided to the Book Entry Only Participant sufficiently in advance so as to permit the Book Entry Only Participant to deliver notice and payment to the Depository and for the Depository in turn to deliver notice and payment to the Warrant Agent prior to Expiry Time. The Depository will initiate the exercise by way of the Confirmation and forward the aggregate Exercise Price electronically to the Warrant Agent and the Warrant Agent will execute the exercise by issuing to the Depository through the book entry registration system the Common Shares to which the exercising Warrantholder is entitled pursuant to the exercise. Any expense associated with the exercise process will be for the account of the entitlement holder exercising the Warrants and/or the Book Entry Only Participant exercising the Warrants on its behalf.
- (e) By causing a Book Entry Only Participant to deliver notice to the Depository, a Warrantholder shall be deemed to have irrevocably surrendered his or her Warrants so exercised and appointed such Book Entry Only Participant to act as his or her exclusive settlement agent with respect to the exercise and the receipt of Common Shares in connection with the obligations arising from such exercise.
- (f) Any notice which the Depository determines to be incomplete, not in proper form or not duly executed shall for all purposes be void and of no effect and the exercise to which it relates shall be considered for all purposes not to have been exercised thereby. A failure by a Book Entry Only Participant to exercise or to give effect to the settlement thereof in accordance with the Warrantholder's instructions will not give rise to any obligations or liability on the part of the Corporation or Warrant Agent to the Book Entry Only Participant or the Warrantholder.
- (g) Any exercise form or Exercise Notice referred to in this Section 3.2 shall be signed by the Registered Warrantholder, or its executors or administrators or other legal representatives or an attorney of the Registered Warrantholder, duly appointed by an instrument in writing satisfactory to the Warrant Agent but such exercise form need not be executed by the Depository.
- (h) Any exercise referred to in this Section 3.2 shall require that the entire Exercise Price for Common Shares subscribed must be paid at the time of subscription and such Exercise Price and original Exercise Notice executed by the Registered Warrantholder or the Confirmation from the Depository must be received by the Warrant Agent prior to the Expiry Time.

- (i) Warrants may only be exercised pursuant to this Section 3.2 by or on behalf of a Registered Warrantholder, as applicable, who makes the certifications set forth on the Exercise Notice set out in Schedule "C" or as provided herein.
- (j) If the form of Exercise Notice set forth in the Warrant Certificate shall have been amended, the Corporation shall cause the amended Exercise Notice to be forwarded to all Registered Warrantholders.
- (k) Exercise Notices and Confirmations must be delivered to the Warrant Agent at any time during the Warrant Agent's actual business hours on any Business Day prior to the Expiry Time. Any Exercise Notice or Confirmations received by the Warrant Agent after business hours on any Business Day other than the Expiry Time will be deemed to have been received by the Warrant Agent on the next following Business Day.
- (1) Any Warrant with respect to which a Confirmation is not received by the Warrant Agent before the Expiry Time shall be deemed to have expired and become void and all rights with respect to such Warrants shall terminate and be cancelled.

3.3 Prohibition on Exercise by U.S. Persons; Legended Certificates; Special Conditions to Exercise of Put Unit Warrants.

- (a) If the issuance of Common Shares upon the exercise of any Warrants requires the maintenance of an effective Registration Statement under the U.S. Securities Act with respect to such Common Shares, then in no event shall such Common Shares be issued except pursuant to an effective Registration Statement under the U.S. Securities Act and the Corporation causes to be delivered to the holder a U.S. Prospectus; provided however, that in the absence of an effective Registration Statement, subject to applicable law, a holder of any Warrant may:
 - (i) exercise such Warrants, if the holder is not a U.S. Purchaser and the holder delivers a duly completed and executed Warrant Exercise Certification certifying that the holder: (A)(1) is not in the United States; (2) is not a U.S. Person and is not exercising the Warrants for, or on behalf or benefit of, a U.S. Person or person in the United States; (3) did not execute or deliver the Warrant Exercise Notice in the United States; (4) agrees not to engage in hedging transactions with regard to the Securities prior to the expiration of the one year distribution compliance period set forth in Rule 903(b)(3) of Regulation S; (5) acknowledges that the Common Shares issuable upon exercise of the Warrants are "restricted securities" as defined in Rule 144 of the U.S. Securities Act and upon the issuance thereof, and until such time as the same is no longer required under the applicable requirements of the U.S. Securities Act or applicable U.S. state laws and regulations, the certificates representing the Common Shares will bear a restrictive legend; and (6) acknowledges that the Corporation shall refuse to register any transfer of any securities not made pursuant to an effective Registration Statement under the U.S. Securities Act, or pursuant to an available exemption from registration under the U.S. Securities Act; and (B) neither the holder nor any affiliate of the holder or other person acting on the holder's behalf has engaged in any "directed selling efforts" (as defined in Regulation S) in the United States; or

- (ii) exercise such Warrants in a transaction that does not require registration under the U.S. Securities Act or any applicable U.S. state laws and regulations and the holder has (A) delivered a duly completed and executed Warrant Exercise Certification certifying that the holder is exercising the Warrants pursuant to such exemptions and (B) furnished to the Corporation, prior to such exercise, an opinion of counsel of recognized standing in form and substance satisfactory to the Corporation to such effect.
- (b) Section 3.3(a) notwithstanding, Put Unit Warrants which bear the legend set forth in Section 2.8(a) may be exercised in the United States by or on behalf of a U.S. Person, and Common Shares issued upon exercise of any such Warrants may be delivered to an address in the United States, provided that the holder delivers a duly completed and executed Warrant Exercise Certification certifying that the holder (a) is the original purchaser of the Warrants pursuant to the Corporation's exercise of its rights pursuant to a certain Put Agreement between the undersigned and the Corporation (the "Put Agreement"), (b) is exercising the Warrants for its own account, (c) is an "accredited investor" as defined in Rule 501(a) of Regulation D under the U.S. Securities Act at the time of exercise of these Warrants, and (d) the representations and warranties of the holder made in the original Put Agreement remain true and correct as of the date of exercise of the Warrants.
- (c) Unless the Warrant is exercised pursuant to an effective Registration Statement the certificate representing the Common Shares issued upon exercise of the Warrant will bear a legend substantially in the form set forth below:
 - THESE SECURITIES HAVE NOT BEEN AND WILL NOT BE REGISTERED UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "U.S. SECURITIES ACT"). THESE SECURITIES MAY BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED ONLY (A) TO THE CORPORATION, (B) IF THE SECURITIES HAVE BEEN REGISTERED IN COMPLIANCE WITH THE REGISTRATION REQUIREMENTS UNDER THE U.S. SECURITIES ACT AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS (C) IN COMPLIANCE WITH THE EXEMPTION FROM THE REGISTRATION REQUIREMENTS UNDER THE U.S. SECURITIES ACT IN ACCORDANCE WITH RULE 144 THEREUNDER, IF APPLICABLE, AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS, OR (D) IN A TRANSACTION THAT DOES NOT REQUIRE REGISTRATION UNDER THE U.S. SECURITIES ACT OR ANY APPLICABLE STATE LAWS AND REGULATIONS GOVERNING THE OFFER AND SALE OF SECURITIES, AND THE HOLDER HAS, PRIOR TO SUCH SALE, FURNISHED TO THE CORPORATION AN OPINION OF COUNSEL OF RECOGNIZED STANDING, OR OTHER EVIDENCE OF EXEMPTION, REASONABLY SATISFACTORY TO THE CORPORATION. HEDGING TRANSACTIONS INVOLVING THE SECURITIES REPRESENTED HEREBY MAY NOT BE CONDUCTED UNLESS IN COMPLIANCE WITH U.S. SECURITIES LAWS.
- (d) If issuance of Common Shares upon the exercise of any Warrant requires the maintenance of an effective Registration Statement under the U.S. Securities Act with respect to such Common Shares then the Corporation shall have the authority to suspend the exercise of any such Warrants while such registration statement is not effective. Similarly, a Warrantholder residing in a state where a required registration or

governmental approval of issuance of the Common Shares is not in effect as of or has not been obtained within a reasonable time after the surrender date of the Warrant Certificate for exercise shall not be entitled to exercise Warrants, unless in the opinion of counsel to the Corporation such registration or approval in such state shall not be required or the Corporation otherwise authorizes the issuance. In such event, the Warrantholder shall be entitled to transfer the Warrants to others, but only prior to the Expiry Time for the Warrants being transferred. If no Registration Statement is effective at any time when any Warrant is exercised, such Warrantholder shall be notified forthwith by the Warrant Agent that such Warrantholder is entitled, at his or her option, to exercise the Warrant only in accordance with the conditions set forth in Sections 3.3(a)(i)-(ii) or 3.3(b) and upon delivery of a Warrant Exercise Certification to the Warrant Agent and the Corporation.

(e) Notwithstanding that the Corporation may not have maintained a current Registration Statement in respect of Common Shares under the U.S. Securities Act, no Warrantholder (whether a U.S. Purchaser or otherwise) shall have any right to receive, and the Corporation shall be under no obligation to pay to any Warrantholder (whether a U.S. Purchaser or otherwise), any cash amount or other consideration or compensation upon exercise of the Warrants, other than as expressly provided by this Warrant Indenture, and the Corporation shall not be under any obligation to redeem or otherwise purchase any Warrants in any circumstance; provided, however, that nothing in this clause shall limit or restrict any remedies of the Warrant Agent or any Warrantholder or Warrantholders in respect of a breach by the Corporation of a representation, warranty or covenant hereunder.

3.4 Transfer Fees and Taxes

If any of the Common Shares subscribed for are to be issued to a person or persons other than the Registered Warrantholder, the Registered Warrantholder shall execute the form of transfer and will comply with such reasonable requirements as the Warrant Agent may stipulate and will pay to the Corporation or the Warrant Agent on behalf of the Corporation, all applicable transfer or similar taxes and the Corporation will not be required to issue or deliver certificates evidencing Common Shares unless or until such Warrantholder shall have paid to the Corporation or the Warrant Agent on behalf of the Corporation, the amount of such tax or shall have established to the satisfaction of the Corporation and the Warrant Agent that such tax has been paid or that no tax is due.

3.5 Warrant Agency

To facilitate the exchange, transfer or exercise of Warrants and compliance with such other terms and conditions hereof as may be required, the Corporation has appointed the Warrant Agency, as the agency at which Warrants may be surrendered for exchange or transfer or at which Warrants may be exercised and the Warrant Agent has accepted such appointment. The Corporation may from time to time designate alternate or additional places as the Warrant Agency (subject to the Warrant Agent's prior approval) and will give notice to the Warrant Agent of any proposed change of the Warrant Agency. Copies of the Warrant register shall also be kept at such other place or places, if any, as the Corporation, with the approval of the Warrant Agent, may designate. The Warrant Agent will from time to time when requested to do so by the Corporation or any Registered Warrantholder, upon payment of the Warrant Agent's reasonable charges, furnish a list of the names and addresses of Registered Warrantholders showing the number of Warrants held by each such Registered Warrantholder.

3.6 Effect of Exercise of Warrant Certificates

- (a) Upon the exercise of Warrant Certificates pursuant to and in compliance with Section 3.2 and subject to Section 3.3 and Section 3.4, the Common Shares to be issued pursuant to the Warrants exercised shall be deemed to have been issued and the person or persons to whom such Common Shares are to be issued shall be deemed to have become the holder or holders of such Common Shares within five Business Days of the Exercise Date unless the register shall be closed on such date, in which case the Common Shares subscribed for shall be deemed to have been issued and such person or persons deemed to have become the holder or holders of record of such Common Shares, on the date on which such register is reopened. It is hereby understood that in order for persons to whom Common Shares are to be issued, to become holders of Common Shares on record on the Exercise Date, beneficial holders must commence the exercise process sufficiently in advance so that the Warrant Agent is in receipt of all items of exercise at least one Business Day prior to such Exercise Date.
- (b) Within five Business Days after the Exercise Date with respect to a Warrant, the Warrant Agent shall cause to be delivered or mailed to the person or persons in whose name or names the Warrant is registered or, if so specified in writing by the holder, cause to be delivered to such person or persons at the Warrant Agency where the Warrant Certificate was surrendered, a certificate or certificates for the appropriate number of Common Shares subscribed for, or any other appropriate evidence of the issuance of Common Shares to such person or persons in respect of Common Shares issued under the book entry registration system.

3.7 Partial Exercise of Warrants; Fractions

- (a) The holder of any Warrants may exercise his right to acquire a number of whole Common Shares less than the aggregate number which the holder is entitled to acquire. In the event of any exercise of a number of Warrants less than the number which the holder is entitled to exercise, the holder of Warrants upon such exercise shall, in addition, be entitled to receive, without charge therefor, a new Warrant Certificate(s), bearing the same legend, if applicable, or other appropriate evidence of Warrants, in respect of the balance of the Warrants held by such holder and which were not then exercised.
- (b) Notwithstanding anything herein contained including any adjustment provided for in Section 4.1, the Corporation shall not be required, upon the exercise of any Warrants, to issue fractions of Common Shares. Warrants may only be exercised in a sufficient number to acquire whole numbers of Common Shares.

3.8 Expiration of Warrants

Immediately after the Expiry Time, all rights under any Warrant in respect of which the right of acquisition provided for herein shall not have been exercised shall cease and terminate and each Warrant shall be void and of no further force or effect.

3.9 Accounting and Recording

(a) The Warrant Agent shall promptly account to the Corporation with respect to Warrants exercised, and shall promptly forward to the Corporation (or into an account or accounts of the Corporation with the bank or trust company designated by the Corporation for that purpose), all monies received by the Warrant Agent on the subscription of Common Shares through the exercise of Warrants. All such monies and any securities or other instruments, from time to time received by the Warrant Agent, shall be received in trust for, and shall be segregated and kept apart by the Warrant Agent, the Warrantholders and the Corporation as their interests may appear in the register maintained by the Warrant Agent.

(b) The Warrant Agent shall record the particulars of Warrants exercised, which particulars shall include the names and addresses of the persons who become holders of Common Shares on exercise and the Exercise Date, in respect thereof. The Warrant Agent shall provide such particulars in writing to the Corporation within five Business Days of any request by the Corporation therefor.

3.10 Securities Restrictions

Notwithstanding anything herein contained, Common Shares will be issued upon exercise of a Warrant only in compliance with the securities laws of any applicable jurisdiction.

ARTICLE 4 ADJUSTMENT OF NUMBER OF COMMON SHARES AND EXERCISE PRICE

4.1 Adjustment of Number of Common Shares and Exercise Price

The subscription rights in effect under the Warrants for Common Shares issuable upon the exercise of the Warrants shall be subject to adjustment from time to time as follows:

- (a) if, at any time during the Adjustment Period, the Corporation shall:
 - (i) subdivide, re-divide or change its outstanding Common Shares into a greater number of Common Shares;
 - (ii) reduce, combine or consolidate its outstanding Common Shares into a lesser number of Common Shares; or
 - (iii) issue Common Shares or securities exchangeable for, or convertible into, Common Shares to all or substantially all of the holders of Common Shares by way of stock dividend or other distribution (other than a distribution of Common Shares upon the exercise of Warrants or any outstanding options);

(any of such events in Section 4.1(a) (i), (ii) or (iii) being called a "Common Share Reorganization") then the Exercise Price as of the effective date or record date of such Common Share Reorganization shall in the case of the events referred to in (i) or (iii) above be decreased in proportion to the number of outstanding Common Shares resulting from such subdivision, re-division, change or distribution, or shall, in the case of the events referred to in (ii) above, be increased in proportion to the number of outstanding Common Shares resulting from such reduction, combination or consolidation by multiplying the Exercise Price in effect immediately prior to such effective date or record date by a fraction, the numerator of which shall be the number of Common Shares outstanding on such effective date or record date before giving effect to such Common Share Reorganization and the denominator of which shall be the number of Common

Shares outstanding as of the effective date or record date after giving effect to such Common Shares Reorganization (including, in the case where securities exchangeable for or convertible into Common Shares are distributed, the number of Common Share that would have been outstanding had such securities been exchanged for or converted into Common Shares on such record date or effective date). Such adjustment shall be made successively whenever any Common Share Reorganization shall occur. Upon any adjustment of the Exercise Price pursuant to Section 4.1(a), the Exchange Rate shall be contemporaneously adjusted by multiplying the number of Common Shares theretofore obtainable on the exercise thereof by a fraction of which the numerator shall be the Exercise Price in effect immediately prior to such adjustment and the denominator shall be the Exercise Price resulting from such adjustment;

- if and whenever at any time during the Adjustment Period, the Corporation shall fix a record date for the issuance of rights, options or warrants to all or substantially all the holders of its outstanding Common Shares entitling them, for a period expiring not more than 45 days after such record date, to subscribe for or purchase Common Shares (or securities convertible or exchangeable into Common Shares) at a price per Common Share (or having a conversion or exchange price per Common Share) less than 95% of the Current Market Price on such record date (a "Rights Offering"), the Exercise Price shall be adjusted immediately after such record date so that it shall equal the amount determined by multiplying the Exercise Price in effect on such record date by a fraction, of which the numerator shall be the total number of Common Shares outstanding on such record date plus a number of Common Shares equal to the number arrived at by dividing the aggregate price of the total number of additional Common Shares offered for subscription or purchase (or the aggregate conversion or exchange price of the convertible or exchangeable securities so offered) by the Current Market Price, and of which the denominator shall be the total number of Common Shares outstanding on such record date plus the total number of additional Common Shares offered for subscription or purchase or into which the convertible or exchangeable securities so offered are convertible or exchangeable; any Common Shares owned by or held for the account of the Corporation shall be deemed not to be outstanding for the purpose of any such computation; such adjustment shall be made successively whenever such a record date is fixed; to the extent that no such rights or warrants are exercised prior to the expiration thereof, the Exercise Price shall be readjusted to the Exercise Price which would then be in effect if such record date had not been fixed or, if any such rights or warrants are exercised, to the Exercise Price which would then be in effect based upon the number of Common Shares (or securities convertible or exchangeable into Common Shares) actually issued upon the exercise of such rights or warrants, as the case may be. Upon any adjustment of the Exercise Price pursuant to this Section 4.1(b), the Exchange Rate will be adjusted immediately after such record date so that it will equal the rate determined by multiplying the Exchange Rate in effect on such record date by a fraction, of which the numerator shall be the Exercise Price in effect immediately prior to such adjustment and the denominator shall be the Exercise Price resulting from such adjustment. Such adjustment will be made successively whenever such a record date is fixed, provided that if two or more such record dates or record dates referred to in this Section 4.1(b) are fixed within a period of 25 Trading Days, such adjustment will be made successively as if each of such record dates occurred on the earliest of such record dates;
- (c) if and whenever at any time during the Adjustment Period the Corporation shall fix a record date for the making of a distribution to all or substantially all the holders of its outstanding Common Shares of (i) securities of any class, whether of the Corporation or

any other trust (other than Common Shares), (ii) rights, options or warrants to subscribe for or purchase Common Shares (or other securities convertible into or exchangeable for Common Shares), other than pursuant to a Rights Offering; (iii) evidences of its indebtedness or (iv) any property or other assets then, in each such case, the Exercise Price shall be adjusted immediately after such record date so that it shall equal the price determined by multiplying the Exercise Price in effect on such record date by a fraction, of which the numerator shall be the total number of Common Shares outstanding on such record date multiplied by the Current Market Price on such record date, less the excess, if any, of the fair market value on such record date, as determined by the Corporation (whose determination shall be conclusive), of such securities or other assets so issued or distributed over the fair market value of any consideration received therefor by the Corporation from the holders of the Common Shares, and of which the denominator shall be the total number of Common Shares outstanding on such record date multiplied by the Current Market Price; and Common Shares owned by or held for the account of the Corporation shall be deemed not to be outstanding for the purpose of any such computation; such adjustment shall be made successively whenever such a record date is fixed; to the extent that such distribution is not so made, the Exercise Price shall be readjusted to the Exercise Price which would then be in effect if such record date had not been fixed. Upon any adjustment of the Exercise Price pursuant to this Section 4.1(c), the Exchange Rate will be adjusted immediately after such record date so that it will equal the rate determined by multiplying the Exchange Rate in effect on such record date by a fraction, of which the numerator shall be the Exercise Price in effect immediately prior to such adjustment and the denominator shall be the Exercise Price resulting from such adjustment;

if and whenever at any time during the Adjustment Period, there is a reclassification of the Common Shares or a capital reorganization of the Corporation other than as described in Section 4.1(a) or a consolidation, amalgamation, arrangement or merger of the Corporation with or into any other body corporate, trust, partnership or other entity, or a sale or conveyance of the property and assets of the Corporation as an entirety or substantially as an entirety to any other body corporate, trust, partnership or other entity, any Registered Warrantholder who has not exercised its right of acquisition prior to the effective date of such reclassification, capital reorganization, consolidation, amalgamation, arrangement or merger, sale or conveyance, upon the exercise of such right thereafter, shall be entitled to receive upon payment of the Exercise Price and shall accept, in lieu of the number of Common Shares that prior to such effective date the Registered Warrantholder would have been entitled to receive, the number of shares or other securities or property of the Corporation or of the body corporate, trust, partnership or other entity resulting from such merger, amalgamation or consolidation, or to which such sale or conveyance may be made, as the case may be, that such Registered Warrantholder would have been entitled to receive on such reclassification, capital reorganization, consolidation, amalgamation, arrangement or merger, sale or conveyance, if, on the effective date thereof, as the case may be, the Registered Warrantholder had been the registered holder of the number of Common Shares to which prior to such effective date it was entitled to acquire upon the exercise of the Warrants. If determined appropriate by the Warrant Agent, relying on advice of Counsel, to give effect to or to evidence the provisions of this Section 4.1(d), the Corporation, its successor, or such purchasing body corporate, partnership, trust or other entity, as the case may be, shall, prior to or contemporaneously with any such reclassification, capital reorganization, consolidation, amalgamation, arrangement, merger, sale or conveyance, enter into an indenture which shall provide, to the extent possible, for the application of the provisions

set forth in this Indenture with respect to the rights and interests thereafter of the Registered Warrantholders to the end that the provisions set forth in this Indenture shall thereafter correspondingly be made applicable, as nearly as may reasonably be, with respect to any shares, other securities or property to which a Registered Warrantholder is entitled on the exercise of its acquisition rights thereafter. Any indenture entered into between the Corporation and the Warrant Agent pursuant to the provisions of this Section 4.1(d) shall be a supplemental indenture entered into pursuant to the provisions of Article 8 hereof. Any indenture entered into between the Corporation, any successor to the Corporation or such purchasing body corporate, partnership, trust or other entity and the Warrant Agent shall provide for adjustments which shall be as nearly equivalent as may be practicable to the adjustments provided in this Section 4.1 and which shall apply to successive reclassifications, capital reorganizations, amalgamations, consolidations, mergers, sales or conveyances;

- (e) in any case in which this Section 4.1 shall require that an adjustment shall become effective immediately after a record date for an event referred to herein, the Corporation may defer, until the occurrence of such event, issuing to the Registered Warrantholder of any Warrant exercised after the record date and prior to completion of such event the additional Common Shares issuable by reason of the adjustment required by such event before giving effect to such adjustment; provided, however, that the Corporation shall deliver to such Registered Warrantholder an appropriate instrument evidencing such Registered Warrantholder's right to receive such additional Common Shares upon the occurrence of the event requiring such adjustment and the right to receive any distributions made on such additional Common Shares declared in favour of holders of record of Common Shares on and after the relevant date of exercise or such later date as such Registered Warrantholder would, but for the provisions of this Section 4.1(e), have become the holder of record of such additional Common Shares pursuant to Section 4.1;
- (f) in any case in which Section 4.1(a)(iii), Section 4.1(b) or Section 4.1(c) require that an adjustment be made to the Exercise Price, no such adjustment shall be made if the Registered Warrantholders of the outstanding Warrants receive, subject to any required stock exchange or regulatory approval, the rights or warrants referred to in Section 4.1(a)(iii), Section 4.1(b) or the shares, rights, options, warrants, evidences of indebtedness or assets referred to in Section 4.1(c), as the case may be, in such kind and number as they would have received if they had been holders of Common Shares on the applicable record date or effective date, as the case may be, by virtue of their outstanding Warrant having then been exercised into Common Shares at the Exercise Price in effect on the applicable record date or effective date, as the case may be;
- (g) the adjustments provided for in this Section 4.1 are cumulative, and shall, in the case of adjustments to the Exercise Price be computed to the nearest whole cent and shall apply to successive subdivisions, re-divisions, reductions, combinations, consolidations, distributions, issues or other events resulting in any adjustment under the provisions of this Section 4.1, provided that, notwithstanding any other provision of this Section, no adjustment of the Exercise Price shall be required unless such adjustment would require an increase or decrease of at least 1% in the Exercise Price then in effect; provided, however, that any adjustments which by reason of this Section 4.1(g) are not required to be made shall be carried forward and taken into account in any subsequent adjustment; and

(h) after any adjustment pursuant to this Section 4.1, the term "Common Shares" where used in this Indenture shall be interpreted to mean securities of any class or classes which, as a result of such adjustment and all prior adjustments pursuant to this Section 4.1, the Registered Warrantholder is entitled to receive upon the exercise of his Warrant, and the number of Common Shares indicated by any exercise made pursuant to a Warrant shall be interpreted to mean the number of Common Shares or other property or securities a Registered Warrantholder is entitled to receive, as a result of such adjustment and all prior adjustments pursuant to this Section 4.1, upon the full exercise of a Warrant.

4.2 Entitlement to Common Shares on Exercise of Warrant

All Common Shares or shares of any class or other securities, which a Registered Warrantholder is at the time in question entitled to receive on the exercise of its Warrant, whether or not as a result of adjustments made pursuant to this Article 4, shall, for the purposes of the interpretation of this Indenture, be deemed to be Common Shares which such Registered Warrantholder is entitled to acquire pursuant to such Warrant.

4.3 No Adjustment for Certain Transactions

Notwithstanding anything in this Article 4, no adjustment shall be made in the acquisition rights attached to the Warrants if the issue of Common Shares is being made pursuant to this Indenture or in connection with (a) any share incentive plan or restricted share plan or share purchase plan in force from time to time for directors, officers, employees, consultants or other service providers of the Corporation; or (b) the satisfaction of existing instruments issued at the date hereof.

4.4 Determination by Independent Firm

In the event of any question arising with respect to the adjustments provided for in this Article 4 such question shall be conclusively determined by an independent firm of chartered accountants other than the Auditors, who shall have access to all necessary records of the Corporation, and such determination shall be binding upon the Corporation, the Warrant Agent, all holders and all other persons interested therein.

4.5 Proceedings Prior to any Action Requiring Adjustment

As a condition precedent to the taking of any action which would require an adjustment in any of the acquisition rights pursuant to any of the Warrants, including the number of Common Shares which are to be received upon the exercise thereof, the Corporation shall take any action which may, in the opinion of Counsel, be necessary in order that the Corporation has unissued and reserved in its authorized capital and may validly and legally issue as fully paid and non-assessable all the Common Shares which the holders of such Warrants are entitled to receive on the full exercise thereof in accordance with the provisions hereof.

4.6 Certificate of Adjustment

The Corporation shall from time to time immediately after the occurrence of any event which requires an adjustment or readjustment as provided in Section 4.1, deliver a certificate of the Corporation to the Warrant Agent specifying the nature of the event requiring the same and the amount of the adjustment or readjustment necessitated thereby and setting forth in reasonable detail the method of calculation and the facts upon which such calculation is based, which certificate shall be supported by a

certificate of the Corporation's Auditors verifying such calculation. The Warrant Agent shall act and rely, and shall be protected in so doing, upon the certificate of the Corporation or of the Corporation's Auditor and any other document filed by the Corporation pursuant to this Article 4 for all purposes.

4.7 Notice of Special Matters

The Corporation covenants with the Warrant Agent that, so long as any Warrant remains outstanding, it will give notice to the Warrant Agent and to the Registered Warrantholders of its intention to fix a record date that is prior to the Expiry Time for any matter for which an adjustment may be required pursuant to Section 4.1 Such notice shall specify the particulars of such event and the record date for such event, provided that the Corporation shall only be required to specify in the notice such particulars of the event as shall have been fixed and determined on the date on which the notice is given. The notice shall be given in each case not less than 14 days prior to such applicable record date. If notice has been given and the adjustment is not then determinable, the Corporation shall promptly, after the adjustment is determinable, file with the Warrant Agent a computation of the adjustment and give notice to the Registered Warrantholders of such adjustment computation.

4.8 No Action after Notice

The Corporation covenants with the Warrant Agent that it will not close its transfer books or take any other corporate action which might deprive the Registered Warrantholder of the opportunity to exercise its right of acquisition pursuant thereto during the period of 14 days after the giving of the certificate or notices set forth in Section 4.6 and Section 4.7.

4.9 Other Action

If the Corporation, after the date hereof, shall take any action affecting the Common Shares other than action described in Section 4.1, which in the reasonable opinion of the directors of the Corporation would materially affect the rights of Registered Warrantholders, the Exercise Price and/or Exchange Rate, the number of Common Shares which may be acquired upon exercise of the Warrants shall be adjusted in such manner and at such time, by action of the directors, acting reasonably and in good faith, in their sole discretion as they may determine to be equitable to the Registered Warrantholders in the circumstances, provided that no such adjustment will be made unless any requisite prior approval of any stock exchange on which the Common Shares are listed for trading has been obtained.

4.10 Protection of Warrant Agent

The Warrant Agent shall not:

- (a) at any time be under any duty or responsibility to any Registered Warrantholder to determine whether any facts exist which may require any adjustment contemplated by Section 4.1, or with respect to the nature or extent of any such adjustment when made, or with respect to the method employed in making the same;
- (b) be accountable with respect to the validity or value (or the kind or amount) of any Common Shares or of any other securities or property which may at any time be issued or delivered upon the exercise of the rights attaching to any Warrant;
- (c) be responsible for any failure of the Corporation to issue, transfer or deliver Common Shares or certificates for the same upon the surrender of any Warrants for the purpose of the exercise of such rights or to comply with any of the covenants contained in this Article; and

(d) incur any liability or be in any way responsible for the consequences of any breach on the part of the Corporation of any of the representations, warranties or covenants herein contained or of any acts of the directors, officers, employees, agents or servants of the Corporation.

4.11 Participation by Warrantholder

No adjustments shall be made pursuant to this Article 4 if the Registered Warrantholders are entitled to participate in any event described in this Article 4 on the same terms, mutatis mutandis, as if the Registered Warrantholders had exercised their Warrants prior to, or on the effective date or record date of, such event.

ARTICLE 5 RIGHTS OF THE CORPORATION AND COVENANTS

5.1 Optional Purchases by the Corporation

Subject to compliance with applicable securities legislation and approval of applicable regulatory authorities, if any, the Corporation may from time to time purchase by private contract or otherwise any of the Warrants. Any such purchase shall be made at the lowest price or prices at which, in the opinion of the directors, such Warrants are then obtainable, plus reasonable costs of purchase, and may be made in such manner, from such persons and on such other terms as the Corporation, in its sole discretion, may determine. In the case of Certificated Warrants, Warrant Certificates representing the Warrants purchased pursuant to this Section 5.1 shall forthwith be delivered to and cancelled by the Warrant Agent and reflected accordingly on the register of Warrants. In the case of Uncertificated Warrants, the Warrants purchased pursuant to this Section 5.1 shall be reflected accordingly on the register of Warrant and in accordance with procedures prescribed by the Depository under the book entry registration system. No Warrants shall be issued in replacement thereof.

5.2 General Covenants

The Corporation covenants with the Warrant Agent that so long as any Warrants remain outstanding:

- (a) it will reserve and keep available a sufficient number of Common Shares for the purpose of enabling it to satisfy its obligations to issue Common Shares upon the exercise of the Warrants;
- (b) it will cause the Common Shares from time to time acquired pursuant to the exercise of the Warrants to be duly issued and delivered in accordance with the Warrants and the terms hereof;
- (c) all Common Shares which shall be issued upon exercise of the right to acquire provided for herein shall be fully paid and non-assessable;
- (d) it will use reasonable commercial efforts to maintain its existence;

- (e) it will use reasonable commercial efforts to maintain the effectiveness of a Registration Statement under the U.S. Securities Act covering the issuance upon exercise of the IPO Unit Warrants and the Compensation Unit Warrants until the earlier of (i) the Expiry Time or (ii) the date all IPO Unit Warrants and Compensation Unit Warrants have been exercised;
- (f) it will use reasonable commercial efforts to ensure that all Common Shares outstanding or issuable from time to time (including without limitation the Common Shares issuable on the exercise of the Warrants) continue to be or are listed and posted for trading on the TSXV (or such other North American stock exchange or quotation system acceptable to the Corporation), provided that this clause shall not be construed as limiting or restricting the Corporation from completing a consolidation, amalgamation, arrangement, takeover bid or merger that would result in the Common Shares ceasing to be listed and posted for trading on the TSXV, so long as the holders of Common Shares receive securities of an entity which is listed on a stock exchange or quotation system in North American, or cash, or the holders of the Common shares have approved the transaction in accordance with the requirements of applicable corporate and securities laws and the policies of the TSXV;
- (g) it will make all requisite filings under applicable Canadian securities legislation including those necessary to remain a reporting issuer not in default in each of the provinces and other Canadian jurisdictions where it is or becomes a reporting issuer;
- (h) generally, it will well and truly perform and carry out all of the acts or things to be done by it as provided in this Indenture; and
- (i) the Corporation will promptly notify the Warrant Agent and the Warrantholders in writing of any default under the terms of this Warrant Indenture which remains unrectified for more than five days following its occurrence.

5.3 Warrant Agent's Remuneration and Expenses

The Corporation covenants that it will pay to the Warrant Agent from time to time reasonable remuneration for its services hereunder and will pay or reimburse the Warrant Agent upon its request for all reasonable expenses, disbursements and advances incurred or made by the Warrant Agent in the administration or execution of the trusts hereby created (including the reasonable compensation and the disbursements of its Counsel and all other advisers and assistants not regularly in its employ) both before any default hereunder and thereafter until all duties of the Warrant Agent hereunder shall be finally and fully performed. Any amount owing hereunder and remaining unpaid after 30 days from the invoice date will bear interest at the then current rate charged by the Warrant Agent against unpaid invoices and shall be payable upon demand. This Section shall survive the resignation or removal of the Warrant Agent and/or the termination of this Indenture.

5.4 Performance of Covenants by Warrant Agent

If the Corporation shall fail to perform any of its covenants contained in this Indenture, the Warrant Agent may notify the Registered Warrantholders of such failure on the part of the Corporation and may itself perform any of the covenants capable of being performed by it but, subject to Section 9.2, shall be under no obligation to perform said covenants or to notify the Registered Warrantholders of such performance by it. All sums expended or advanced by the Warrant Agent in so doing shall be repayable as provided in Section 5.3. No such performance, expenditure or advance by the Warrant Agent shall relieve the Corporation of any default hereunder or of its continuing obligations under the covenants herein contained.

5.5 Enforceability of Warrants

The Corporation covenants and agrees that it is duly authorized to create and issue the Warrants to be issued hereunder and that the Warrants, when issued and Authenticated as herein provided, will be valid and enforceable against the Corporation in accordance with the provisions hereof and the terms hereof and that, subject to the provisions of this Indenture, the Corporation will cause the Common Shares from time to time acquired upon exercise of Warrants issued under this Indenture to be duly issued and delivered in accordance with the terms of this Indenture.

ARTICLE 6 ENFORCEMENT

6.1 Suits by Registered Warrantholders

All or any of the rights conferred upon any Registered Warrantholder by any of the terms of this Indenture may be enforced by the Registered Warrantholder by appropriate proceedings but without prejudice to the right which is hereby conferred upon the Warrant Agent to proceed in its own name to enforce each and all of the provisions herein contained for the benefit of the Registered Warrantholders.

6.2 Suits by the Corporation

The Corporation shall have the right to enforce full payment of the Exercise Price of all Common Shares issued by the Warrant Agent to a Registered Warrantholder hereunder and shall be entitled to demand such payment from the Registered Warrantholder or alternatively to instruct the Warrant Agent to cancel the share certificates and amend the securities register accordingly.

6.3 Immunity of Shareholders, etc.

The Warrant Agent and the Warrantholders hereby waive and release any right, cause of action or remedy now or hereafter existing in any jurisdiction against any incorporator or any past, present or future shareholder, trustee, employee or agent of the Corporation or any successor Corporation on any covenant, agreement, representation or warranty by the Corporation herein.

6.4 Waiver of Default

Upon the happening of any default hereunder:

- (a) the Registered Warrantholders of not less than 51% of the Warrants then outstanding shall have power (in addition to the powers exercisable by Extraordinary Resolution) by requisition in writing to instruct the Warrant Agent to waive any default hereunder and the Warrant Agent shall thereupon waive the default upon such terms and conditions as shall be prescribed in such requisition; or
- (b) the Warrant Agent shall have power to waive any default hereunder upon such terms and conditions as the Warrant Agent may deem advisable, on the advice of Counsel, if, in the Warrant Agent's opinion, based on the advice of Counsel, the same shall have been cured or adequate provision made therefor;

provided that no delay or omission of the Warrant Agent or of the Registered Warrantholders to exercise any right or power accruing upon any default shall impair any such right or power or shall be construed to be a waiver of any such default or acquiescence therein and provided further that no act or omission either of the Warrant Agent or of the Registered Warrantholders in the premises shall extend to or be taken in any manner whatsoever to affect any subsequent default hereunder of the rights resulting therefrom.

ARTICLE 7 MEETINGS OF REGISTERED WARRANTHOLDERS

7.1 Right to Convene Meetings

The Warrant Agent may at any time and from time to time, and shall on receipt of a written request of the Corporation or of a Warrantholders' Request and upon being indemnified and funded to its reasonable satisfaction by the Corporation or by the Registered Warrantholders signing such Warrantholders' Request against the costs which may be incurred in connection with the calling and holding of such meeting, convene a meeting of the Registered Warrantholders. If the Warrant Agent fails to so call a meeting within seven days after receipt of such written request of the Corporation or within 30 days after receipt of such Warrantholders' Request and the indemnity and funding given as aforesaid, the Corporation or such Registered Warrantholders, as the case may be, may convene such meeting. Every such meeting shall be held in the City of Vancouver, British Columbia or at such other place as may be approved or determined by the Warrant Agent.

7.2 Notice

At least 21 days' prior written notice of any meeting of Registered Warrantholders shall be given to the Registered Warrantholders in the manner provided for in Section 10.2 and a copy of such notice shall be sent by mail to the Warrant Agent (unless the meeting has been called by the Warrant Agent) and to the Corporation (unless the meeting has been called by the Corporation). Such notice shall state the time when and the place where the meeting is to be held, shall state briefly the general nature of the business to be transacted thereat and shall contain such information as is reasonably necessary to enable the Registered Warrantholders to make a reasoned decision on the matter, but it shall not be necessary for any such notice to set out the terms of any resolution to be proposed or any of the provisions of this Section 7.2.

7.3 Chairman

An individual (who need not be a Registered Warrantholder) designated in writing by the Warrant Agent shall be chairman of the meeting and if no individual is so designated, or if the individual so designated is not present within fifteen minutes from the time fixed for the holding of the meeting, the Registered Warrantholders present in person or by proxy shall choose an individual present to be chairman.

7.4 Quorum

Subject to the provisions of Section 7.11, at any meeting of the Registered Warrantholders a quorum shall consist of Registered Warrantholder(s) present in person or by proxy and entitled to purchase at least (a) 50% of the aggregate number of Common Shares which could be acquired pursuant to all the then outstanding Warrants, if summoned by Registered Warrantholders or on a Warrantholders' request or (b) 10% of the aggregate number of Common Shares which could be acquired pursuant to all the then outstanding Warrants, if summoned by the Corporation. If a quorum of the

Registered Warrantholders shall not be present within thirty minutes from the time fixed for holding any meeting, the meeting, if summoned by Registered Warrantholders or on a Warrantholders' Request, shall be dissolved; but in any other case the meeting shall be adjourned to the same day in the next week (unless such day is not a Business Day, in which case it shall be adjourned to the next following Business Day) at the same time and place and no notice of the adjournment need be given. Any business may be brought before or dealt with at an adjourned meeting which might have been dealt with at the original meeting in accordance with the notice calling the same. No business shall be transacted at any meeting unless a quorum be present at the commencement of business. At the adjourned meeting the Registered Warrantholders present in person or by proxy shall form a quorum and may transact the business for which the meeting was originally convened, notwithstanding that they may not be entitled to acquire at least 10% of the aggregate number of Common Shares which may be acquired pursuant to all then outstanding Warrants.

7.5 Power to Adjourn

The chairman of any meeting at which a quorum of the Registered Warrantholders is present may, with the consent of the meeting, adjourn any such meeting, and no notice of such adjournment need be given except such notice, if any, as the meeting may prescribe.

7.6 Show of Hands

Every question submitted to a meeting shall be decided in the first place by a majority of the votes given on a show of hands except that votes on an Extraordinary Resolution shall be given in the manner hereinafter provided. At any such meeting, unless a poll is duly demanded as herein provided, a declaration by the chairman that a resolution has been carried or carried unanimously or by a particular majority or lost or not carried by a particular majority shall be conclusive evidence of the fact.

7.7 Poll and Voting

- (a) On every Extraordinary Resolution, and on any other question submitted to a meeting and after a vote by show of hands when demanded by the chairman or by one or more of the Registered Warrantholders acting in person or by proxy and entitled to acquire in the aggregate at least 5% of the aggregate number of Common Shares which could be acquired pursuant to all the Warrants then outstanding, a poll shall be taken in such manner as the chairman shall direct. Questions other than those required to be determined by Extraordinary Resolution shall be decided by a majority of the votes cast on the poll.
- (b) On a show of hands, every person who is present and entitled to vote, whether as a Registered Warrantholder or as proxy for one or more absent Registered Warrantholders, or both, shall have one vote. On a poll, each Registered Warrantholder present in person or represented by a proxy duly appointed by instrument in writing shall be entitled to one vote in respect of each Warrant then held or represented by it. A proxy need not be a Registered Warrantholder. The chairman of any meeting shall be entitled, both on a show of hands and on a poll, to vote in respect of the Warrants, if any, held or represented by him.

7.8 Regulations

(a) The Warrant Agent, or the Corporation with the approval of the Warrant Agent, may from time to time make and from time to time vary such regulations as it shall think fit for the setting of the record date for a meeting for the purpose of determining Registered Warrantholders entitled to receive notice of and to vote at the meeting.

(b) Any regulations so made shall be binding and effective and the votes given in accordance therewith shall be valid and shall be counted. Save as such regulations may provide, the only persons who shall be recognized at any meeting as a Registered Warrantholder, or be entitled to vote or be present at the meeting in respect thereof (subject to Section 7.9), shall be Registered Warrantholders or proxies of Registered Warrantholders.

7.9 Corporation and Warrant Agent May be Represented

The Corporation and the Warrant Agent, by their respective directors, officers, agents, and employees and the Counsel for the Corporation and for the Warrant Agent may attend any meeting of the Registered Warrantholders.

7.10 Powers Exercisable by Extraordinary Resolution

In addition to all other powers conferred upon them by any other provisions of this Indenture or by law, the Registered Warrantholders at a meeting shall, subject to the provisions of Section 7.11, have the power exercisable from time to time by Extraordinary Resolution:

- (a) to agree to any modification, abrogation, alteration, compromise or arrangement of the rights of Registered Warrantholders or the Warrant Agent in its capacity as warrant agent hereunder (subject to the Warrant Agent's prior consent, acting reasonably) or on behalf of the Registered Warrantholders against the Corporation whether such rights arise under this Indenture or otherwise;
- (b) to amend, alter or repeal any Extraordinary Resolution previously passed or sanctioned by the Registered Warrantholders;
- (c) to direct or to authorize the Warrant Agent, subject to Section 9.2(2) hereof, to enforce any of the covenants on the part of the Corporation contained in this Indenture or to enforce any of the rights of the Registered Warrantholders in any manner specified in such Extraordinary Resolution or to refrain from enforcing any such covenant or right;
- (d) to waive, and to direct the Warrant Agent to waive, any default on the part of the Corporation in complying with any provisions of this Indenture either unconditionally or upon any conditions specified in such Extraordinary Resolution;
- (e) to restrain any Registered Warrantholder from taking or instituting any suit, action or proceeding against the Corporation for the
 enforcement of any of the covenants on the part of the Corporation in this Indenture or to enforce any of the rights of the
 Registered Warrantholders;
- (f) to direct any Registered Warrantholder who, as such, has brought any suit, action or proceeding to stay or to discontinue or otherwise to deal with the same upon payment of the costs, charges and expenses reasonably and properly incurred by such Registered Warrantholder in connection therewith;
- (g) to assent to any change in or omission from the provisions contained in this Indenture or any ancillary or supplemental instrument which may be agreed to by the Corporation, and to authorize the Warrant Agent to concur in and execute any ancillary or supplemental indenture embodying the change or omission;

- (h) with the consent of the Corporation, such consent not to be unreasonably withheld, to remove the Warrant Agent or its successor in office and to appoint a new warrant agent or warrant agents to take the place of the Warrant Agent so removed; and
- to assent to any compromise or arrangement with any creditor or creditors or any class or classes of creditors, whether secured or otherwise, and with holders of any shares or other securities of the Corporation.

7.11 Meaning of Extraordinary Resolution

- (a) The expression "Extraordinary Resolution" when used in this Indenture means, subject as hereinafter provided in this Section 7.11 and in Section 7.14, a resolution proposed at a meeting of Registered Warrantholders duly convened for that purpose and held in accordance with the provisions of this Article 7 and passed by the affirmative votes of Registered Warrantholders holding not less than 66 2/3% of the aggregate number of Warrants represented in person or by proxy at the meeting and voted on the poll upon such resolution.
- (b) Subject to Section 7.14, votes on an Extraordinary Resolution shall always be given on a poll and no demand for a poll on an Extraordinary Resolution shall be necessary.

7.12 Powers Cumulative

Any one or more of the powers or any combination of the powers in this Indenture stated to be exercisable by the Registered Warrantholders by Extraordinary Resolution or otherwise may be exercised from time to time and the exercise of any one or more of such powers or any combination of powers from time to time shall not be deemed to exhaust the right of the Registered Warrantholders to exercise such power or powers or combination of powers then or thereafter from time to time.

7.13 Minutes

Minutes of all resolutions and proceedings at every meeting of Registered Warrantholders shall be made and duly entered in books to be provided from time to time for that purpose by the Warrant Agent at the expense of the Corporation, and any such minutes as aforesaid, if signed by the chairman or the secretary of the meeting at which such resolutions were passed or proceedings had shall be prima facie evidence of the matters therein stated and, until the contrary is proved, every such meeting in respect of the proceedings of which minutes shall have been made shall be deemed to have been duly convened and held, and all resolutions passed thereat or proceedings taken shall be deemed to have been duly passed and taken.

7.14 Instruments in Writing

All actions which may be taken and all powers that may be exercised by the Registered Warrantholders at a meeting held as provided in this Article 7 may also be taken and exercised by Registered Warrantholders holding not less than 66 2/3% of the aggregate number of all of the then outstanding Warrants by an instrument in writing signed in one or more counterparts by such Registered Warrantholders in person or by attorney duly appointed in writing, and the expression "Extraordinary Resolution" when used in this Indenture shall include an instrument so signed.

7.15 Binding Effect of Resolutions

Every resolution and every Extraordinary Resolution passed in accordance with the provisions of this Article 7 at a meeting of Registered Warrantholders shall be binding upon all the Warrantholders, whether present at or absent from such meeting, and every instrument in writing signed by Registered Warrantholders in accordance with Section 7.14 shall be binding upon all the Warrantholders, whether signatories thereto or not, and each and every Warrantholder and the Warrant Agent (subject to the provisions for indemnity herein contained) shall be bound to give effect accordingly to every such resolution and instrument in writing.

7.16 Holdings by Corporation Disregarded

In determining whether Registered Warrantholders holding Warrants evidencing the entitlement to acquire the required number of Common Shares are present at a meeting of Registered Warrantholders for the purpose of determining a quorum or have concurred in any consent, waiver, Extraordinary Resolution, Warrantholders' Request or other action under this Indenture, Warrants owned legally or beneficially by the Corporation shall be disregarded in accordance with the provisions of Section 10.7.

ARTICLE 8 SUPPLEMENTAL INDENTURES

8.1 Provision for Supplemental Indentures for Certain Purposes

From time to time, the Corporation (when authorized by action of the directors) and the Warrant Agent may, subject to the provisions hereof and they shall, when so directed in accordance with the provisions hereof, execute and deliver by their proper officers, indentures or instruments supplemental hereto, which thereafter shall form part hereof, for any one or more or all of the following purposes:

- (a) setting forth any adjustments resulting from the application of the provisions of Article 4;
- (b) adding to the provisions hereof such additional covenants and enforcement provisions as, in the opinion of Counsel, are necessary or advisable in the premises, provided that the same are not in the opinion of the Warrant Agent, relying on the advice of Counsel, prejudicial to the interests of the Registered Warrantholders;
- (c) giving effect to any Extraordinary Resolution passed as provided in Section 7.11;
- (d) making such provisions not inconsistent with this Indenture as may be necessary or desirable with respect to matters or questions arising hereunder or for the purpose of obtaining a listing or quotation of the Warrants on any stock exchange or quotation system, provided that such provisions are not, in the opinion of the Warrant Agent, relying on the advice of Counsel, prejudicial to the interests of the Registered Warrantholders;
- (e) adding to or altering the provisions hereof in respect of the transfer of Warrants, making provision for the exchange of Warrants, and making any modification in the form of the Warrant Certificates which does not affect the substance thereof;
- (f) modifying any of the provisions of this Indenture, including relieving the Corporation from any of the obligations, conditions or restrictions herein contained, provided that

such modification or relief shall be or become operative or effective only if, in the opinion of the Warrant Agent, relying on the advice of Counsel, such modification or relief in no way prejudices any of the rights of the Registered Warrantholders or of the Warrant Agent, and provided further that the Warrant Agent may in its sole discretion decline to enter into any such supplemental indenture which in its opinion may not afford adequate protection to the Warrant Agent when the same shall become operative;

- (g) providing for the issuance of additional Warrants hereunder, including Warrants in excess of the number set out in Section 2.1 and any consequential amendments hereto as may be required by the Warrant Agent relying on the advice of Counsel.
- (h) for any other purpose not inconsistent with the terms of this Indenture, including the correction or rectification of any ambiguities, defective or inconsistent provisions, errors, mistakes or omissions herein, provided that in the opinion of the Warrant Agent, relying on the advice of Counsel, the rights of the Warrant Agent and of the Registered Warrantholders are in no way prejudiced thereby.

8.2 Successor Entities

In the case of the consolidation, amalgamation, arrangement, merger or transfer of the undertaking or assets of the Corporation as an entirety or substantially as an entirety to or with another entity ("successor entity"), the successor entity resulting from such consolidation, amalgamation, arrangement, merger or transfer (if not the Corporation) shall expressly assume, by supplemental indenture satisfactory in form to the Warrant Agent and executed and delivered to the Warrant Agent, the due and punctual performance and observance of each and every covenant and condition of this Indenture to be performed and observed by the Corporation.

ARTICLE 9 CONCERNING THE WARRANT AGENT

9.1 Trust Indenture Legislation

- (a) If and to the extent that any provision of this Indenture limits, qualifies or conflicts with a mandatory requirement of Applicable Legislation, such mandatory requirement shall prevail.
- (b) The Corporation and the Warrant Agent agree that each will, at all times in relation to this Indenture and any action to be taken hereunder, observe and comply with and be entitled to the benefits of Applicable Legislation.

9.2 Rights and Duties of Warrant Agent

- (a) In the exercise of the rights and duties prescribed or conferred by the terms of this Indenture, the Warrant Agent shall exercise that degree of care, diligence and skill that a reasonably prudent warrant agent would exercise in comparable circumstances. No provision of this Indenture shall be construed to relieve the Warrant Agent from liability for its own negligent action, wilful misconduct, bad faith or fraud under this Indenture.
- (b) The obligation of the Warrant Agent to commence or continue any act, action or proceeding for the purpose of enforcing any rights of the Warrant Agent or the Registered Warrantholders hereunder shall be conditional upon the Registered Warrantholders

furnishing, when required by notice by the Warrant Agent, sufficient funds to commence or to continue such act, action or proceeding and an indemnity reasonably satisfactory to the Warrant Agent to protect and to hold harmless the Warrant Agent and its officers, directors, employees and agents, against the costs, charges and expenses and liabilities to be incurred thereby and any loss and damage it may suffer by reason thereof. None of the provisions contained in this Indenture shall require the Warrant Agent to expend or to risk its own funds or otherwise to incur financial liability in the performance of any of its duties or in the exercise of any of its rights or powers unless indemnified and funded as aforesaid.

- (c) The Warrant Agent may, before commencing or at any time during the continuance of any such act, action or proceeding, require the Registered Warrantholders, at whose instance it is acting to deposit with the Warrant Agent the Warrants Certificates held by them, for which Warrants the Warrant Agent shall issue receipts.
- (d) Every provision of this Indenture that by its terms relieves the Warrant Agent of liability or entitles it to rely upon any evidence submitted to it is subject to the provisions of Applicable Legislation.

9.3 Evidence, Experts and Advisers

- (a) In addition to the reports, certificates, opinions and other evidence required by this Indenture, the Corporation shall furnish to the Warrant Agent such additional evidence of compliance with any provision hereof, and in such form, as may be prescribed by Applicable Legislation or as the Warrant Agent may reasonably require by written notice to the Corporation.
- (b) In the exercise of its rights and duties hereunder, the Warrant Agent may, if it is acting in good faith, act and rely as to the truth of the statements and the accuracy of the opinions expressed in statutory declarations, opinions, reports, written requests, consents, or orders of the Corporation, certificates of the Corporation or other evidence furnished to the Warrant Agent pursuant to a request of the Warrant Agent, provided that such evidence complies with Applicable Legislation and that the Warrant Agent complies with Applicable Legislation and that the Warrant Agent examines the same and determines that such evidence complies with the applicable requirements of this Indenture.
- (c) The Warrant Agent may employ or retain such Counsel, accountants, appraisers or other experts or advisers as it may reasonably require for the purpose of determining and discharging its duties hereunder and may pay reasonable remuneration for all services so performed by any of them, without taxation of costs of any Counsel, and shall not be responsible for any misconduct or negligence on the part of any such experts or advisers who have been appointed with due care by the Warrant Agent.
- (d) The Warrant Agent may act and rely and shall be protected in acting and relying in good faith on the opinion or advice of or information obtained from any Counsel, accountant, appraiser, engineer or other expert or adviser, whether retained or employed by the Corporation or by the Warrant Agent, in relation to any matter arising in the administration of the agency hereof.
- (e) The Warrant Agent may act and rely and shall be protected in acting and relying upon any resolution, certificate, statement, instrument, opinion, report, notice, request, consent, order, letter, telegram, cablegram or other paper document believed by it to be genuine and to have been signed, sent or presented by or on behalf of the proper party or parties.

- (f) Whenever Applicable Legislation requires that evidence referred to in subsection (i) be in the form of a statutory declaration, the Warrant Agent may accept such statutory declaration in lieu of a certificate of the Corporation required by any provision hereof. Any such statutory declaration may be made by any one or more of the Chairman or Chief Financial Officer of the Corporation or by any other officer(s) or director(s) of the Corporation to whom such authority is delegated by the directors from time to time.
- (g) Proof of execution of any document or instrument in writing; including a Registered Warrantholders' Request, by a holder may be made by the certificate of a notary public, or other officer with similar powers, that the person signing such instrument acknowledged to him the execution thereof, or by an affidavit of a witness to such execution, or in any other manner the Warrant Agent considers adequate.

9.4 Documents, Monies, etc. Held by Warrant Agent

- (a) Any monies, securities, documents of title or other instruments that may at any time be held by the Warrant Agent shall be placed in the deposit vaults of the Warrant Agent or of any Canadian chartered bank listed in Schedule I of the Bank Act (Canada), or deposited for safekeeping with any such bank. Any monies held pending the application or withdrawal thereof under any provisions of this Indenture, shall be held, invested and reinvested in "Permitted Investments" as directed in writing by the Corporation. "Permitted Investments" shall be treasury bills guaranteed by the Government of Canada having a term to maturity not to exceed 90 days, or term deposits or bankers' acceptances of a Canadian chartered bank having a term to maturity not to exceed 90 days, or such other investments that is in accordance with the Warrant Agent's standard type of investments. Unless otherwise specifically provided herein, all interest or other income received by the Warrant Agent in respect of such deposits and investments shall belong to the Corporation.
- (b) Any written direction for the investment or release of funds received shall be received by the Warrant Agent by 9:00a.m. (Vancouver time) on the Business Day on which such investment or release is to be made, failing which such direction will be handled on a commercially reasonable efforts basis and may result in funds being invested or released on the next Business Day.
- (c) The Warrant Agent shall have no responsibility or liability for any diminution of any funds resulting from any investment made in accordance with this Indenture, including any losses on any investment liquidated prior to maturity in order to make a payment required hereunder.
- (d) In the event that the Warrant Agent does not receive a direction or only a partial direction, the Warrant Agent may hold cash balances constituting part or all of such monies and may, but need not, invest same in its deposit department, the deposit department of one of its affiliates, or the deposit department of a Canadian chartered bank; but the Warrant Agent, its affiliates or a Canadian chartered bank shall not be liable to account for any profit to any parties to this Indenture or to any other person or entity.

9.5 Actions by Warrant Agent to Protect Interest

The Warrant Agent shall have power to institute and to maintain such actions and proceedings as it may consider necessary or expedient to preserve, protect or enforce its interests and the interests of the Registered Warrantholders.

9.6 Warrant Agent Not Required to Give Security

The Warrant Agent shall not be required to give any bond or security in respect of the execution of the agency and powers of this Indenture or otherwise in respect of the premises.

9.7 Protection of Warrant Agent

By way of supplement to the provisions of any law for the time being relating to the Warrant Agent it is expressly declared and agreed as follows:

- (a) the Warrant Agent shall not be liable for or by reason of any statements of fact or recitals in this Indenture or in the Warrant Certificates (except the representation contained in Section 9.9 or in the authentication of the Warrant Agent on the Warrant Certificates) or be required to verify the same, but all such statements or recitals are and shall be deemed to be made by the Corporation;
- (b) nothing herein contained shall impose any obligation on the Warrant Agent to see to or to require evidence of the registration or filing (or renewal thereof) of this Indenture or any instrument ancillary or supplemental hereto;
- (c) the Warrant Agent shall not be bound to give notice to any person or persons of the execution hereof;
- (d) the Warrant Agent shall not incur any liability or responsibility whatever or be in any way responsible for the consequence of any breach on the part of the Corporation of any of its covenants herein contained or of any acts of any directors, officers, employees, agents or servants of the Corporation;
- (e) the Corporation will fully indemnify and hold the Warrant Agent and its officers, directors, employees and agents harmless from and against any and all actions and suits whether groundless or otherwise and from and against any and all losses, damages, costs, charges, counsel fees, payments, expenses and liabilities arising directly or indirectly out the performance of its duties and obligations under this indenture, except for any liability arising out of the Warrant Agent's gross negligence or intentional misconduct;
- (f) this indemnification shall survive the resignation or removal of the Warrant Agent or the termination or discharge of this Indenture; and
- (g) notwithstanding any other provision of this Indenture, and whether such losses or damages are foreseeable or unforeseeable, the Warrant Agent shall not be liable under any circumstances whatsoever for any (a) breach by any other party of securities law or other rule of any securities regulatory authority, (b) lost profits (c) any action taken, suffered, or omitted by it or for any error of judgement made by the Warrant Agent in the performance of its duties under this Agreement except for gross negligence on the part of the Warrant Agent or (d) special, indirect, incidental, consequential, exemplary, aggravated or punitive losses or damages.

9.8 Replacement of Warrant Agent; Successor by Merger

- (a) The Warrant Agent may resign its agency and be discharged from all further duties and liabilities hereunder, subject to this Section 9.8, by giving to the Corporation not less than 60 days' prior notice in writing or such shorter prior notice as the Corporation may accept as sufficient. The Registered Warrantholders by Extraordinary Resolution shall have power at any time to remove the existing Warrant Agent and to appoint a new warrant agent. In the event of the Warrant Agent resigning or being removed as aforesaid or being dissolved, becoming bankrupt, going into liquidation or otherwise becoming incapable of acting hereunder, the Corporation shall forthwith appoint a new warrant agent unless a new warrant agent has already been appointed by the Registered Warrantholders; failing such appointment by the Corporation, the retiring Warrant Agent or any Registered Warrantholder may apply to a judge of a court of the Province of British Columbia on such notice as such judge may direct, for the appointment of a new warrant agent; but any new warrant agent so appointed by the Corporation or by the Court shall be subject to removal as aforesaid by the Registered Warrantholders. Any new warrant agent appointed under any provision of this Section 9.8 shall be an entity authorized to carry on the business of a trust company in each of the provinces of Canada. On any such appointment the new warrant agent shall be vested with the same powers, rights, duties and responsibilities as if it had been originally named herein as Warrant Agent hereunder.
- (b) Upon the appointment of a successor warrant agent, the Corporation shall promptly notify the Registered Warrantholders thereof in the manner provided for in Section 10.2.
- (c) Any Warrant Certificates Authenticated but not delivered by a predecessor Warrant Agent may be Authenticated by the successor Warrant Agent in the name of the predecessor or successor Warrant Agent.
- (d) Any corporation into which the Warrant Agent may be merged or consolidated or amalgamated, or any corporation resulting therefrom to which the Warrant Agent shall be a party, or any corporation succeeding to substantially the corporate trust business of the Warrant Agent shall be the successor to the Warrant Agent hereunder without any further act on its part or any of the parties hereto, provided that such corporation would be eligible for appointment as successor Warrant Agent under Section 9.8(1).

9.9 Conflict of Interest

(a) The Warrant Agent represents to the Corporation that at the time of execution and delivery hereof no material conflict of interest exists between its role as a Warrant Agent hereunder and its role in any other capacity and agrees that in the event of a material conflict of interest arising hereafter it will, within 90 days after ascertaining that it has such material conflict of interest, either eliminate the same or assign its agency hereunder to a successor Warrant Agent approved by the Corporation and meeting the requirements set forth in Section 9.8(1). Notwithstanding the foregoing provisions of this Section 9.9(1), if any such material conflict of interest exists or hereafter shall exist, the validity and enforceability of this Indenture and the Warrant Certificate shall not be affected in any manner whatsoever by reason thereof.

(b) Subject to Section 9.9(1), the Warrant Agent, in its personal or any other capacity, may buy, lend upon and deal in securities of the Corporation and generally may contract and enter into financial transactions with the Corporation without being liable to account for any profit made thereby.

9.10 Acceptance of Agency

The Warrant Agent hereby accepts the agency in this Indenture declared and provided for and agrees to perform the same upon the terms and conditions herein set forth.

9.11 Warrant Agent Not to be Appointed Receiver

The Warrant Agent and any person related to the Warrant Agent shall not be appointed a receiver, a receiver and manager or liquidator of all or any part of the assets or undertaking of the Corporation.

9.12 Warrant Agent Not Required to Give Notice of Default

The Warrant Agent shall not be bound to give any notice or do or take any act, action or proceeding by virtue of the powers conferred on it hereby unless and until it shall have been required so to do under the terms hereof; nor shall the Warrant Agent be required to take notice of any default hereunder, unless and until notified in writing of such default, which notice shall distinctly specify the default desired to be brought to the attention of the Warrant Agent and in the absence of any such notice the Warrant Agent may for all purposes of this Indenture conclusively assume that no default has been made in the observance or performance of any of the representations, warranties, covenants, agreements or conditions contained herein. Any such notice shall in no way limit any discretion herein given to the Warrant Agent to determine whether or not the Warrant Agent shall take action with respect to any default.

9.13 Anti-Money Laundering

- (a) Each party to this Agreement other than the Warrant Agent hereby represents to the Warrant Agent that any account to be opened by, or interest to be held by the Warrant Agent in connection with this Agreement, for or to the credit of such party, either (i) is not intended to be used by or on behalf of any third party; or (ii) is intended to be used by or on behalf of a third party, in which case such party hereto agrees to complete and execute forthwith a declaration in the Warrant Agent's prescribed form as to the particulars of such third party.
- (b) The Warrant Agent shall retain the right not to act and shall not be liable for refusing to act if, due to a lack of information or for any other reason whatsoever, the Warrant Agent, in its sole judgment, determines that such act might cause it to be in non-compliance with any applicable anti-money laundering, anti-terrorist or economic sanctions legislation, regulation or guideline. Further, should the Warrant Agent, in its sole judgment, determine at any time that its acting under this Agreement has resulted in its being in non-compliance with any applicable anti-money laundering, anti-terrorist or economic sanctions legislation, regulation or guideline, then it shall have the right to resign on 10 days written notice to the other parties to this Agreement, provided (i) that the Warrant Agent's written notice shall describe the circumstances of such non-compliance; and (ii) that if such circumstances are rectified to the Warrant Agent's satisfaction within such 10 day period, then such resignation shall not be effective.

9.14 Compliance with Privacy Code

The Corporation acknowledges that the Warrant Agent may, in the course of providing services hereunder, collect or receive financial and other personal information about such parties and/or their representatives, as individuals, or about other individuals related to the subject matter hereof, and use such information for the following purposes:

- (a) to provide the services required under this Indenture and other services that may be requested from time to time;
- (b) to help the Warrant Agent manage its servicing relationships with such individuals;
- (c) to meet the Warrant Agent's legal and regulatory requirements; and
- (d) if Social Insurance Numbers are collected by the Warrant Agent, to perform tax reporting and to assist in verification of an individual's identity for security purposes.

The Corporation acknowledges and agrees that the Warrant Agent may receive, collect, use and disclose personal information provided to it or acquired by it in the course of its acting as agent hereunder for the purposes described above and, generally, in the manner and on the terms described in its Privacy Code, which the Warrant Agent shall make available on its website or upon request, including revisions thereto. Further, the Corporation agrees that it shall not provide or cause to be provided to the Warrant Agent any personal information relating to an individual who is not a party to this Indenture unless the Corporation has assured itself that such individual understands and has consented to the aforementioned uses and disclosures.

9.15 Securities Exchange Commission Certification

(a) Currently reporting:

The Corporation confirms that it has either (i) a class of securities registered pursuant to Section 12 of the US Securities Exchange Act of 1934, as amended (the "Act"); or (ii) a reporting obligation pursuant to Section 15(d) of the Act. The Corporation covenants that in the event that any such registration or reporting obligation shall be terminated by the Corporation in accordance with the Act, the Corporation shall promptly notify the Warrant Agent of such termination and such other information as the Warrant Agent may require at the time. The Corporation acknowledges that the Warrant Agent is relying upon the foregoing representation and covenants in order to meet certain United States SEC obligations with respect to those clients who are filing with the SEC.

ARTICLE 10 GENERAL

10.1 Notice to the Corporation and the Warrant Agent

- (a) Unless herein otherwise expressly provided, any notice to be given hereunder to the Corporation or the Warrant Agent shall be deemed to be validly given if delivered, sent by registered letter, postage prepaid or if faxed:
 - (i) If to the Corporation:

Cohbar, Inc. PO Box 955 Mill Valley, CA 94942

Attention: President and Chief Executive Officer

Facsimile Number: (415) 381-8980 Email: jon.stern@cohbar.com

With a copy to:

Peter B. Cancelmo Garvey Schubert Barer 1191 Second Avenue, 18th Floor Seattle, WA 98101 Facsimile Number: 206-464-012

Facsimile Number: 206-464-0125 Email: pcancelmo@gsblaw.com

(ii) If to the Warrant Agent:

CST Trust Company 1600-1066 West Hastings St. Vancouver, British Columbia V6E-VX1

VOL-VAI

Attention: Director, Relationship Management

Facsimile Number: 604-235-3705

and any such notice delivered in accordance with the foregoing shall be deemed to have been received and given on the date of delivery or, if mailed, on the fifth Business Day following the date of mailing such notice or, if faxed or sent by electronic transmission with confirmation of receipt, on the next Business Day following the date of transmission.

- (b) The Corporation or the Warrant Agent, as the case may be, may from time to time notify the other in the manner provided in Section 10.1(1) of a change of address which, from the effective date of such notice and until changed by like notice, shall be the address of the Corporation or the Warrant Agent, as the case may be, for all purposes of this Indenture.
- (c) If, by reason of a strike, lockout or other work stoppage, actual or threatened, involving postal employees, any notice to be given to the Warrant Agent or to the Corporation hereunder could reasonably be considered unlikely to reach its destination, such notice shall be valid and effective only if it is delivered to the named officer of the party to which it is addressed, as provided in Section 10.1(1), or given by facsimile or other means of prepaid, transmitted and recorded communication.

10.2 Notice to Registered Warrantholders

(a) Unless otherwise provided herein, notice to the Registered Warrantholders under the provisions of this Indenture shall be valid and effective if delivered or sent by ordinary prepaid post addressed to such holders at their post office addresses appearing on the register hereinbefore mentioned and shall be deemed to have been effectively received

and given on the date of delivery or, if mailed, on the third Business Day following the date of mailing such notice. In the event that Warrants are held in the name of the Depository, a copy of such notice shall also be sent by electronic communication to the Depository and shall be deemed received and given on the day it is so sent.

(b) If, by reason of a strike, lockout or other work stoppage, actual or threatened, involving postal employees, any notice to be given to the Registered Warrantholders hereunder could reasonably be considered unlikely to reach its destination, such notice shall be valid and effective only if it is delivered to such Registered Warrantholders to the address for such Registered Warrantholders contained in the register maintained by the Warrant Agent or such notice may be given, at the Corporation's expense, by means of publication in the Globe and Mail, National Edition, or any other English language daily newspaper or newspapers of general circulation in Canada, in each two successive weeks, the first such notice to be published within 5 business days of such event, and any so notice published shall be deemed to have been received and given on the latest date the publication takes place.

10.3 Ownership of Warrants

The Corporation and the Warrant Agent may deem and treat the Registered Warrantholders as the absolute owner thereof for all purposes, and the Corporation and the Warrant Agent shall not be affected by any notice or knowledge to the contrary except where the Corporation or the Warrant Agent is required to take notice by statute or by order of a court of competent jurisdiction. The receipt of any such Registered Warrantholder of the Common Shares which may be acquired pursuant thereto shall be a good discharge to the Corporation and the Warrant Agent for the same and neither the Corporation nor the Warrant Agent shall be bound to inquire into the title of any such holder except where the Corporation or the Warrant Agent is required to take notice by statute or by order of a court of competent jurisdiction.

10.4 Counterparts

This Indenture may be executed in several counterparts, each of which when so executed shall be deemed to be an original and such counterparts together shall constitute one and the same instrument and notwithstanding their date of execution they shall be deemed to be dated as of the date hereof.

10.5 Satisfaction and Discharge of Indenture

Upon the earlier of:

- (a) the date by which there shall have been delivered to the Warrant Agent for exercise or cancellation all Warrants theretofore Authenticated hereunder, in the case of Certificated Warrants (or such other instructions, in a form satisfactory to the Warrant Agent), in the case of Uncertificated Warrants, or by way of standard processing through the book entry only system in the case of a CDS Global Warrant; and
- (b) the Expiry Time;

and if all certificates or other entry on the register representing Common Shares required to be issued in compliance with the provisions hereof have been issued and delivered hereunder or to the Warrant Agent in accordance with such provisions, this Indenture shall cease to be of further

effect and the Warrant Agent, on demand of and at the cost and expense of the Corporation and upon delivery to the Warrant Agent of a certificate of the Corporation stating that all conditions precedent to the satisfaction and discharge of this Indenture have been complied with, shall execute proper instruments acknowledging satisfaction of and discharging this Indenture. Notwithstanding the foregoing, the indemnities provided to the Warrant Agent by the Corporation hereunder shall remain in full force and effect and survive the termination of this Indenture.

10.6 Provisions of Indenture and Warrants for the Sole Benefit of Parties and Registered Warrantholders

Nothing in this Indenture or in the Warrants, expressed or implied, shall give or be construed to give to any person other than the parties hereto and the Registered Warrantholders, as the case may be, any legal or equitable right, remedy or claim under this Indenture, or under any covenant or provision herein or therein contained, all such covenants and provisions being for the sole benefit of the parties hereto and the Registered Warrantholders.

10.7 Common Shares or Warrants Owned by the Corporation or its Subsidiaries—Certificate to be Provided

For the purpose of disregarding any Warrants owned legally or beneficially by the Corporation in Section 7.16, the Corporation shall provide to the Warrant Agent, from time to time, a certificate of the Corporation setting forth as at the date of such certificate:

- (a) the names (other than the name of the Corporation) of the Registered Warrantholders which, to the knowledge of the Corporation, are owned by or held for the account of the Corporation; and
- (b) the number of Warrants owned legally or beneficially by the Corporation;

and the Warrant Agent, in making the computations in Section 7.16, shall be entitled to rely on such certificate without any additional evidence.

10.8 Severability

If, in any jurisdiction, any provision of this Indenture or its application to any party or circumstance is restricted, prohibited or unenforceable, such provision will, as to such jurisdiction, be ineffective only to the extent of such restriction, prohibition or unenforceability without invalidating the remaining provisions of this Indenture and without affecting the validity or enforceability of such provision in any other jurisdiction or without affecting its application to other parties or circumstances.

10.9 Force Majeure

No party shall be liable to the other, or held in breach of this Indenture, if prevented, hindered, or delayed in the performance or observance of any provision contained herein by reason of act of God, riots, terrorism, acts of war, epidemics, governmental action or judicial order, earthquakes, or any other similar causes (including, but not limited to, mechanical, electronic or communication interruptions, disruptions or failures). Performance times under this Indenture shall be extended for a period of time equivalent to the time lost because of any delay that is excusable under this Section.

10.10 Assignment, Successors and Assigns

Neither of the parties hereto may assign its rights or interest under this Indenture, except as provided in Section 9.8 in the case of the Warrant Agent, or as provided in Section 8.2 in the case of the Corporation. Subject thereto, this Indenture shall enure to the benefit of and be binding upon the parties hereto and their respective successors and permitted assigns.

10.11 Rights of Rescission and Withdrawal for Holders

Should a holder of Warrants exercise any legal, statutory, contractual or other right of withdrawal or rescission that may be available to it, and the holder's funds which were paid on exercise have already been released to the Corporation by the Warrant Agent, the Warrant Agent shall not be responsible for ensuring the exercise is cancelled and a refund is paid back to the holder. In such cases, the holder shall seek a refund directly from the Corporation and subsequently, the Corporation, upon surrender to the Corporation or the Warrant Agent of any underlying shares that may have been issued, or such other procedure as agreed to by the parties hereto, shall instruct the Warrant Agent in writing, to cancel the exercise transaction and any such underlying shares on the register, which may have already been issued upon the Warrant exercise. In the event that any payment is received from the Corporation by virtue of the holder being a shareholder for such Warrants that were subsequently rescinded, such payment must be returned to the Corporation by such holder. The Warrant Agent shall not be under any duty or obligation to take any steps to ensure or enforce that the funds are returned pursuant to this section, nor shall the Warrant Agent be in any other way responsible in the event that any payment is not delivered or received pursuant to this section. Notwithstanding the foregoing, in the event that the Corporation provides the refund to the Warrant Agent for distribution to the holder, the Warrant Agent shall return such funds to the holder as soon as reasonably practicable, and in so doing, the Warrant Agent shall incur no liability with respect to the delivery or non-delivery of any such funds.

IN WITNESS WHEREOF the parties hereto have executed this Indenture under the hands of their proper officers in that behalf as of the date first written above.

COHBAR, INC.

By: /s/ Jon Stern

Name: Jon Stern

Title: Chief Executive Officer

By: /s/ Jeffrey Biunno

Name: Jeffrey Biunno Title: Chief Financial Officer

CST TRUST COMPANY

By: /s/ Van Bot

Name: Van Bot

Title: Authorized Signatory

By: /s/ Leslie MacFarlane

Name: Leslie MacFarlane Title: Authorized Signatory

SCHEDULE "A" FORM OF WARRANT

SUBJECT TO THE COMPANY'S ACCELERATION RIGHT, THE WARRANTS EVIDENCED HEREBY ARE EXERCISABLE AT OR BEFORE 5:00 P.M. (TORONTO TIME) ON JANUARY 6, 2017 AFTER WHICH TIME THE WARRANTS EVIDENCED HEREBY SHALL BE DEEMED TO BE VOID AND OF NO FURTHER FORCE OR EFFECT.

For all IPO Warrants and Compensation Unit Warrants sold outside the United States and registered in the name of the Depository, include the following legend:

(INSERT IF BEING ISSUED TO CDS) UNLESS THIS CERTIFICATE IS PRESENTED BY AN AUTHORIZED REPRESENTATIVE OF CDS CLEARING AND DEPOSITORY SERVICES INC. ("CDS") TO COHBAR, INC. (THE "ISSUER") OR ITS AGENT FOR REGISTRATION OF TRANSFER, EXCHANGE OR PAYMENT, AND ANY CERTIFICATE ISSUED IN RESPECT THEREOF IS REGISTERED IN THE NAME OF CDS & CO., OR SUCH OTHER NAME AS IS REQUESTED BY AN AUTHORIZED REPRESENTATIVE OF CDS (AND ANY PAYMENT IS MADE TO CDS & CO. OR TO SUCH OTHER ENTITY AS IS REQUESTED BY AN AUTHORIZED REPRESENTATIVE OF CDS), ANY TRANSFER, PLEDGE OR OTHER USE HEREOF FOR VALUE OR OTHERWISE BY OR TO ANY PERSON IS WRONGFUL SINCE THE REGISTERED HOLDER HEREOF, CDS, HAS A PROPERTY INTEREST IN THE SECURITIES REPRESENTED BY THIS CERTIFICATE HEREIN AND IT IS A VIOLATION OF ITS RIGHTS FOR ANOTHER PERSON TO HOLD, TRANSFER OR DEAL WITH THIS CERTIFICATE.

For all Put Unit Warrants include the following legends:

(INSERT INTO PUT UNIT WARRANTS) THESE WARRANTS AND THE SECURITIES DELIVERABLE UPON THE EXERCISE THEREOF HAVE NOT BEEN AND WILL NOT BE REGISTERED UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "U.S. SECURITIES ACT"). THESE SECURITIES MAY BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED ONLY (A) TO THE CORPORATION, (B) IF THE SECURITIES HAVE BEEN REGISTERED IN COMPLIANCE WITH THE REGISTRATION REQUIREMENTS UNDER THE U.S. SECURITIES ACT AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS (C) IN COMPLIANCE WITH THE EXEMPTION FROM THE REGISTRATION REQUIREMENTS UNDER THE U.S. SECURITIES ACT IN ACCORDANCE WITH RULE 144 THEREUNDER, IF APPLICABLE, AND IN ACCORDANCE WITH APPLICABLE STATE SECURITIES LAWS, OR (D) IN A TRANSACTION THAT DOES NOT REQUIRE REGISTRATION UNDER THE U.S. SECURITIES ACT OR ANY APPLICABLE STATE LAWS AND REGULATIONS GOVERNING THE OFFER AND SALE OF SECURITIES, AND THE HOLDER HAS, PRIOR TO SUCH SALE, FURNISHED TO THE CORPORATION AN OPINION OF COUNSEL OF RECOGNIZED STANDING, OR OTHER EVIDENCE OF EXEMPTION, REASONABLY SATISFACTORY TO THE CORPORATION. HEDGING TRANSACTIONS INVOLVING THE SECURITIES REPRESENTED HEREBY MAY NOT BE CONDUCTED UNLESS IN COMPLIANCE WITH U.S. SECURITIES LAWS.

THE SECURITIES EVIDENCED HEREBY AND THE SECURITIES ISSUABLE UPON EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE U.S. SECURITIES ACT OR U.S. STATE SECURITIES LAWS. THESE WARRANTS MAY NOT BE EXERCISED IN THE UNITED STATES OR BY OR ON BEHALF OF, OR FOR THE ACCOUNT OR BENEFIT OF, A U.S. PERSON UNLESS THIS SECURITY AND THE COMMON SHARES ISSUABLE UPON EXERCISE OF THIS SECURITY HAVE BEEN REGISTERED UNDER THE U.S. SECURITIES ACT AND THE

APPLICABLE STATE SECURITIES LEGISLATION OR AN EXEMPTION FROM SUCH REGISTRATION REQUIREMENTS IS AVAILABLE. "UNITED STATES" AND "U.S. PERSON" ARE AS DEFINED BY REGULATION S UNDER THE U.S. SECURITIES ACT.

WARRANT

To acquire Common Shares of

COHBAR, INC.

(incorporated pursuant to the laws of the State of Delaware)

Warrant Certificate No. <@>

Certificate for <@>

Warrants, each entitling the holder to acquire one (1) Common Share (subject to adjustment as provided for in the Warrant Indenture (as defined below)

CUSIP 19249J117

ISIN CA US19249J117

THIS IS TO CERTIFY THAT, for value received,

(the "Warrantholder") is the registered holder of the number of common share purchase warrants (the "Warrants") of Cohbar, Inc. (the "Corporation") specified above, and is entitled, on exercise of these Warrants upon and subject to the terms and conditions set forth herein and in the Warrant Indenture, to purchase for each Warrant, at any time before 5:00 p.m. (Toronto time) on January 6, 2017 (the "Expiry Time"), subject to the Acceleration Right, one fully paid and non-assessable share of the Corporation's common stock, US\$0.001 par value per share (a "Common Share"), subject to adjustment in accordance with the terms of the Warrant Indenture.

For the purpose of this Warrant Certificate and the Warrant Indenture, "Acceleration Right" means the right of the Company to accelerate the Expiry Time to a date that is not the less than 30 days following delivery of the Acceleration Notice if, at any time after the Common Shares are first traded on the TSXV, the volume weighted average trading price of the Common Shares on the TSXV, or if the Common Shares are not then listed on the TSXV, on such other stock exchange on which the Common Shares are principally traded, equals or exceeds \$3.00 for 20 consecutive Trading Days, provided that the Corporation provides the Acceleration Notice to each registered holder of Warrants within five (5) Business Days.

The right to purchase Common Shares may only be exercised by the Warrantholder within the time set forth above by:

- (a) duly completing and executing the exercise form (the "Exercise Form") attached hereto; and
- (b) surrendering this warrant certificate (the "Warrant Certificate"), with the Exercise Form to the Warrant Agent at the principal office of the Warrant Agent, in the city of Vancouver, British Columbia, together with a certified cheque, bank draft or money order in the lawful money of the United States payable to or to the order of the Corporation in an amount equal to the Exercise Price of the Common Shares so subscribed for.

Subject to adjustment thereof in the events and in the manner set forth in the Warrant Indenture hereinafter referred to, the exercise price payable for each Common Share upon the exercise of Warrants shall be \$2.00 per Common Share (the "Exercise Price").

The surrender of this Warrant Certificate, the duly completed Exercise Form and payment as provided above will be deemed to have been effected only on personal delivery thereof to, or if sent by mail or other means of transmission on actual receipt thereof by, the Warrant Agent at its principal office as set out above.

Certificates for the Common Shares subscribed for will be mailed to the persons specified in the Exercise Form at their respective addresses specified therein or, if so specified in the Exercise Form, delivered to such persons at the office where this Warrant Certificate is surrendered. If fewer Common Shares are purchased than the number that can be purchased pursuant to this Warrant Certificate, the holder hereof will be entitled to receive without charge a new Warrant Certificate in respect of the balance of the Common Shares not so purchased. No fractional Common Shares will be issued upon exercise of any Warrant.

This Warrant Certificate evidences Warrants of the Corporation issued or issuable under the provisions of a warrant indenture (which indenture together with all other instruments supplemental or ancillary thereto is herein referred to as the "Warrant Indenture") dated as of January 6, 2015 between the Corporation and CST Trust Company, as Warrant Agent, to which Warrant Indenture reference is hereby made for particulars of the rights of the holders of Warrants, the Corporation and the Warrant Agent in respect thereof and the terms and conditions on which the Warrants are issued and held, all to the same effect as if the provisions of the Warrant Indenture were herein set forth, to all of which the holder, by acceptance hereof, assents. The Corporation will furnish to the holder, on request and without charge, a copy of the Warrant Indenture.

On presentation at the principal office of the Warrant Agent as set out above, subject to the provisions of the Warrant Indenture and on compliance with the reasonable requirements of the Warrant Agent, one or more Warrant Certificates may be exchanged for one or more Warrant Certificates entitling the holder thereof to purchase in the aggregate an equal number of Common Shares as are purchasable under the Warrant Certificate(s) so exchanged.

If the issuance of Common Shares upon exercise hereof is not registered under an effective Registration Statement under the United States Securities Act of 1933, as amended (the "U.S. Securities Act"), or U.S. state securities laws, then these Warrants may not be exercised in the United States or by or on behalf of, or for the account or benefit of, a U.S. Person or a person in the United States absent an exemption from the registration requirements of the U.S. Securities Act and applicable state securities law.

The Warrant Indenture contains provisions for the adjustment of the Exercise Price payable for each Common Share upon the exercise of Warrants and the number of Common Shares issuable upon the exercise of Warrants upon the occurrence of the events and in the manner set forth therein.

The Warrant Indenture also contains provisions making binding on all holders of Warrants outstanding thereunder resolutions passed at meetings of holders of Warrants held in accordance with the provisions of the Warrant Indenture and instruments in writing signed by Warrantholders of Warrants entitled to purchase a specific majority of the Common Shares that can be purchased pursuant to such Warrants.

Nothing contained in this Warrant Certificate, the Warrant Indenture or elsewhere shall be construed as conferring upon the holder hereof any right or interest whatsoever as a holder of Common Shares or any other right or interest except as herein and in the Warrant Indenture expressly provided. In the event of any discrepancy between anything contained in this Warrant Certificate and the terms and conditions of the Warrant Indenture, the terms and conditions of the Warrant Indenture shall govern.

Warrants may only be transferred in compliance with the conditions of the Warrant Indenture on the register to be kept by the Warrant Agent in Vancouver, British Columbia, or such other registrar as the Corporation, with the approval of the Warrant Agent, may appoint at such other place or places, if any, as may be designated, upon surrender of this Warrant Certificate to the Warrant Agent or other registrar accompanied by a written instrument of transfer in form and execution satisfactory to the Warrant Agent or other registrar and upon compliance with the conditions prescribed in the Warrant Indenture and with such reasonable requirements as the Warrant Agent or other registrar may prescribe and upon the transfer being duly noted thereon by the Warrant Agent or other registrar. Time is of the essence hereof.

This Warrant Certificate will not be valid for any purpose until it has been countersigned by or on behalf of the Warrant Agent from time to time under the Warrant Indenture.

The parties hereto have declared that they have required that these presents and all other documents related hereto be in the English language. Les parties aux présentes déclarent qu'elles ont exigé que la présente convention, de même que tous les documents s'y rapportant, soient rédigés en anglais.

IN WITNESS WHEREOF the Corporation has caused this Warrant Certificate to be duly executed as of January 6, 2015.

Coun	tersigned and Registered by:	СОН	BAR, INC.
CST	TRUST COMPANY		
		BY:	
By:			Jon Stern, Chief Executive Officer
	Authorized Signatory	By:	
			Jeffrey Biunno, Chief Financial Officer
		А	4

SCHEDULE "B"

FORM OF TRANSFER

To: CST Trust Company

FOR VALUE RECEIVED the undersigned hereby sells, assigns and transfers to (print name and address) of the Warrants represented by this Warrants Certificate and hereby irrevocable constitutes and as its attorney with full power of substitution to transfer the said securities on the appropriate register of the Warrant appoints Agent. In the case of a warrant certificate that contains a U.S. restrictive legend, the undersigned hereby represents, warrants and certifies that (one (only) of the following must be checked): (A) the transfer is being made only to the Corporation; (B) the transfer is being made outside the United States in accordance with Regulation S under the U.S. Securities Act, and in compliance with any applicable local securities laws and regulations and the undersigned has furnished to the Corporation and the Warrant Agent an opinion of counsel of recognized standing in form and substance reasonably satisfactory to the Corporation and the Warrant Agent to such effect, or (C) the transfer is being made within the United States or to, or for the account or benefit of, a U.S. Person, in a transaction that does not require registration under the U.S. Securities Act or any applicable state securities laws and the undersigned has furnished to the Corporation and the Warrant Agent an opinion of counsel of recognized standing in form and substance reasonably satisfactory to the Corporation and the Warrant Agent to such effect.

In the case of a warrant certificate that does not contain a U.S. restrictive legend, if the proposed transfer is to, or for the account or benefit of a U.S. Person or to a person in the United States, the undersigned hereby represents, warrants and certifies that the transfer of the Warrants is being completed pursuant to an exemption from the registration requirements of the U.S. Securities Act and any applicable state securities laws, in which case the undersigned has furnished to the Corporation and the Warrant Agent an opinion of counsel of recognized standing in form and substance reasonably satisfactory to the Corporation and the Warrant Agent to such effect.

\square If transfer is to a U.S. Person, check this box.	
DATED this day of , 20 .	
SPACE FOR GUARANTEES OF SIGNATURES (BELOW)) Signature of Transferor
Guarantor's Signature/Stamp	Name of Transferor

REASON FOR TRANSFER—For US Residents only (where the individual(s) or corporation receiving the securities is a US resident). Please select only one (see instructions below).

Gift	Estate	Private Sale	Other (or no change in ownership)
Date of Event (Date of	gift, death or sale):	Value per Warrant on	the date of event:
/ /		\$	CAD OR USD

CERTAIN REQUIREMENTS RELATING TO TRANSFERS—READ CAREFULLY

The signature(s) of the transferor(s) must correspond with the name(s) as written upon the face of this certificate(s), in every particular, without alteration or enlargement, or any change whatsoever. All securityholders or a legally authorized representative must sign this form. The signature(s) on this form must be guaranteed in accordance with the transfer agent's then current guidelines and requirements at the time of transfer. Notarized or witnessed signatures are not acceptable as guaranteed signatures. As at the time of closing, you may choose one of the following methods (although subject to change in accordance with industry practice and standards):

- Canada and the USA: A Medallion Signature Guarantee obtained from a member of an acceptable Medallion Signature Guarantee Program (STAMP, SEMP, NYSE, MSP). Many commercial banks, savings banks, credit unions, and all broker dealers participate in a Medallion Signature Guarantee Program. The Guarantor must affix a stamp bearing the actual words "Medallion Guaranteed", with the correct prefix covering the face value of the certificate.
- Canada: A Signature Guarantee obtained from an authorized officer of the Royal Bank of Canada, Scotia Bank or TD Canada Trust. The Guarantor must affix a stamp bearing the actual words "Signature Guaranteed", sign and print their full name and alpha numeric signing number. Signature Guarantees are not accepted from Treasury Branches, Credit Unions or Caisse Populaires unless they are members of a Medallion Signature Guarantee Program. For corporate holders, corporate signing resolutions, including certificate of incumbency, are also required to accompany the transfer, unless there is a "Signature & Authority to Sign Guarantee" Stamp affixed to the transfer (as opposed to a "Signature Guaranteed" Stamp) obtained from an authorized officer of the Royal Bank of Canada, Scotia Bank or TD Canada Trust or a Medallion Signature Guarantee with the correct prefix covering the face value of the certificate.
- Outside North America: For holders located outside North America, present the certificates(s) and/or document(s) that require a guarantee to a local financial institution that has a corresponding Canadian or American affiliate which is a member of an acceptable Medallion Signature Guarantee Program. The corresponding affiliate will arrange for the signature to be over-guaranteed.

OR

The signature(s) of the transferor(s) must correspond with the name(s) as written upon the face of this certificate(s), in every particular, without alteration or enlargement, or any change whatsoever. The signature(s) on this form must be guaranteed by an authorized officer of Royal Bank of Canada, Scotia Bank or TD Canada Trust whose sample signature(s) are on file with the transfer agent, or by a member of an acceptable Medallion Signature Guarantee Program (STAMP, SEMP, NYSE, MSP). Notarized or witnessed signatures are not acceptable as guaranteed signatures. The Guarantor must affix a stamp bearing the actual words: "SIGNATURE GUARANTEED", "MEDALLION GUARANTEED" OR "SIGNATURE & AUTHORITY TO SIGN GUARANTEE", all in accordance with the transfer agent's then current guidelines and requirements at the time of transfer. For corporate holders, corporate signing resolutions, including certificate of incumbency, will also be required to accompany the transfer unless there is a "SIGNATURE & AUTHORITY TO SIGN GUARANTEE" Stamp affixed to the Form of Transfer obtained from an authorized officer of the Royal Bank of Canada, Scotia Bank or TD Canada Trust or a "MEDALLION GUARANTEED" Stamp affixed to the Form of Transfer, with the correct prefix covering the face value of the certificate.

REASON FOR TRANSFER—FOR US RESIDENTS ONLY

Consistent with US IRS regulations, CST is required to request cost basis information from US securityholders. Please indicate the reason for requesting the transfer as well as the date of event relating to the reason. The event date is not the day in which the transfer is finalized, but rather the date of the event which led to the transfer request (i.e. date of gift, date of death of the securityholder, or the date the private sale took place).

SCHEDULE "C"

EXERCISE FORM

TO: COHBAR, INC.

AND TO: CST TRUST COMPANY, as Warrant Agent

1600-1066 West Hastings St. Vancouver, British Columbia

V6E-VX1

The undersigned holder of the Warrants evidenced by this Warrant Certificate hereby exercises the right to acquire (A) Common Shares of Cohbar, Inc.

Aggregate Exercise Price Payable:

(# of Common Shares multiplied by USD \$2.00, subject to adjustment)

The undersigned hereby exercises the right of such holder to be issued, and hereby subscribes for, Common Shares that are issuable pursuant to the exercise of such Warrants on the terms specified in such Warrant Certificate and in the Warrant Indenture. Any capitalized term in this Warrant Certificate that is not otherwise defined herein, shall have the meaning ascribed thereto in the Warrant Indenture.

The undersigned hereby acknowledges that the undersigned is aware that the Common Shares received on exercise may be subject to restrictions on resale under applicable securities legislation and that unless the Common Shares issued upon exercise of this Warrant are registered under the United States Securities Act of 1933, as amended (the "U.S. Securities Act"), and any applicable state securities laws, such Common Shares will bear a legend restricting the transfer without registration under the U.S. Securities Act and applicable state securities laws substantially the form set forth in Section 3.3(c) of the Warrant Indenture.

Unless the Warrant Agent has received a written confirmation from the Corporation to the effect that the issuance of the Common Shares upon exercise hereof is registered under the United States Securities Act of 1933, as amended (the "U.S. Securities Act"), and any applicable state securities laws, you must complete the appended Warrant Exercise Certification.

The undersigned hereby irrevocably directs that the said Common Shares be issued, registered and delivered as follows:

Name(s) in Full and Social Address(es) Number of
Insurance Number(s) Common Shares
(if applicable)

Please print full name in which certificates representing the Common Shares are to be issued. If any Common Shares are to be issued to a person or persons other than the registered holder, the registered holder must pay to the Warrant Agent all eligible transfer taxes or other government charges, if any, and the Form of Transfer must be duly executed.

Once completed and executed, this Exercise Form mus Corporate Trust .	t be mailed or delivered to CST Trust Company, c/o General Manager,
DATED this day of , 20 .	
)))
Witness) Signature of Warrantholder (must be the same as appears on the face of the Warrant Certificate)
	Name of Registered Warrantholder
1 6	the address set out above. Certificates will be delivered or mailed as soon as the Warrant Agent.

SCHEDULE "D" WARRANT EXERCISE CERTIFICATION

(TO BE COMPLETED ONLY IF A REGISTRATION STATEMENT IS NOT EFFECTIVE)

To: COHBAR, INC.

And To: CST TRUST COMPANY

The undersigned holder of the within Warrant Certificate, pursuant to the Warrant Indenture mentioned therein, hereby exercises certain Warrants (the "Exercised Warrants") evidenced thereby and hereby subscribes for a number of Common Shares of COHBAR, INC. equal to such number of Common Shares or number or amount of other securities or property, or combination thereof, to which such exercise entitles the undersigned under the provisions of the Warrant Indenture at an aggregate price equal to the product of the Exercise Price and the number of Exercised Warrants, and on the terms specified in such Warrant Certificate and the Warrant Indenture, and in payment therefor, delivers herewith a bank draft, certified cheque or money order payable to COHBAR, INC. Capitalized terms not defined herein shall have the definitions set forth in the Warrant Indenture.

The undersigned represents that it (A) has had access to such current public information concerning COHBAR, INC. as it considered necessary in connection with its investment decision and (B) understands that the securities issuable upon exercise hereof have not been registered under the United States Securities Act of 1933, as amended (the "U.S. Securities Act").

The undersigned represents and warrants as follows: [one must be checked, check only one]

- A. The undersigned is not a U.S. Purchaser and it (1) is not in the United States; (2) is not a U.S. Person and is not exercising the Warrants for, or on behalf or benefit of, a U.S. Person or person in the United States; (3) did not execute or deliver the Warrant Exercise Form in the United States; (4) agrees not to engage in hedging transactions with regard to the Securities prior to the expiration of the one-year distribution compliance period set forth in Rule 903(b)(3) of Regulation S; (5) acknowledges that the Common Shares issuable upon exercise of the Warrants are "restricted securities" as defined in Rule 144 of the U.S. Securities Act and upon the issuance thereof, and until such time as the same is no longer required under the applicable requirements of the U.S. Securities Act or applicable U.S. state laws and regulations, the certificates representing the Common Shares will bear a restrictive legend; and (6) acknowledges that the Corporation shall refuse to register any transfer of the Securities not made in accordance with the provisions of Regulation S, pursuant to registration under the U.S. Securities Act, or pursuant to an available exemption from registration under the U.S. Securities Act; and (B) the holder has not engaged in any "directed selling efforts" (as defined in Regulation S) in the United States.
- B. The undersigned is tendering with this Warrant Exercise Certification a written opinion of counsel (which will not be sufficient unless it is in form and substance reasonably satisfactory to the Corporation) or such other evidence reasonably satisfactory to the Corporation to the effect that the Common Shares may be issued and delivered upon exercise of the Warrants pursuant to a valid exemption from the registration requirements of the U.S. Securities Act and applicable U.S. state laws and regulations.

	C.	The Warrants are Put Unit Warrants and the undersigned (a) is the original U.S. purchaser who purchased the Warrants pursuant to the Company's exercise of its rights pursuant to a certain Put Agreement between the undersigned and the Company (the "Put Agreement"), (b) is exercising the Warrants for its own account, (c) is an "accredited investor" as defined in Rule 501(a) of Regulation D under the U.S. Securities Act of 1933, as amended (the "U.S. Securities Act") at the time of exercise of these Warrants, and (d) the representations and warranties of the holder made in the original Put Agreement remain true and correct as of the date of exercise of these Warrants.
Warra substa	ant will antially	gned holder understands that the certificate representing the Corporation's Common Shares issued upon exercise of this I bear a legend restricting the transfer without registration under the U.S. Securities Act and applicable state securities laws the form set forth in Section 3.3(c) of the Warrant Indenture. "United States" and "U.S. Person" are as defined in Rule 902 of S under the U.S. Securities Act.
It is u	ndersto	ood that the Corporation and CST Trust Company may require evidence to verify the foregoing representations.
Name Nome		
Addre	ess:	Please print or type name and address (including postal code)
Adres	sse:	
		Warrants being Exercised:
Numl		
Numb	per of V	
Numb	per of V	s day of .
Numb	per of V	s day of . uaranteed by:
Numb	per of V	s day of . naranteed by: Name of registered holder (please print)
Numb	per of V	Name of registered holder (please print) Signature of or on behalf of registered holder

COHBAR, INC.

AMENDED AND RESTATED 2011 EQUITY INCENTIVE PLAN

ADOPTED BY THE BOARD OF DIRECTORS: NOVEMBER 3, 2014 APPROVED BY THE STOCKHOLDERS: DECEMBER 15, 2014 TERMINATION DATE: DECEMBER 15, 2024

1. GENERAL.

- (a) Eligible Stock Award Recipients. The persons eligible to receive Stock Awards are Employees, Directors and Consultants.
- **(b) Available Stock Awards.** The Plan provides for the grant of the following Stock Awards: (i) Incentive Stock Options; (ii) Nonstatutory Stock Options; (iii) Stock Appreciation Rights; (iv) Restricted Stock Awards; and (v) Restricted Stock Unit Awards.
- **(c) Purpose.** The Company, by means of the Plan, seeks to secure and retain the services of the group of persons eligible to receive Stock Awards as set forth in Section 1(a), to provide incentives for such persons to exert maximum efforts for the success of the Company and any Affiliate, and to provide a means by which such eligible recipients may be given an opportunity to benefit from increases in value of the Common Stock through the granting of Stock Awards.

2. ADMINISTRATION.

- (a) Administration by Board. The Board shall administer the Plan unless and until the Board delegates administration of the Plan to a Committee or Committees, as provided in Section 2(c).
 - (b) Powers of Board. The Board shall have the power, subject to, and within the limitations of, the express provisions of the Plan:
- (i) To determine from time to time (A) which of the persons eligible under the Plan shall be granted Stock Awards, (B) when and how each Stock Award shall be granted, (C) what type or combination of types of Stock Award shall be granted, (D) the provisions of each Stock Award granted (which need not be identical), including the time or times when a person shall be permitted to receive cash or Common Stock pursuant to a Stock Award, (E) the number of shares of Common Stock with respect to which a Stock Award shall be granted to each such person, and (F) the Fair Market Value applicable to a Stock Award.
- (ii) To construe and interpret the Plan and Stock Awards granted under it, and to establish, amend and revoke rules and regulations for administration of the Plan. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan or in any Stock Award Agreement, in a manner and to the extent it shall deem necessary or expedient to make the Plan or Stock Award fully effective.

- (iii) To settle all controversies regarding the Plan and Stock Awards granted under it.
- (iv) To accelerate the time at which a Stock Award may first be exercised or the time during which a Stock Award or any part thereof will vest in accordance with the Plan, notwithstanding the provisions in the Stock Award stating the time at which it may first be exercised or the time during which it will vest.
- (v) To suspend or terminate the Plan at any time. Suspension or termination of the Plan shall not impair rights and obligations under any Stock Award granted while the Plan is in effect except with the written consent of the affected Participant.
- (vi) To amend the Plan in any respect the Board deems necessary or advisable, including, without limitation, amendments relating to Incentive Stock Options and certain nonqualified deferred compensation under Section 409A of the Code and/or to bring the Plan or Stock Awards granted under the Plan into compliance therewith, subject to the limitations, if any, of applicable law. The Board may by resolution amend the Plan and any option granted hereunder without stockholder approval, subject to applicable laws and the policies of the Exchange. However, the Board will not be permitted, in the absence of stockholder and Exchange approval, to amend the terms of an option held by an Insider of the Company (other than to extend the expiry date of an option, subject to such date being extended by virtue of Section 5 below). Except as provided above, rights under any Stock Award granted before amendment of the Plan shall not be impaired by any amendment of the Plan unless (1) the Company requests the consent of the affected Participant, and (2) such Participant consents in writing.
- (vii) To submit any amendment to the Plan for stockholder approval, including, but not limited to, amendments to the Plan intended to satisfy the requirements of Section 422 of the Code regarding Incentive Stock Options.
- (viii) To approve forms of Stock Award Agreements for use under the Plan and to amend the terms of any one or more Stock Awards, including, but not limited to, amendments to provide terms more favorable to the Participant than previously provided in the Stock Award Agreement, subject to any specified limits in the Plan that are not subject to Board discretion; provided, however, that except with respect to amendments that disqualify or impair the status of an Incentive Stock Option, a Participant's rights under any Stock Award shall not be impaired by any such amendment unless (A) the Company requests the consent of the affected Participant, and (B) such Participant consents in writing. Notwithstanding the foregoing, subject to the limitations of applicable law, if any, the Board may amend the terms of any one or more Stock Awards without the affected Participant's consent if necessary to maintain the qualified status of the Stock Award as an Incentive Stock Option or to bring the Stock Award into compliance with Section 409A of the Code.

- (ix) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company and that are not in conflict with the provisions of the Plan or Stock Awards.
- (x) To adopt such procedures and sub-plans as are necessary or appropriate to permit participation in the Plan by Employees, Directors or Consultants who are foreign nationals or employed outside the United States.
- (xi) Subject to the policies of the Exchange and, if required by such policies, approval of the stockholders of the Company, to effect, at any time and from time to time, with the consent of any adversely affected Participant, (A) the reduction of the exercise price (or strike price) of any outstanding Option or SAR under the Plan, (B) the cancellation of any outstanding Option or SAR under the Plan and the grant in substitution therefore of (1) a new Option or SAR under the Plan or another equity plan of the Company covering the same or a different number of shares of Common Stock, (2) a Restricted Stock Award, (3) a Restricted Stock Unit Award, (4) cash and/or (5) other valuable consideration (as determined by the Board, in its sole discretion), or (C) any other action that is treated as a repricing under generally accepted accounting principles; *provided*, *however*, that no such reduction or cancellation may be effected if it is determined, in the Company's sole discretion, that such reduction or cancellation would result in any such outstanding Option becoming subject to the requirements of Section 409A of the Code; and *provided further that*, any reduction in the exercise price (or strike price) of an Option or SAR held by an Optionholder or SAR holder who is an Insider of the Company at the time of the proposed reduction will require disinterested shareholder approval.
- (c) Delegation to Committee. The Board may delegate some or all of the administration of the Plan to a Committee or Committees. If administration of the Plan is delegated to a Committee, the Committee shall have, in connection with the administration of the Plan, the powers theretofore possessed by the Board that have been delegated to the Committee, including the power to delegate to a subcommittee of the Committee any of the administrative powers the Committee is authorized to exercise (and references in this Plan to the Board shall thereafter be to the Committee or subcommittee), subject, however, to such resolutions, not inconsistent with the provisions of the Plan, as may be adopted from time to time by the Board. The Committee may, at any time, abolish the subcommittee and/or revest in the Committee any powers delegated to the subcommittee. The Board may retain the authority to concurrently administer the Plan with the Committee and may, at any time, revest in the Board some or all of the powers previously delegated.
- (d) Effect of Board's Decision. All determinations, interpretations and constructions made by the Board in good faith shall not be subject to review by any person and shall be final, binding and conclusive on all persons.

3. SHARES SUBJECT TO THE PLAN.

(a) Share Reserve. The Plan is a "fixed" stock option plan reserving for issuance up to a maximum of 20% of the Company's issued shares of Common Stock as of the date of the implementation of the Plan. Subject to Section 9(a) relating to Capitalization Adjustments, the

aggregate number of shares of Common Stock that may be issued pursuant to Stock Awards from and after the Effective Date shall not exceed six million four hundred fifty-three thousand sixty-nine (6,453,069) (the "*Share Reserve*"). Further, if a Stock Award or any portion thereof (i) expires or otherwise terminates without all of the shares covered by such Stock Award having been issued or (ii) is settled in cash (*i.e.*, the Participant receives cash rather than stock), such expiration, termination or settlement shall not reduce (or otherwise offset) the number of shares of Common Stock that may be available for issuance under the Plan. For clarity, the limitation in this Section 3(a) is a limitation in the number of shares of Common Stock that may be issued pursuant to the Plan. Accordingly, this Section 3(a) does not limit the granting of Stock Awards except as provided in Section 7(a).

- **(b)** Reversion of Shares to the Share Reserve. If any shares of Common Stock issued pursuant to a Stock Award are forfeited back to or repurchased by the Company because of the failure to meet a contingency or condition required to vest such shares in the Participant, then the shares that are forfeited or repurchased shall revert to and again become available for issuance under the Plan. Also, any shares reacquired by the Company pursuant to Section 8(g) or as consideration for the exercise of an Option shall again become available for issuance under the Plan. Notwithstanding the provisions of this Section 3(b), any such shares shall not be subsequently issued pursuant to the exercise of Incentive Stock Options.
- **(c) Source of Shares.** The stock issuable under the Plan shall be shares of authorized but unissued or reacquired Common Stock, including shares repurchased by the Company on the open market or otherwise.

4. ELIGIBILITY.

- (a) Eligibility for Specific Stock Awards. Incentive Stock Options may be granted only to employees of the Company or a "parent corporation" or "subsidiary corporation" thereof (as such terms are defined in Sections 424(e) and 424(f) of the Code). Stock Awards other than Incentive Stock Options may be granted to Employees, Directors and Consultants; *provided, however*, that Nonstatutory Stock Options and SARs may not be granted to Employees, Directors and Consultants who are providing Continuous Service only to any "parent" of the Company, as such term is defined in Rule 405, unless the stock underlying such Stock Awards is treated as "service recipient stock" under Section 409A of the Code because the Stock Awards are granted pursuant to a corporate transaction (such as a spin off transaction) or unless such Stock Awards comply with the distribution requirements of Section 409A of the Code.
- **(b) Ten Percent Stockholders.** A Ten Percent Stockholder shall not be granted an Incentive Stock Option unless the exercise price of such Option is at least one hundred ten percent (110%) of the Fair Market Value on the date of grant and the Option is not exercisable after the expiration of five (5) years from the date of grant.
- **(c) Consultants.** A Consultant shall not be eligible for the grant of a Stock Award if, at the time of grant, either the offer or sale of the Company's securities to such Consultant is not exempt under Rule 701 because of the nature of the services that the Consultant is providing to the Company, because the Consultant is not a natural person, or because of any other provision of Rule 701, unless the Company determines that such grant need not comply with the requirements of Rule 701 and will satisfy another exemption under the Securities Act as well as comply with the securities laws of all other relevant jurisdictions.

5. PROVISIONS RELATING TO OPTIONS AND STOCK APPRECIATION RIGHTS.

Each Option or SAR shall be in such form and shall contain such terms and conditions as the Board shall deem appropriate. All Options shall be separately designated Incentive Stock Options or Nonstatutory Stock Options at the time of grant, and, if certificates are issued, a separate certificate or certificates shall be issued for shares of Common Stock purchased on exercise of each type of Option. If an Option is not specifically designated as an Incentive Stock Option, then the Option shall be a Nonstatutory Stock Option. For greater certainty, no SAR will be granted under the Plan so long as the shares of Common Stock of the Company are listed on the TSX-V and the Company is classified as a "Tier 2 Issuer" by the TSX-V. The provisions of separate Options or SARs need not be identical; *provided*, *however*, that each Option Agreement or Stock Appreciation Right Agreement shall conform to (through incorporation of provisions hereof by reference in the applicable Stock Award Agreement or otherwise) the substance of each of the following provisions:

- (a) Term. Subject to the provisions Section 4(b) regarding Ten Percent Stockholders, no Option or SAR shall be exercisable after the expiration of ten (10) years from the date of its grant or such shorter period specified in the Stock Award Agreement.
- (i) Should the expiry date of an option fall within a Black Out Period (as hereinafter defined) or within nine business days following the expiration of a Black Out Period, such expiry date of the option shall be automatically extended without any further act or formality to that date which is the tenth business day after the end of the Black Out Period, such tenth business day to be considered the expiry date for such option for all purposes under the Plan. The ten business day period referred to in this paragraph may not be extended by the Board. "Black Out Period" means the period during which the relevant Participant is prohibited from exercising an option due to trading restrictions imposed by the Company pursuant to any policy of the Company respecting restrictions on trading that is in effect at that time.
- **(b) Exercise Price.** Subject to the provisions of Section 4(b) regarding Incentive Stock Options granted to Ten Percent Stockholders, the exercise price (or strike price) of each Option or SAR shall be not less than one hundred percent (100%) of the Fair Market Value of the Common Stock subject to the Option or SAR on the date the Option or SAR is granted. Notwithstanding the foregoing, an Option or SAR may be granted with an exercise price (or strike price) lower than one hundred percent (100%) of the Fair Market Value of the Common Stock subject to the Option or SAR if such Option or SAR is granted pursuant to an assumption of or substitution for another option or stock appreciation right pursuant to a Corporate Transaction and in a manner consistent with the provisions of Sections 409A and 424(a) of the Code (whether or not such stock awards are Incentive Stock Options). Each SAR will be denominated in shares of Common Stock equivalents. For so long as such shares are listed on the TSX-V, no option shall be granted with an exercise price at a discount to the market price. The "market price" shall have the meaning defined in the rules and policies of the TSX-V. Notwithstanding the foregoing, options may be granted with a per Share exercise price other than

as required above in accordance with, and pursuant to, a transaction described in Section 424 of the Code, subject to the rules of the Exchange. Once the exercise price has been determined by the Board, accepted by the Exchange and the option has been granted, the exercise price of an option may be reduced upon receipt of Board approval, subject to the policies of the Exchange and, if required by such policies, approval of the stockholders of the Company.

- (c) Number of Optioned Shares. The aggregate number of shares of Common Stock that may be issued pursuant to the exercise of options awarded under the Plan and all other security based compensation arrangements of the Company six million four hundred fifty-three thousand sixty-nine (6,453,069) shares, being 20% of the shares of Common Stock issued as of the date of the implementation of the Plan, subject to the following additional limitations for so long as the shares of Common Stock are listed on the TSX-V:
- (i) the aggregate number of options granted to any one person under the Plan within any 12-month period must not exceed five percent (5%) of the then outstanding shares of Common Stock (on a non-diluted basis), unless the Company has obtained disinterested stockholder approval;
- (ii) in the aggregate, no more than ten percent (10%) of the issued and outstanding shares of Common Stock (on a non-diluted basis) may be reserved at any time for Insiders under the Plan, unless the Company has obtained disinterested stockholder approval;
- (iii) the number of options granted to Insiders within any 12-month period, cannot exceed ten percent (10%) of the issued and outstanding shares of Common Stock, unless the Company has obtained disinterested stockholder approval;
- (iv) the aggregate number of Options granted to any one Consultant within a 12-month period shall not exceed two percent (2%) of the issued shares of Common Stock of the Company;
- (v) the aggregate number of Options granted to all Persons retained to provide Investor Relations Activities within a 12-month period shall not exceed two percent (2%) of the issued shares of Common Stock of the Company. For the avoidance of doubt, options may not be granted to Consultants performing investor relations activities; and
- (vi) the number of shares of Common Stock subject to an option granted to any one Participant shall be determined by the Board, but no one Participant shall be granted an option to purchase a number of shares of Common Stock that exceeds the maximum number permitted by the Exchange.
- **(d) Purchase Price for Options.** The purchase price of Common Stock acquired pursuant to the exercise of an Option shall be paid as determined by the Board in its sole discretion, by any combination of the methods of payment set forth below, *only to the extent* permitted by applicable law and by permission of the TSX-V so long as the shares of Common Stock of the Company are listed on the TSX-V. The Board shall have the authority to grant Options that do not permit all of the following methods of payment (or otherwise restrict the ability to use certain methods) and to grant Options that require the consent of the Company to utilize a particular method of payment. The permitted methods of payment are as follows:

- (i) by cash, check, bank draft or money order payable to the Company;
- (ii) pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board that, prior to the issuance of the stock subject to the Option, results in either the receipt of cash (or check) by the Company or the receipt of irrevocable instructions to pay the aggregate exercise price to the Company from the sales proceeds;
 - (iii) by delivery to the Company (either by actual delivery or attestation) of shares of Common Stock;
- (iv) if the Option is a Nonstatutory Stock Option, by a "net exercise" arrangement pursuant to which the Company will reduce the number of shares of Common Stock issued upon exercise by the largest whole number of shares with a Fair Market Value that does not exceed the aggregate exercise price; provided, however, that the Company shall accept a cash or other payment from the Participant to the extent of any remaining balance of the aggregate exercise price not satisfied by such reduction in the number of whole shares to be issued; provided further, that shares of Common Stock will no longer be outstanding under an Option and will not be exercisable thereafter to the extent that (A) shares are used to pay the exercise price pursuant to the "net exercise," (B) shares are delivered to the Participant as a result of such exercise, and (C) shares are withheld to satisfy tax withholding obligations;
- (v) according to a deferred payment or similar arrangement with the Optionholder; provided, however, that interest shall compound at least annually and shall be charged at the minimum rate of interest necessary to avoid (A) the imputation of interest income to the Company and compensation income to the Optionholder under any applicable provisions of the Code, and (B) the classification of the Option as a liability for financial accounting purposes; or
 - (vi) in any other form of legal consideration that may be acceptable to the Board.
- (e) Exercise and Payment of a SAR. To exercise any outstanding SAR, the Participant must provide written notice of exercise to the Company in compliance with the provisions of the Stock Appreciation Right Agreement evidencing such SAR. The appreciation distribution payable on the exercise of a SAR will be not greater than an amount equal to the excess of (A) the aggregate Fair Market Value (on the date of the exercise of the SAR) of a number of shares of Common Stock equal to the number of Common Stock equivalents in which the Participant is vested under such SAR, and with respect to which the Participant is exercising the SAR on such date, over (B) the strike price that will be determined by the Board at the time of grant of the SAR. The appreciation distribution in respect to a SAR may be paid in Common Stock, in cash, in any combination of the two or in any other form of consideration, as determined by the Board and contained in the Stock Appreciation Right Agreement evidencing such SAR.

- **(f) Transferability and Assignability of Options and SARs.** All benefits, rights and options accruing to any Participant or Optionholder in accordance with the terms and conditions with the Plan shall not be transferable or assignable except as provided herein or as otherwise determined by the Board under terms that are not prohibited by applicable tax and securities laws, and <u>then only to the extent</u>, if any, permitted by the TSX-V so long as the shares of Common Stock of the Company are listed on the TSX-V. The Board may, in its sole discretion, impose additional limitations on the transferability of Options and SARs in the applicable Stock Award Agreement.
- (i) Restrictions on Transfer. An Option or SAR shall not be transferable except by will or by the laws of descent and distribution and shall be exercisable during the lifetime of the Participant only by the Participant; *provided, however*, that the Board may, in its sole discretion, permit transfer of the Option or SAR and in a manner that is not prohibited by applicable tax and securities laws (including but not limited to Rule 701) upon the Participant's request. Except as explicitly provided herein, neither an Option nor a SAR may be transferred for consideration.
- (ii) **Domestic Relations Orders.** Upon receiving written permission from the Board or its duly authorized designee, an Option or SAR may be transferred pursuant to a domestic relations order; *provided, however*, that if an Option is an Incentive Stock Option, such Option may be deemed to be a Nonstatutory Stock Option as a result of such transfer.
- (iii) Beneficiary Designation. Upon receiving written permission from the Board or its duly authorized designee, the Participant may, by delivering written notice to the Company, in a form provided by or otherwise satisfactory to the Company, designate a third party who, in the event of the death of the Participant, shall thereafter be the beneficiary of the Option or SAR with the right to exercise the Option or SAR and receive the Common Stock or other consideration resulting from such exercise. In the absence of such a designation, the executor or administrator of the Participant's estate shall be entitled to exercise the Option or SAR and receive the Common Stock or other consideration resulting from such exercise.
- **(g) Vesting Generally.** The total number of shares of Common Stock subject to an Option or SAR may vest and therefore become exercisable in periodic installments that may or may not be equal. The Option or SAR may be subject to such other terms and conditions on the time or times when it may or may not be exercised (which may be based on the satisfaction of performance goals or other criteria) as the Board may deem appropriate. The vesting provisions of individual Options or SARs may vary, provided that all Options issued to persons retained to provide Investor Relations Activities must vest in stages over a period of not less than 12 months with no more than 1/4 vesting in any three month period. The provisions of this Section 5(g) are subject to any Option or SAR provisions governing the minimum number of shares of Common Stock as to which an Option or SAR may be exercised.
- **(h) Termination of Continuous Service.** Except as otherwise provided in the applicable Stock Award Agreement or other agreement between the Participant and the Company, in the event that a Participant's Continuous Service terminates (other than for Cause or upon the Participant's death or Disability), the Participant may exercise his or her Option or

SAR (to the extent that the Participant was entitled to exercise such Stock Award as of the date of termination of Continuous Service) but only within such period of time ending on the earlier of (i) the date three (3) months following the termination of the Participant's Continuous Service (or such longer or shorter period specified in the applicable Stock Award Agreement, which period shall not be less than thirty (30) days if necessary to comply with applicable laws unless such termination is for Cause) or (ii) the expiration of the term of the Option or SAR as set forth in the Stock Award Agreement. If, after termination of Continuous Service, the Participant does not exercise his or her Option or SAR within the time specified herein or in the Stock Award Agreement (as applicable), the Option or SAR shall terminate.

- (i) Extension of Termination Date. Except as otherwise provided in the applicable Stock Award Agreement or other agreement between the Participant and the Company, if the exercise of the Option or SAR following the termination of the Participant's Continuous Service (other than for Cause or upon the Participant's death or Disability) would be prohibited at any time solely because the issuance of shares of Common Stock would violate the registration requirements under the Securities Act, then the Option or SAR shall terminate on the earlier of (i) the expiration of a total period of three (3) months (that need not be consecutive) after the termination of the Participant's Continuous Service during which the exercise of the Option or SAR would not be in violation of such registration requirements, (ii) the date 12 months after the termination of such Participant's Continuous Service, or (iii) the expiration of the term of the Option or SAR as set forth in the applicable Stock Award Agreement. In addition, unless otherwise provided in a Participant's Stock Award Agreement, if the immediate sale of any Common Stock received upon exercise of an Option or SAR following the termination of the Participant's Continuous Service (other than for Cause) would violate the Company's insider trading policy, then the Option or SAR shall terminate on the earlier of (i) the expiration of a period equal to the applicable post-termination exercise period after the termination of the Participant's Continuous Service during which the sale of Common Stock received upon exercise of the Option or SAR would not be in violation of the Company's insider trading policy, or (ii) the expiration of the term of the Option or SAR as set forth in the applicable Stock Award Agreement.
- (j) Disability of Participant. Except as otherwise provided in the applicable Stock Award Agreement or other agreement between the Participant and the Company, in the event that a Participant's Continuous Service terminates as a result of the Participant's Disability, the Participant may exercise his or her Option or SAR (to the extent that the Participant was entitled to exercise such Option or SAR as of the date of termination of Continuous Service), but only within such period of time ending on the earlier of (i) the date twelve (12) months following such termination of Continuous Service (or such shorter period specified in the Stock Award Agreement, which period shall not be less than six (6) months if necessary to comply with applicable laws), or (ii) the expiration of the term of the Option or SAR as set forth in the Stock Award Agreement. If, after termination of Continuous Service, the Participant does not exercise his or her Option or SAR within the time specified herein or in the Stock Award Agreement (as applicable), the Option or SAR shall terminate.

- (k) Death of Participant. Except as otherwise provided in the applicable Stock Award Agreement or other agreement between the Participant and the Company, in the event that (i) a Participant's Continuous Service terminates as a result of the Participant's death, or (ii) the Participant dies within the period (if any) specified in the Stock Award Agreement after the termination of the Participant's Continuous Service for a reason other than death, then the Option or SAR may be exercised (to the extent the Participant was entitled to exercise such Option or SAR as of the date of death) by the Participant's estate or administrator or by a person who acquired the right to exercise the Option or SAR by bequest or inheritance or by a person designated to exercise the Option or SAR upon the Participant's death, but only within the period ending on the earlier of (i) the date twelve (12) months following the date of death (or such shorter period specified in the Stock Award Agreement, which period shall not be less than six (6) months if necessary to comply with applicable laws), or (ii) the expiration of the term of such Option or SAR as set forth in the Stock Award Agreement. If, after the Participant's death, the Option or SAR is not exercised within the time specified herein or in the Stock Award Agreement (as applicable), the Option or SAR shall terminate.
- (I) Termination for Cause. Except as explicitly provided otherwise in a Participant's Stock Award Agreement, if a Participant's Continuous Service is terminated for Cause, the Option or SAR shall terminate upon the termination date of such Participant's Continuous Service, and the Participant shall be prohibited from exercising his or her Option or SAR from and after the time of such termination of Continuous Service.
- (m) Non-Exempt Employees. No Option or SAR granted to an Employee who is a non-exempt employee for purposes of the Fair Labor Standards Act of 1938, as amended, shall be first exercisable for any shares of Common Stock until at least six months following the date of grant of the Option or SAR. Notwithstanding the foregoing, consistent with the provisions of the Worker Economic Opportunity Act, (i) in the event of the Participant's death or Disability, (ii) upon a Corporate Transaction in which such Option or SAR is not assumed, continued or substituted, (iii) upon a Change in Control in which the vesting of such Options or SARs accelerates, or (iv) upon the Participant's retirement (as such term is defined for purposes of the Fair Labor Standards Act of 1938) the vested portion of any Options and SARs may be exercised earlier than six months following the date of grant. The foregoing provision is intended to operate so that any income derived by a non-exempt employee in connection with the exercise or vesting of an Option or SAR will be exempt from his or her regular rate of pay.
- (n) Early Exercise of Options. An Option may, but need not, include a provision whereby the Optionholder may elect at any time before the Optionholder's Continuous Service terminates to exercise the Option as to any part or all of the shares of Common Stock subject to the Option prior to the full vesting of the Option. Subject to the "Repurchase Limitation" in Section 8(l), any unvested shares of Common Stock so purchased may be subject to a repurchase right in favor of the Company or to any other restriction the Board determines to be appropriate. Provided that the "Repurchase Limitation" in Section 8(l) is not violated, the Company shall not be required to exercise its repurchase right until at least six (6) months (or such longer or shorter period of time required to avoid classification of the Option as a liability for financial accounting purposes) have elapsed following exercise of the Option unless the Board otherwise specifically provides in the Option Agreement.

- (o) **Right of Repurchase.** Subject to the "Repurchase Limitation" in Section 8(1), an Option or SAR may include a provision whereby the Company may elect to repurchase all or any part of the vested shares of Common Stock acquired by the Participant pursuant to the exercise of the Option or SAR.
- (p) Right of First Refusal. An Option or SAR may include a provision whereby the Company may elect to exercise a right of first refusal following receipt of notice from the Participant of the intent to transfer all or any part of the shares of Common Stock received upon the exercise of the Option or SAR. Such right of first refusal shall be subject to the "Repurchase Limitation" in Section 8(l). Except as expressly provided in this Section 5(p) or in the Stock Award Agreement, such right of first refusal shall otherwise comply with any applicable provisions of the bylaws of the Company.

6. PROVISIONS OF RESTRICTED STOCK AWARDS AND RESTRICTED STOCK UNITS.

- (a) Restricted Stock Awards. For greater certainty, no Restricted Stock Awards will be granted under the Plan so long as the shares of Common Stock of the Company are listed on the TSX-V and the Company is classified as a "Tier 2 Issuer" by the TSX-V. Each Restricted Stock Award Agreement shall be in such form and shall contain such terms and conditions as the Board shall deem appropriate. To the extent consistent with the Company's bylaws, at the Board's election, shares of Common Stock may be (A) held in book entry form subject to the Company's instructions until any restrictions relating to the Restricted Stock Award lapse, or (B) evidenced by a certificate, which certificate shall be held in such form and manner as determined by the Board. The terms and conditions of Restricted Stock Award Agreements may change from time to time, and the terms and conditions of separate Restricted Stock Award Agreements need not be identical; *provided, however*, that each Restricted Stock Award Agreement shall conform to (through incorporation of the provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions:
- (i) Consideration. A Restricted Stock Award may be awarded in consideration for (A) cash or cash equivalents, (B) past or future services actually or to be rendered to the Company or an Affiliate, or (C) any other form of legal consideration that may be acceptable to the Board in its sole discretion and permissible under applicable law.
- (ii) Vesting. Subject to the "Repurchase Limitation" in Section 8(l), shares of Common Stock awarded under the Restricted Stock Award Agreement may be subject to forfeiture to the Company in accordance with a vesting schedule to be determined by the Board.
- (iii) Termination of Participant's Continuous Service. If a Participant's Continuous Service terminates, the Company may receive through a forfeiture condition or a repurchase right, any or all of the shares of Common Stock held by the Participant that have not vested as of the date of termination of Continuous Service under the terms of the Restricted Stock Award Agreement.

- (iv) Transferability. Rights to acquire shares of Common Stock under the Restricted Stock Award Agreement shall be transferable by the Participant only upon such terms and conditions as are set forth in the Restricted Stock Award Agreement, as the Board shall determine in its sole discretion, so long as Common Stock awarded under the Restricted Stock Award Agreement remains subject to the terms of the Restricted Stock Award Agreement.
- (v) **Dividends.** A Restricted Stock Award Agreement may provide that any dividends paid on Restricted Stock will be subject to the same vesting and forfeiture restrictions as apply to the shares subject to the Restricted Stock Award to which they relate.
- **(b) Restricted Stock Unit Awards.** For greater certainty, no Restricted Stock Unit Awards will be granted under the Plan so long as the shares of Common Stock of the Company are listed on the TSX-V and the Company is classified as a "Tier 2 Issuer" by the TSX-V. Each Restricted Stock Unit Award Agreement shall be in such form and shall contain such terms and conditions as the Board shall deem appropriate. The terms and conditions of Restricted Stock Unit Award Agreements may change from time to time, and the terms and conditions of separate Restricted Stock Unit Award Agreements need not be identical, *provided, however*, that each Restricted Stock Unit Award Agreement shall conform to (through incorporation of the provisions hereof by reference in the Agreement or otherwise) the substance of each of the following provisions:
- (i) Consideration. At the time of grant of a Restricted Stock Unit Award, the Board will determine the consideration, if any, to be paid by the Participant upon delivery of each share of Common Stock subject to the Restricted Stock Unit Award. The consideration to be paid (if any) by the Participant for each share of Common Stock subject to a Restricted Stock Unit Award may be paid in any form of legal consideration that may be acceptable to the Board in its sole discretion and permissible under applicable law.
- (ii) Vesting. At the time of the grant of a Restricted Stock Unit Award, the Board may impose such restrictions or conditions to the vesting of the Restricted Stock Unit Award as it, in its sole discretion, deems appropriate.
- (iii) Payment. A Restricted Stock Unit Award may be settled by the delivery of shares of Common Stock, their cash equivalent, any combination thereof or in any other form of consideration, as determined by the Board and contained in the Restricted Stock Unit Award Agreement.
- (iv) Additional Restrictions. At the time of the grant of a Restricted Stock Unit Award, the Board, as it deems appropriate, may impose such restrictions or conditions that delay the delivery of the shares of Common Stock (or their cash equivalent) subject to a Restricted Stock Unit Award to a time after the vesting of such Restricted Stock Unit Award.
- (v) Dividend Equivalents. Dividend equivalents may be credited in respect of shares of Common Stock covered by a Restricted Stock Unit Award, as determined by the Board and contained in the Restricted Stock Unit Award Agreement. At the sole discretion of the Board, such dividend equivalents may be converted into additional shares of Common Stock

covered by the Restricted Stock Unit Award in such manner as determined by the Board. Any additional shares covered by the Restricted Stock Unit Award credited by reason of such dividend equivalents will be subject to all the terms and conditions of the underlying Restricted Stock Unit Award Agreement to which they relate.

- (vi) Termination of Participant's Continuous Service. Except as otherwise provided in the applicable Restricted Stock Unit Award Agreement, such portion of the Restricted Stock Unit Award that has not vested will be forfeited upon the Participant's termination of Continuous Service.
- (vii) Compliance with Section 409A of the Code. Notwithstanding anything to the contrary set forth herein, any Restricted Stock Unit Award granted under the Plan that is not exempt from the requirements of Section 409A of the Code shall contain such provisions so that such Restricted Stock Unit Award will comply with the requirements of Section 409A of the Code. Such restrictions, if any, shall be determined by the Board and contained in the Restricted Stock Unit Award Agreement evidencing such Restricted Stock Unit Award. For example, such restrictions may include, without limitation, a requirement that any Common Stock that is to be issued in a year following the year in which the Restricted Stock Unit Award vests must be issued in accordance with a fixed pre-determined schedule.

7. COVENANTS OF THE COMPANY.

- (a) Availability of Shares. During the terms of the Stock Awards, the Company shall keep available at all times the number of shares of Common Stock reasonably required to satisfy such Stock Awards.
- **(b) Securities Law Compliance.** The Company shall seek to obtain from each regulatory commission or agency having jurisdiction over the Plan such authority as may be required to grant Stock Awards and to issue and sell shares of Common Stock upon exercise of the Stock Awards; provided, however, that this undertaking shall not require the Company to register under the Securities Act the Plan, any Stock Award or any Common Stock issued or issuable pursuant to any such Stock Award. If, after reasonable efforts, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary for the lawful issuance and sale of Common Stock under the Plan, the Company shall be relieved from any liability for failure to issue and sell Common Stock upon exercise of such Stock Awards unless and until such authority is obtained. A Participant shall not be eligible for the grant of a Stock Award or the subsequent issuance of Common Stock pursuant to the Stock Award if such grant or issuance would be in violation of any applicable securities law.
- **(c)** Necessary Approvals. The ability of a Participant to exercise options and the obligations of the Company to issue and deliver shares of Common Stock in accordance with the Plan is subject to any approvals which may be required from stockholders of the Company and any regulatory authority or stock exchange having jurisdiction over the securities of the Company. If any shares of Common Stock cannot be issued to any Participant for whatever reason, the obligation of the Company to issue such shares of Common Stock shall terminate and any option exercise price paid to the Company will be returned to the Participant.

(d) Time of Granting Options. The date of grant of an option shall, for all purposes, be the date on which the Board makes the determination granting such option, or such later date as is determined by the Board. Notice of the determination shall be given to each Participant to whom an option is so granted within a reasonable time after the date of such grant.

(e) Conditions Upon Issuance of Shares.

- (i) Legal Compliance. Shares of Common Stock shall not be issued pursuant to the exercise of an option unless the exercise of such option and the issuance and delivery of such Shares comply with Applicable Laws and shall be further subject to the approval of counsel for the Company with respect to such compliance.
- (ii) Investment Representations. As a condition to the exercise of an option, the Board may require a person exercising such option to represent and warrant at the time of any such exercise that the shares of Common Stock are being purchased only for investment and without any present intention to sell or distribute such shares if, in the opinion of counsel for the Company, such a representation is required.
- **(f) Inability to Obtain Authority.** The Company's inability to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any shares of Common Stock hereunder, shall relieve the Company of any liability in respect of the failure to issue or sell such shares as to which such requisite authority shall not have been obtained.
- (g) No Obligation to Notify or Minimize Taxes. The Company shall have no duty or obligation to any Participant to advise such holder as to the time or manner of exercising such Stock Award. Furthermore, the Company shall have no duty or obligation to warn or otherwise advise such holder of a pending termination or expiration of a Stock Award or a possible period in which the Stock Award may not be exercised. The Company has no duty or obligation to minimize the tax consequences of a Stock Award to the holder of such Stock Award.

8. MISCELLANEOUS.

- (a) Use of Proceeds from Sales of Common Stock. Proceeds from the sale of shares of Common Stock pursuant to Stock Awards shall constitute general funds of the Company.
- **(b)** Corporate Action Constituting Grant of Stock Awards. Corporate action constituting a grant by the Company of a Stock Award to any Participant shall be deemed completed as of the date of such corporate action, unless otherwise determined by the Board, regardless of when the instrument, certificate, or letter evidencing the Stock Award is communicated to, or actually received or accepted by, the Participant.

- (c) Stockholder Rights. No Participant shall be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares of Common Stock subject to such Stock Award unless and until (i) such Participant has satisfied all requirements for the exercise of the Stock Award, or the issuance of shares thereunder, pursuant to its terms, and (ii) the issuance of the Common Stock pursuant to the Stock Award has been entered into the books and records of the Company.
- (d) No Employment or Other Service Rights. Nothing in the Plan, any Stock Award Agreement or any other instrument executed thereunder or in connection with any Stock Award granted pursuant thereto shall confer upon any Participant any right to continue to serve the Company or an Affiliate in the capacity in effect at the time the Stock Award was granted or shall affect the right of the Company or an Affiliate to terminate (i) the employment of an Employee with or without notice and with or without cause, (ii) the service of a Consultant pursuant to the terms of such Consultant's agreement with the Company or an Affiliate, or (iii) the service of a Director pursuant to the bylaws of the Company or an Affiliate, and any applicable provisions of the corporate law of the state in which the Company or the Affiliate is incorporated, as the case may be.
- **(e) Incentive Stock Option \$100,000 Limitation.** To the extent that the aggregate Fair Market Value (determined at the time of grant) of Common Stock with respect to which Incentive Stock Options are exercisable for the first time by any Optionholder during any calendar year (under all plans of the Company and any Affiliates) exceeds one hundred thousand dollars (\$100,000), the Options or portions thereof that exceed such limit (according to the order in which they were granted) shall be treated as Nonstatutory Stock Options, notwithstanding any contrary provision of the applicable Option Agreement.
- (f) Investment Assurances. The Company may require a Participant, as a condition of exercising or acquiring Common Stock under any Stock Award, (i) to give written assurances satisfactory to the Company as to the Participant's knowledge and experience in financial and business matters and/or to employ a purchaser representative reasonably satisfactory to the Company who is knowledgeable and experienced in financial and business matters and that he or she is capable of evaluating, alone or together with the purchaser representative, the merits and risks of exercising the Stock Award, and (ii) to give written assurances satisfactory to the Company stating that the Participant is acquiring Common Stock subject to the Stock Award for the Participant's own account and not with any present intention of selling or otherwise distributing the Common Stock. The foregoing requirements, and any assurances given pursuant to such requirements, shall be inoperative if (x) the issuance of the shares upon the exercise or acquisition of Common Stock under the Stock Award has been registered under a then currently effective registration statement under the Securities Act, or (y) as to any particular requirement, a determination is made by counsel for the Company that such requirement need not be met in the circumstances under the then applicable securities laws. The Company may, upon advice of counsel to the Company, place legends on stock certificates issued under the Plan as such counsel deems necessary or appropriate in order to comply with applicable securities laws, including, but not limited to, legends restricting the transfer of the Common Stock.

- (g) Withholding Obligations. Unless prohibited by the terms of a Stock Award Agreement, the Company may, in its sole discretion, satisfy any federal, state or local tax withholding obligation relating to a Stock Award by any of the following means (in addition to the Company's right to withhold from any compensation paid to the Participant by the Company) or by a combination of such means: (i) causing the Participant to tender a cash payment; (ii) withholding shares of Common Stock from the shares of Common Stock issued or otherwise issuable to the Participant in connection with the Stock Award; provided, however, that no shares of Common Stock are withheld with a value exceeding the minimum amount of tax required to be withheld by law (or such lesser amount as may be necessary to avoid classification of the Stock Award as a liability for financial accounting purposes); (iii) withholding payment from any amounts otherwise payable to the Participant; (iv) withholding cash from a Stock Award settled in cash; or (v) by such other method as may be set forth in the Stock Award Agreement.
- (h) Electronic Delivery. Any reference herein to a "written" agreement or document shall include any agreement or document delivered electronically or posted on the Company's intranet.
- (i) Deferrals. To the extent permitted by applicable law, the Board, in its sole discretion, may determine that the delivery of Common Stock or the payment of cash, upon the exercise, vesting or settlement of all or a portion of any Stock Award may be deferred and may establish programs and procedures for deferral elections to be made by Participants. Deferrals by Participants will be made in accordance with Section 409A of the Code. Consistent with Section 409A of the Code, the Board may provide for distributions while a Participant is still an employee. The Board is authorized to make deferrals of Stock Awards and determine when, and in what annual percentages, Participants may receive payments, including lump sum payments, following the Participant's termination of employment or retirement, and implement such other terms and conditions consistent with the provisions of the Plan and in accordance with applicable law.
- (j) Compliance with Section 409A. To the extent that the Board determines that any Stock Award granted hereunder is subject to Section 409A of the Code, the Stock Award Agreement evidencing such Stock Award shall incorporate the terms and conditions necessary to avoid the consequences specified in Section 409A(a)(1) of the Code. To the extent applicable, the Plan and Stock Award Agreements shall be interpreted in accordance with Section 409A of the Code.
- (k) Compliance with Exemption Provided by Rule 12h-1(f). If at the end of the Company's most recently completed fiscal year: (i) the aggregate of the number of persons who hold outstanding compensatory employee stock options to purchase shares of Common Stock granted pursuant to the Plan or otherwise (such persons, "Holders of Options") equals or exceeds 500, and (ii) the Company's assets exceed \$10 million, then the following restrictions shall apply during any period during which the Company does not have a class of its securities registered under Section 12 of the Exchange Act and is not required to file reports under Section 15(d) of the Exchange Act: (A) the Options and, prior to exercise, the shares of Common Stock acquired upon exercise of the Options may not be transferred until the Company is no longer

relying on the exemption provided by Rule 12h-1(f) promulgated under the Exchange Act ("Rule 12h-1(f)"), except: (1) as permitted by Rule 701(c) promulgated under the Securities Act, (2) to a guardian upon the disability of the Holder of Options, or (3) to an executor upon the death of the Holder of Options (collectively, the "Permitted Transferees"); provided, however, that the following transfers are permitted: (i) transfers by the Holder of Options to the Company, and (ii) transfers in connection with a change of control or other acquisition involving the Company, if following such transaction, the Options no longer remain outstanding and the Company is no longer relying on the exemption provided by Rule 12h-1(f); provided further, that any Permitted Transferees may not further transfer the Options; (B) except as otherwise provided in (A) above, the Options and shares of Common Stock acquired upon exercise of the Options are restricted as to any pledge, hypothecation, or other transfer, including any short position, any "put equivalent position" as defined by Rule 16a-1(h) promulgated under the Exchange Act, or any "call equivalent position" as defined by Rule 16a-1(b) promulgated under the Exchange Act by the Holder of Options prior to exercise of an Option until the Company is no longer relying on the exemption provided by Rule 12h-1(f); and (C) at any time that the Company is relying on the exemption provided by Rule 12h-1(f), the Company shall deliver to Holders of Options (whether by physical or electronic delivery or written notice of the availability of the information on an internet site) the information required by Rule 701(e)(3), 701(e)(4) and 701(e)(5) promulgated under the Securities Act every six months, including financial statements that are not more than 180 days old; provided, however, that the Company may condition the delivery of such information upon the Holder of Options' agreement to maintain its confidentiality.

(I) Repurchase Limitation. The terms of any repurchase right shall be specified in the Stock Award Agreement. The repurchase price for vested shares of Common Stock shall be the Fair Market Value of the shares of Common Stock on the date of repurchase. The repurchase price for unvested shares of Common Stock shall be the lower of (i) the Fair Market Value of the shares of Common Stock on the date of repurchase or (ii) their original purchase price. However, the Company shall not exercise its repurchase right until at least six months (or such longer or shorter period of time necessary to avoid classification of the Stock Award as a liability for financial accounting purposes) have elapsed following delivery of shares of Common Stock subject to the Stock Award, unless otherwise specifically provided by the Board.

9. ADJUSTMENTS UPON CHANGES IN COMMON STOCK; OTHER CORPORATE EVENTS.

- (a) Capitalization Adjustments. In the event of a Capitalization Adjustment, the Board shall appropriately and proportionately adjust: (i) the class(es) and maximum number of securities subject to the Plan pursuant to Section 3(a), and (ii) the class(es) and number of securities and price per share of stock subject to outstanding Stock Awards. The Board shall make such adjustments, and its determination shall be final, binding and conclusive.
- **(b) Dissolution or Liquidation.** Except as otherwise provided in the Stock Award Agreement, in the event of a dissolution or liquidation of the Company, all outstanding Stock Awards (other than Stock Awards consisting of vested and outstanding shares of Common Stock not subject to a forfeiture condition or the Company's right of repurchase) shall terminate immediately prior to the completion of such dissolution or liquidation, and the shares of

Common Stock subject to the Company's repurchase rights or subject to a forfeiture condition may be repurchased or reacquired by the Company notwithstanding the fact that the holder of such Stock Award is providing Continuous Service.

- (c) Corporate Transaction. The following provisions shall apply to Stock Awards in the event of a Corporate Transaction unless otherwise provided in the instrument evidencing the Stock Award or any other written agreement between the Company or any Affiliate and the holder of the Stock Award or unless otherwise expressly provided by the Board at the time of grant of a Stock Award. Except as otherwise stated in the Stock Award Agreement, in the event of a Corporate Transaction, then, notwithstanding any other provision of the Plan, the Board may take one or more of the following actions with respect to Stock Awards, contingent upon the closing or completion of the Corporate Transaction:
- (i) arrange for the surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company) to assume or continue all or any portion of the Stock Award or to substitute a similar stock award for all or any portion of the Stock Award (including, but not limited to, an award to acquire the same consideration paid to the stockholders of the Company pursuant to the Corporate Transaction);
- (ii) arrange for the assignment of any reacquisition or repurchase rights held by the Company in respect of Common Stock issued pursuant to the Stock Award to the surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company);
- (iii) accelerate the vesting, in whole or in part, of the Stock Award (and, if applicable, the time at which the Stock Award may be exercised), with such Stock Award terminating if not exercised (if applicable) at or prior to the effective time of the Corporate Transaction; *provided, however*, that the Board may require Participants to complete and deliver to the Company a notice of exercise before the effective date of a Corporate Transaction, which is contingent upon the effectiveness of such Corporate Transaction;
 - (iv) arrange for the lapse of any reacquisition or repurchase rights held by the Company with respect to the Stock Award;
- (v) cancel or arrange for the cancellation of all or any portion of the Stock Award, to the extent not vested or not exercised prior to the effective time of the Corporate Transaction, in exchange for such cash consideration, if any, as the Board, in its sole discretion, may consider appropriate; and
- (vi) make a payment, in such form as may be determined by the Board equal to the excess, if any, of (A) the value of the property the holder of the Stock Award would have received upon the exercise of the Stock Award, over (B) any exercise price payable by such holder in connection with such exercise. For clarity, this payment may be zero (\$0) if the value of the property is equal to or less than the exercise price. Payments under this provision may be delayed to the same extent that payment of consideration to the holders of the Company's Common Stock in connection with the Corporate Transaction is delayed as a result of escrows, earn outs, holdbacks or any other contingencies.

The Board need not take the same action with respect to all Stock Awards or with respect to all Participants. The Board may take different actions with respect to the vested and unvested portions of a Stock Award.

(d) Change in Control. A Stock Award may be subject to additional acceleration of vesting and exercisability upon or after a Change in Control as may be provided in the Stock Award Agreement for such Stock Award or as may be provided in any other written agreement between the Company or any Affiliate and the Participant, but in the absence of such provision, no such acceleration shall occur.

10. TERMINATION OR SUSPENSION OF THE PLAN.

- (a) Plan Term. The Plan has been adopted by the Board subject to approval by (i) the TSX-V and, (ii) the Company's stockholders, and shall become effective upon the receipt of such approvals. The Plan shall continue in effect for a term of ten (10) years from the later of (A) the effective date of the Plan or (B) the most recent Board or stockholder approval of an increase in the number of shares of Common Stock reserved for issuance under the Plan. The Board may suspend or terminate the Plan at any time.
- **(b) No Impairment of Rights.** Suspension or termination of the Plan shall not impair rights and obligations under any Stock Award granted while the Plan is in effect except with the written consent of the affected Participant.

11. CHOICE OF LAW.

The law of the State of Delaware shall govern all questions concerning the construction, validity and interpretation of this Plan, without regard to that state's conflict of laws rules.

- 12. DEFINITIONS. As used in the Plan, the following definitions shall apply to the capitalized terms indicated below:
- (a) "Affiliate" means, at the time of determination, any "parent" or "majority-owned subsidiary" of the Company, as such terms are defined in Rule 405. The Board shall have the authority to determine the time or times at which "parent" or "majority-owned subsidiary" status is determined within the foregoing definition.
 - **(b)** "Board" means the Board of Directors of the Company.
- (c) "Capitalization Adjustment" means any change that is made in, or other events that occur with respect to, the Common Stock subject to the Plan or subject to any Stock Award after the Effective Date without the receipt of consideration by the Company (through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, large nonrecurring cash dividend, stock split, reverse stock split

liquidating dividend, combination of shares, exchange of shares, change in corporate structure, or any similar equity restructuring transaction, as that term is used in Financial Accounting Standards Board Accounting Standards Codification Topic 718. Notwithstanding the foregoing, the conversion of any convertible securities of the Company shall not be treated as a Capitalization Adjustment.

- (d) "Cause" shall have the meaning ascribed to such term in any written agreement between the Participant and the Company defining such term and, in the absence of such agreement, such term means with respect to a Participant, the occurrence of any of the following events: (i) such Participant's commission of any felony or any crime involving fraud, dishonesty or moral turpitude under the laws of the United States or any state thereof; (ii) such Participant's attempted commission of, or participation in, a fraud or act of dishonesty against the Company; (iii) such Participant's intentional, material violation of any contract or agreement between the Participant and the Company or of any statutory duty owed to the Company; (iv) such Participant's unauthorized use or disclosure of the Company's confidential information or trade secrets; or (v) such Participant's gross misconduct. The determination that a termination of the Participant's Continuous Service is either for Cause or without Cause shall be made by the Company in its sole discretion. Any determination by the Company that the Continuous Service of a Participant was terminated with or without Cause for the purposes of outstanding Stock Awards held by such Participant shall have no effect upon any determination of the rights or obligations of the Company or such Participant for any other purpose.
- (e) "Change in Control" means the occurrence, in a single transaction or in a series of related transactions, of any one or more of the following events:
- (i) any Exchange Act Person becomes the Owner, directly or indirectly, of securities of the Company representing more than fifty percent (50%) of the combined voting power of the Company's then outstanding securities other than by virtue of a merger, consolidation or similar transaction. Notwithstanding the foregoing, a Change in Control shall not be deemed to occur (A) on account of the acquisition of securities of the Company directly from the Company, (B) on account of the acquisition of securities of the Company by an investor, any affiliate thereof or any other Exchange Act Person that acquires the Company's securities in a transaction or series of related transactions the primary purpose of which is to obtain financing for the Company through the issuance of equity securities or (C) solely because the level of Ownership held by any Exchange Act Person (the "Subject Person") exceeds the designated percentage threshold of the outstanding voting securities as a result of a repurchase or other acquisition of voting securities by the Company reducing the number of shares outstanding, provided that if a Change in Control would occur (but for the operation of this sentence) as a result of the acquisition of voting securities by the Company, and after such share acquisition, the Subject Person becomes the Owner of any additional voting securities that, assuming the repurchase or other acquisition had not occurred, increases the percentage of the then outstanding voting securities Owned by the Subject Person over the designated percentage threshold, then a Change in Control shall be deemed to occur:

(ii) there is consummated a merger, consolidation or similar transaction involving (directly or indirectly) the Company and, immediately after the consummation of such merger, consolidation or similar transaction, the stockholders of the Company immediately prior thereto do not Own, directly or indirectly, either (A) outstanding voting securities representing more than fifty percent (50%) of the combined outstanding voting power of the surviving Entity in such merger, consolidation or similar transaction or (B) more than fifty percent (50%) of the combined outstanding voting power of the parent of the surviving Entity in such merger, consolidation or similar transaction, in each case in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such transaction; or

(iii) there is consummated a sale, lease, exclusive license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries, other than a sale, lease, license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries to an Entity, more than fifty percent (50%) of the combined voting power of the voting securities of which are Owned by stockholders of the Company in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such sale, lease, license or other disposition.

Notwithstanding the foregoing definition or any other provision of this Plan, (A) the term Change in Control shall not include a sale of assets, merger or other transaction effected exclusively for the purpose of changing the domicile of the Company, and (B) the definition of Change in Control (or any analogous term) in an individual written agreement between the Company or any Affiliate and the Participant shall supersede the foregoing definition with respect to Stock Awards subject to such agreement; *provided*, *however*, that if no definition of Change in Control or any analogous term is set forth in such an individual written agreement, the foregoing definition shall apply.

- (f) "Code" means the Internal Revenue Code of 1986, as amended, including any applicable regulations and guidance thereunder.
- **(g)** "Committee" means a committee of one or more Directors to whom authority has been delegated by the Board in accordance with Section 2(c), provided, however, that at any time the shares of Common Stock are registered under Section 12(b) or 12(g) of the Exchange Act, then the Board shall consider that certain exemptions from the provisions of Section 16(b) of the Exchange Act may not be available if such committee is composed other than solely of at least two (2) members who are "nonemployee directors" as contemplated by Rule 16b-3 of the Exchange Act.
 - (h) "Common Stock" means the common stock of the Company.
 - (i) "Company" means CohBar, Inc., a Delaware corporation.
 - (i) "Consultant" means an individual, other than an Employee or Director, that:

- (i) is engaged to provide on an ongoing bona fide basis, consulting, technical, management or other services to the Company or to an affiliate of the Company, other than services provided in relation to a distribution of securities of the Company;
 - (ii) provides the services under a written contract between the Company or any affiliate and the individual;
- (iii) in the reasonable opinion of the Company, spends or will spend a significant amount of time and attention on the affairs and business of the Company or an affiliate of the Company; and
- (iv) has a relationship with the Company or an affiliate of the Company that enables the individual to be knowledgeable about the business and affairs of the Company.
- (k) "Continuous Service" means that the Participant's service with the Company or an Affiliate, whether as an Employee, Director or Consultant, is not interrupted or terminated. A change in the capacity in which the Participant renders service to the Company or an Affiliate as an Employee, Director, or Consultant or a change in the Entity for which the Participant renders such service, provided that there is no interruption or termination of the Participant's service with the Company or an Affiliate, shall not terminate a Participant's Continuous Service; provided, however, that if the Entity for which a Participant is rendering service ceases to qualify as an Affiliate, as determined by the Board in its sole discretion, such Participant's Continuous Service shall be considered to have terminated on the date such Entity ceases to qualify as an Affiliate. For example, a change in status from an employee of the Company to a consultant of an Affiliate or to a Director shall not constitute an interruption of Continuous Service. To the extent permitted by law, the Board or the chief executive officer of the Company, in that party's sole discretion, may determine whether Continuous Service shall be considered interrupted in the case of (i) any leave of absence approved by the Board or chief executive officer, including sick leave, military leave or any other personal leave, or (ii) transfers between the Company, an Affiliate, or their successors. Notwithstanding the foregoing, a leave of absence shall be treated as Continuous Service for purposes of vesting in a Stock Award only to such extent as may be provided in the Company's leave of absence policy, in the written terms of any leave of absence agreement or policy applicable to the Participant, or as otherwise required by law.
- (I) "Corporate Transaction" means the occurrence, in a single transaction or in a series of related transactions, of any one or more of the following events:
- (i) the consummation of a sale or other disposition of all or substantially all, as determined by the Board in its sole discretion, of the consolidated assets of the Company and its Subsidiaries;
 - (ii) the consummation of a sale or other disposition of at least fifty percent (50%) of the outstanding securities of the Company;
- (iii) the consummation of a merger, consolidation or similar transaction following which the Company is not the surviving corporation; or

- (iv) the consummation of a merger, consolidation or similar transaction following which the Company is the surviving corporation but the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into other property, whether in the form of securities, cash or otherwise.
 - (m) "Director" means a member of the Board.
- (n) "Disability" means, with respect to a Participant, the inability of a Participant to engage in any substantially gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than twelve (12) months as provided in Sections 22(e)(3) and 409A(a)(2)(e)(i) of the Code and shall be determined by the Board on the basis of such medical evidence as the Board deems warranted under the circumstances.
- (o) "Effective Date" means the effective date of this Plan, which is the earlier of (i) the date that this Plan is first approved by the Company's stockholders, or (ii) the date this Plan is adopted by the Board.
- (p) "Employee" means any individual who is considered an employee of the Company or Affiliate under the Code (and for whom income tax deductions must be made at source).
 - (q) "Entity" means a corporation, partnership, limited liability company or other entity.
- (r) "Exchange" means the TSX-V, or any other securities exchange or quotation system on which the shares of Common Stock may be listed or quoted at the relevant time.
 - (s) "Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.
- (t) "Exchange Act Person" means any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act, except that "Exchange Act Person" shall not include (i) the Company or any Subsidiary of the Company, (ii) any employee benefit plan of the Company or any Subsidiary of the Company or any trustee or other fiduciary holding securities under an employee benefit plan of the Company or any Subsidiary of the Company, (iii) an underwriter temporarily holding securities pursuant to an offering of such securities, (iv) an Entity Owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their Ownership of stock of the Company; or (v) any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act) that, as of the Effective Date, is the Owner, directly or indirectly, of securities of the Company representing more than fifty percent (50%) of the combined voting power of the Company's then outstanding securities.

- (u) "Fair Market Value" means, as of any date, the value of shares of Common Stock determined as follows:
- (i) if the shares of Common Stock are listed on any established stock exchange, including without limitation the TSX Venture Exchange or the Toronto Stock Exchange, or a national market system, including without limitation the Nasdaq Global Select Market, the Nasdaq Global Market or the Nasdaq Capital Market, its Fair Market Value shall be the closing sales price for such stock (or the closing bid, if no sales were reported) as quoted on such exchange or system on the day of determination, as reported by such stock exchange or national market system;
- (ii) if shares of Common Stock are regularly quoted by a recognized securities dealer but selling prices are not reported, its Fair Market Value shall be the mean between the high bid and low asked prices for the shares of Common Stock on the day of determination; or
- (iii) in the absence of an established market for the shares of Common Stock, the Fair Market Value thereof shall be determined by the Board in compliance with Section 409A of the Code or, in the case of an Incentive Stock Option, in compliance with Section 422 of the Code.
- (v) "Incentive Stock Option" means an option that qualifies as an "incentive stock option" within the meaning of Section 422 of the Code and the regulations promulgated thereunder.
 - (w) "Insider" has the meaning defined in the rules of TSX-V.
 - (x) "Investor Relations Activities" has the meaning defined in the rules of TSX-V.
 - (y) "Nonstatutory Stock Option" means an Option that does not qualify as an Incentive Stock Option.
 - (z) "Officer" means any person designated by the Company as an officer.
- (aa) "Option" means an Incentive Stock Option or a Nonstatutory Stock Option to purchase shares of Common Stock granted pursuant to the Plan.
- **(bb)** "Option Agreement" means a written agreement between the Company and an Optionholder evidencing the terms and conditions of an Option grant. Each Option Agreement shall be subject to the terms and conditions of the Plan.
- (cc) "Optionholder" means a person to whom an Option is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Option.
- (dd) "Own," "Owned," "Owner," "Ownership" A person or Entity shall be deemed to "Own," to have "Owned," to be the "Owner" of, or to have acquired "Ownership" of securities if such person or Entity, directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has or shares voting power, which includes the power to vote or to direct the voting, with respect to such securities.

- (ee) "Participant" means a person to whom a Stock Award is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Stock Award.
 - (ff) "person" has the meaning set forth in the TSX Venture Exchange Corporate Finance Manual, as amended from time to time.
 - (gg) "Plan" means this CohBar, Inc. 2011 Equity Incentive Plan.
- (hh) "Restricted Stock Award" means an award of shares of Common Stock which is granted pursuant to the terms and conditions of Section 6(a).
- (ii) "Restricted Stock Award Agreement" means a written agreement between the Company and a holder of a Restricted Stock Award evidencing the terms and conditions of a Restricted Stock Award. Each Restricted Stock Award Agreement shall be subject to the terms and conditions of the Plan.
- (jj) "Restricted Stock Unit Award" means a right to receive shares of Common Stock which is granted pursuant to the terms and conditions of Section 6(b).
- (kk) "Restricted Stock Unit Award Agreement" means a written agreement between the Company and a holder of a Restricted Stock Unit Award evidencing the terms and conditions of a Restricted Stock Unit Award grant. Each Restricted Stock Unit Award Agreement shall be subject to the terms and conditions of the Plan.
 - (II) "Rule 405" means Rule 405 promulgated under the Securities Act.
 - (mm) "Rule 701" means Rule 701 promulgated under the Securities Act.
 - (nn) "Securities Act" means the Securities Act of 1933, as amended.
- (00) "Stock Appreciation Right" or "SAR" means a right to receive the appreciation on Common Stock that is granted pursuant to the terms and conditions of Section 5.
- (pp) "Stock Appreciation Right Agreement" means a written agreement between the Company and a holder of a Stock Appreciation Right evidencing the terms and conditions of a Stock Appreciation Right grant. Each Stock Appreciation Right Agreement shall be subject to the terms and conditions of the Plan.
- (qq) "Stock Award" means any right to receive Common Stock granted under the Plan, including an Incentive Stock Option, a Nonstatutory Stock Option, a Restricted Stock Award, a Restricted Stock Unit Award, or a Stock Appreciation Right.

- **(rr)** "Stock Award Agreement" means a written agreement between the Company and a Participant evidencing the terms and conditions of a Stock Award grant. Each Stock Award Agreement shall be subject to the terms and conditions of the Plan.
- (ss) "Subsidiary" means, with respect to the Company, (i) any corporation of which more than fifty percent (50%) of the outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether, at the time, stock of any other class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time, directly or indirectly, Owned by the Company, and (ii) any partnership, limited liability company or other entity in which the Company has a direct or indirect interest (whether in the form of voting or participation in profits or capital contribution) of more than fifty percent (50%).
- (tt) "Ten Percent Stockholder" means a person who Owns (or is deemed to Own pursuant to Section 424(d) of the Code) stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or any Affiliate.
 - (uu) "TSX-V" means the TSX Venture Exchange.

CohBar, Inc. Completes US\$11.25M IPO to Discover and Develop Novel Mitochondrial-derived Therapeutics

— Company Begins Trading on TSX Venture Exchange under the Symbol COB.U—

Pasadena, California — January 7 2015 — CohBar, Inc. (TSX-V: COB.U), an innovative biotechnology company focused on discovering mitochondrial-derived peptides (MDPs) and developing them into novel therapeutics to treat major diseases and extend healthy lifespans, today announced that it completed an Initial Public Offering (IPO) on January 6, 2015 of 11,250,000 units at a price of US\$1.00 per unit, providing gross proceeds of US\$11,250,000. Concurrently with the IPO, the company completed a previously-subscribed private placement of an additional 2,700,000 units for gross proceeds of US\$2,700,000 million, resulting in a total raise of almost US\$14 million. The initial public offering was conducted in Canada and registered in the United States pursuant to a registration statement filed with the U.S. Securities and Exchange Commission. CohBar has 32,265,343 shares of common stock issued and outstanding after completion of these financings.

The company also announced that its common stock was accepted for listing on the TSX Venture Exchange with commencement of trading on January 8, 2015, under ticker symbol COB.U. CohBar will report as a U.S. public company.

All units consist of one share of CohBar's common stock and one-half of one common stock purchase warrant. Each whole warrant is exercisable to acquire one share of CohBar's common stock at a price of US\$2.00 per share at any time up to January 6, 2017, subject to CohBar's right to accelerate the expiration time of the warrants if at any time the volume-weighted average trading price of its common stock is equal to or exceeds US\$3.00 per share for twenty (20) consecutive trading days.

Haywood Securities Inc. acted as sole agent for the IPO. CohBar issued a total of 786,696 compensation options to the agent and the selling group members in connection with the IPO. Each compensation option entitles the holder thereof to acquire a unit at a price of US\$1.00 at any time until July 6, 2016.

"We believe that the recently expanded mitochondrial genome holds vast potential for identifying and developing a new class of therapeutics," said Jon Stern, Chief Executive Officer of CohBar, "and our founders and their academic colleagues have spent more than a decade mining the mitochondrial genome and discovering genes that encode novel secreted peptides. Our current focus is on advancing therapeutic candidates that exploit these novel peptides for the treatment of diseases such as type 2 diabetes, cancer, Alzheimer's disease and atherosclerosis."

Mr. Stern continued, "The successful completion of these financing activities will enable us to expand and accelerate our research and development programs, and our listing on the TSX Venture Exchange together with registration in the U.S. should provide the company with broader access to the capital markets in support of our future needs."

CohBar's Scientific Leadership

CohBar's scientific leadership is centered around the company's founders, Dr. Pinchas Cohen, Dean of the Davis School of Gerontology at the University of Southern California, and Dr. Nir Barzilai, Professor of Genetics and Director of the Institute for Aging Research at the Albert Einstein College of Medicine, and is supported by its co-founders, Dr. David Sinclair, Professor of Genetics at Harvard Medical School, and Dr. John Amatruda, former Senior Vice President and Franchise Head for Diabetes and Obesity at

Merck Research Laboratories. Their multi-disciplinary expertise, and their investigations into and knowledge of age-related diseases, has enabled and focused CohBar's research efforts on the mitochondrial genome and its potential to yield mitochondrial-derived peptides, or MDP's, which the company believes have significant potential for development into therapeutic treatments for major age-related diseases such as type 2 diabetes, cancer, Alzheimer's disease and atherosclerosis.

The Mitochondrial Genome and CohBar's MDPs

Mitochondria, the only parts of the cell other than the nucleus that have their own genome, produce energy and regulate cell death in response to signals received from the cell. Until recently, scientists believed the mitochondrial genome contained only 37 genes and had been relatively unexplored as a focus of drug discovery efforts. Research by CohBar founders and their academic collaborators revealed that the mitochondrial genome has as many as 80 distinct new genes that encode peptides (small proteins in the body) which the company refers to as mitochondrial-derived peptides, or MDPs. MDPs influence cellular activities by acting as messengers between cells, triggering intra-cellular changes that affect cell growth and differentiation, and play a role in metabolism. MDPs represent a diverse collection of peptides, which the company believes have the potential to lead to novel therapeutics for a number of age-related diseases.

About CohBar

CohBar, Inc. (TSXV: COB.U) is an innovative biotechnology company focused on discovering mitochondrial-derived peptides (MDPs) and developing them into novel therapeutics to treat major diseases and extend healthy life spans. CohBar is a Delaware corporation headquartered in California that is listed on the TSXV and registered as a public company in the U.S.

For more company information, please visit <u>www.cohbar.com</u>. For additional company filings, please visit <u>www.sedar.com</u> and <u>www.sec.gov</u>.

Forward-looking statements

This news release contains forward-looking statements, including: statements concerning: the company's plans, prospects, resources and capabilities; anticipated research and development activities; the results and timing of the company's research programs; and the company's future financing needs and access to capital. Forward-looking statements are based on current expectations, estimates and projections that involve a number of risks and uncertainties that could cause actual results to differ materially from those anticipated by CohBar. These risks and uncertainties include CohBar's ability to retain key personnel, expand its research operations, and successfully advance its research programs. Additional assumptions, risks and uncertainties are described in detail in our registration statements, reports and other filings with the Securities and Exchange Commission and applicable Canadian securities regulators, which are available on our website, and at www.sedar.com. You are cautioned that such statements are not guarantees of future performance and that our actual results may differ materially from those set forth in the forward-looking statements. The forward-looking statements and other information contained in this news release are made as of the date hereof and CohBar does not undertake any obligation to update publicly or revise any forward-looking statements or information, whether as a result of new information, future events or otherwise, unless so required by applicable securities laws.

Neither TSX Venture Exchange nor its Regulation Services Provider (as that term is defined in the policies of the TSX Venture Exchange) accepts responsibility for the adequacy or accuracy of this release.

Contact:

Jon Stern, CEO (415) 388-2222 jon.stern@cohbar.com